ORANGE COUNTY BOARD OF SUPERVISORS

Agenda Revisions and Supplementals

Note: This supplemental agenda is updated daily showing items that have been added, continued, deleted or modified. No new supplemental items will be added to the agenda following close of business on Friday immediately prior to a Board meeting.

July 23, 2024

PRESENTATIONS (9:00 A.M.)

Supervisor Foley will be presenting a resolution recognizing Lasagna Love on National Lasagna Day July 29, 2024

DISCUSSION

25. Revised Title to read:

OC Community Resources - Acting as the Board of Supervisors - Approve Sale and Assignment and Assumption Agreements assigning HOME Investment Partnerships Program loan (\$500,000) from Jamboree-Tal Housing, L.P. to Amistad Partners I LP for Mendocino at Talega I; approve Amended and Restated Regulatory Agreements extending affordability period for 55 years; approve Subordination Agreements subordinating debt (\$21,000,000 first trust deed; \$6,300,000 second trust deed); approve waiver of transfer provision; and authorize Director or designee to subordinate additional senior debt and execute agreements and related documents under certain conditions; Acting as the Orange County Housing Authority - Approve Sale and Assignment and Assumption Agreements assigning Housing Successor loan (\$2,423,384) from Jamboree-Tal Housing, L.P. to Amistad Partners I L.P. for Mendocino at Talega I; approve Amended and Restated Regulatory Agreements extending affordability period for 55 years; approve Subordination Agreements subordinating debt (\$21,000,000 first trust deed; \$6,300,000 second trust deed); approve waiver of transfer provision; and authorize Executive Director of Orange County Housing Authority or designee to subordinate additional senior debt and execute agreements and related documents under certain conditions; approve Assignment and Assumption Agreement assigning Housing Successor loan (\$728,176) from Jamboree-Tal II Housing, L.P. to Amistad Housing Partners II LP for Mendocino at Talega II; approve Amended and Restated Regulatory Agreement extending affordability period for 55 years; approve Subordination Agreements subordinating debt (\$11,500,000 first trust deed); approve waiver of transfer provision; and authorize Executive Director of Orange County Housing Authority or designee to subordinate additional senior debt and execute agreements and related documents under certain conditions - District 5 (Continued from 6/4/24, Item 21; 6/25/24, Item 47)

THE FOLLOWING AGENDA ITEMS HAVE HAD CHANGES TO THEIR RECOMMENDED ACTIONS SINCE RELEASE OF THE AGENDA TO THE PUBLIC:

Item: 25

Supplemental Item(s)

- S49A. **Supervisor Sarmiento** Adopt resolution recognizing week of September 16 22, 2024 as "National Rehabilitation Awareness Week"
- S49B. **Supervisor Sarmiento** Adopt resolution recognizing week of October 20 26, 2024 as "Respiratory Care Week"

REVISIONS AND SUPPLEMENTALS TO JULY 23, 2024 AGENDA - PAGE 1 OF 2

ORANGE COUNTY BOARD OF SUPERVISORS

Agenda Revisions and Supplementals

Note: This supplemental agenda is updated daily showing items that have been added, continued, deleted or modified. No new supplemental items will be added to the agenda following close of business on Friday immediately prior to a Board meeting.

S49C Chairman Wagner - Commission to End Homelessness - Appoint Cecilia Bustamante Pixa, Seal Beach, for term ending 12/5/24

REVISIONS AND SUPPLEMENTALS TO JULY 23, 2024 AGENDA - PAGE 2 OF 2



MEMORANDUM

Could for DW

To: Robin Stieler, Clerk of the Board From: Supervisor Katrina Foley, Fifth District

Date: July 17, 2024

RE: Presentation to recognize Lasagna Love

Please add the following presentation to the July 23, 2024 Board of Supervisors meeting:

Supervisor Foley presenting a resolution recognizing Lasagna Love



Lasagna Love

. - 1

WHEREAS, Lasagna Love is a global nonprofit and grassroots movement that aims to positively impact communities by connecting neighbors through homemade meal delivery; and

WHEREAS, Lasagna Love originated at the beginning of the pandemic when a Southern California mom wanted to support families in her community. Lasagna Love seeks to eliminate stigmas associated with asking for help when it is needed most; and

WHEREAS, July 29th marks National Lasagna Day, promoting opportunity for people to indulge in this delicious layered pasta dish and share a meal with their loved ones; and

WHEREAS, Lasagna's origins trace to the rich culinary traditions of Italy, particularly the region of Emilia-Romagna. Originally evolving through the Middle Ages, lasagna was traditionally served during wedding feasts, Christmas festivities, and in the central farmlands of Marches; and

WHEREAS, Immigrants settled in various regions across the United States and introduced their native dishes to their new communities. Lasagna, with its hearty and flavorful layers of pasta, cheese, and sauce, quickly gained popularity among both Italian Americans and the broader American population; and

WHEREAS, Much like its namesake, Lasagna Love grew into an international movement delivering 3,500 lasagnas weekly to families in Australia, Canada and the United States.; and

WHEREAS, In Orange County more than 45 volunteers and 335 chefs have contributed efforts to help their community, one 9-by-13-inch pan of lasagna at a time; and

WHEREAS, Lasagna Love is dedicated to creating a ripple effect of positivity and goodwill in communities across the globe through their grassroots efforts; and

NOW, THEREFORE BE IT RESOLVED THAT THE ORANGE COUNTY BOARD OF

SUPERVISORS proclaim, recognize and appreciate Lasagna Love on National Lasagna Day 2024 for the extraordinary service and dedication to communities in Orange County and around the world. The Lasagna Love impact will undoubtedly inspire others to engage and contribute to the fabric of our community for years to follow.



Revision to ASR and/or Attachments

| Date: | 7/12/2024 | |
|----------|---|--|
| To: | Clerk of the Board of Supervisors | |
| CC: | County Executive Office Maguine Discontrol Orange, ou-CEO, endlandford aguine endland aguine | |
| From: | Dylan Wright, Director, OC Community Resources AV HV | |
| Re: | ASR Control #: <u>24-000269</u> , Meeting Date <u>7/23/2024</u> , Item No. # <u>25</u> | |
| Subject: | Approve Agreement for Mendocino at Talega I and II Resydications | |

Explanation:

The ASR Recommended Actions, Background Information, and Attachments have been revised in pursuant to County Counsel's advice.

Revised Recommended Action(s)

Acting as the Board of Supervisors:

- 1. Approve Assignment and Assumption Agreements to assign the County \$500,000 HOME Investment Partnerships Program Loan and Regulatory Agreement with Jamboree-Tal Housing, L.P. to a new limited partnership, Amistad Housing Partners I LP., upon resyndication and new tax-exempt bond financing for Mendocino at Talega I.
- 2. Approve Amended and Restated Regulatory Agreement for the \$500,000 HOME Investment Partnerships Program loan for Mendocino at Talega I to extend affordability period for 55 years.
- 3. Approve Subordination Agreements for Mendocino at Talega I to subordinate the County \$500,000 HOME Investment Partnerships Program loan plus accrued interest to a new first trust deed loan of approximately \$21 million at construction financing and \$10.6 million at permanent financing and a second trust deed loan of approximately \$6.3 million plus accrued interest at construction and permanent financing, with the ability to increase 10 percent due to an increase of construction costs, as set forth in this Agenda Staff Report, and authorize the OC Community Resources Director or designee to subordinate to additional senior debt up to 100

July 15, 2024

- 1011. Approve Amended and Restated Regulatory Agreement for the \$728,176 Housing Successor loan for Mendocino at Talega II to extend the affordability period for another 55 years.
- 1412. Approve Subordination Agreements for Mendocino at Talega II to subordinate the Housing Successor \$728,176 loan plus accrued interest to a new first trust deed loan of approximately \$11.5 million at construction financing and \$6.5 million at permanent financing and approximately \$4.7 million plus accrued interest at permanent financing; with the ability to increase 10 percent due to an increase of construction costs, as set forth in this Agenda Staff Report, and authorize the Orange County Housing Authority Executive Director or designee to subordinate to additional senior debt up to 100 percent of the cumulative loan to value based on the as-built appraised market value, if necessary, based on any future changes in the project financing.
- 4213. Approve waiving the requirement for a \$5,000 deposit and pay down of the outstanding \$2,423,384 Housing Successor loan for Mendocino at Talega I and \$728,176 loan for Mendocino at Talega II by one percent of the current outstanding balance under the transfer provisions of the Loan and Regulatory Agreement for Mendocino at Talega II.
- 4314. Authorize the Orange County Housing Authority Executive Director or designee to execute subordination agreements; an amended and restated set of loan documents and restrictive covenants; and such additional agreements, contracts, instructions and instruments necessary or appropriate for construction and permanent loan financing for Mendocino at Talega I and Mendocino at Talega II.
- 15. Approve the sale of Mendocino at Talega II by Jamboree-Tal II Housing, L.P. to Amistad Housing Partners II LP as described in the Purchase and Sale Agreement and Escrow Instructions included as Attachment R.

Make modifications to the:

Subject 🛛 🖾 Background Information 🗌 Summary 🗌 Financial Impact

Under Background Information, paragraphs 4 and 5:

For Talega I, Jamboree is requesting: approval of the Assignment and Assumption Agreements to assign the Talega I HSA and HOME loan and regulatory agreements from the current partnership entity, Jamboree-Tal Housing, L.P., to the new project limited partnership, Amistad Housing Partners I, LP. formed by JHC-Amistad LLC, the Managing General Partner, whose sole member is Jamboree, and the Investor Limited Partner (Attachments B, C, D and E); approval of the Amended and Restated Regulatory Agreements to extend the Agreement (Talega I)

+

Attachment L – Assignment and Assumption Agreement for \$728,176 Housing Successor Loan Agreement (Talega II)

Attachment M – Assignment and Assumption Agreement for \$728,176 Housing Successor Regulatory Agreement (Talega II)

Attachment N – Amended and Restated Regulatory Agreement for \$728,176 Housing Successor Loan (Talega II)

Attachment O – Subordination Agreement with First Trust Deed for Agreement \$728,176 Housing Successor Loan (Talega II)

Attachment P – Subordination Agreement with California Department of Housing and Community Development for \$728,176 Housing Successor Loan and Regulatory Agreement (Talega II)

Attachment Q – Talega I Purchase and Sale Agreement and Escrow Instructions Attachment R – Talega II Purchase and Sale Agreement and Escrow Instructions

Revised Attachments (attach revised attachment(s) and redlined copy(s))

Adding two additional attachments:

Attachment Q – Talega I Purchase and Sale Agreement and Escrow Instructions Attachment R – Talega II Purchase and Sale Agreement and Escrow Instructions

PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS

THIS PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS ("Agreement") is made as of this _____ day of _____, 2024 (the "Effective Date") by and between JAMBOREE-TAL HOUSING, L.P., a California limited partnership ("Seller"), and AMISTAD HOUSING PARTNERS I LP, a California limited partnership ("Buyer").

$\underline{R} \, \underline{E} \, \underline{C} \, \underline{I} \, \underline{T} \, \underline{A} \, \underline{L} \, \underline{S}$

A. Seller is the owner of that certain real property (the "Real Property") located in the City of San Clemente (the "City"), County of Orange, State of California, legally described on <u>Exhibit "A"</u>, which Real Property is improved with a 124 unit affordable housing development project.

B. Buyer wishes to purchase and Seller wishes to sell the Real Property, including the buildings, parking areas, improvements and fixtures now or hereafter located thereon (the "Improvements"), on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of \$100.00 paid by Buyer to Seller and in consideration of the mutual covenants, promises and undertakings set forth herein, Buyer and Seller hereby agree as follows:

I AGREEMENT TO PURCHASE

1.1 <u>The Property</u>. Subject to all of the terms, conditions and provisions of this Agreement, and for the consideration herein set forth, Seller hereby agrees to sell and Buyer hereby agrees to buy the following:

(a) Seller's fee estate in and to the Real Property, together with the Improvements and all tenements, hereditaments and appurtenances thereto;

(b) All furniture, personal property, machinery, apparatus and equipment which is owned by Seller and is located on the Real Property (the "Personal Property");

(c) All signs, logos, trade names or styles relating to the Real Property, if any, owned by Seller and to be assigned to Buyer pursuant to Section 2.5(m) hereof;

(d) To the extent assignable, all labor, service, supply and maintenance contracts relating to the Improvements or the Real Property which are to be assumed by Buyer pursuant to Section 3.2 below;

(e) All leases, lease deposits, and prepaid rentals related to the Real Property or the Improvements; and

(f) Any and all transferable guaranties and warranties covering the Real Property, the Improvements or any part thereof, if any, owned by Seller and which are

assignable by Seller, in each case, excluding any cash and any Personal Property of third parties.

The Real Property and the other property described in Paragraphs (a) through (f) above are hereinafter collectively referred to as the "Property".

1.2 <u>Purchase Price</u>. The purchase price which Seller agrees to accept and Buyer agrees to pay for the Property is the sum of SEVENTEEN MILLION ONE HUNDRED FIFTY THOUSAND DOLLARS (\$17,150,000) ("Purchase Price"). At Closing, Buyer shall receive credits against the Purchase Price in the amounts of: (1) \$470,836 for the existing replacement and operating reserves (the "Reserve Credit"), and (2) \$7,701 for short term work to be completed at the Property, as required by the California Tax Credit Allocation Committee ("TCAC") ("Short Term Work Credit").

1.3 <u>Carryback Financing</u>. In connection with the sale of the Property, Seller has agreed to make a loan to Buyer in the current estimated amount of Two Million Seven Hundred Seventy Four Thousand Four Ninety Five Dollars (\$2,774,095,) (the "Seller Loan"), which comprises the sum of the Purchase Price adjusted for closing costs and prorations, less the amounts assumed by Buyer pursuant to Section 1.4 below. To evidence the Loan, Buyer shall execute and deliver a Promissory Note Secured by Subordinated Deed of Trust in the form of <u>Exhibit "B"</u> attached hereto (the "Note") and a Subordinated Deed of Trust in the form of <u>Exhibit "C"</u> attached hereto (the "Seller Deed of Trust"). The Seller Deed of Trust shall be recorded against the Real Property and against Buyer's leasehold interest in certain other real property at Closing. Notwithstanding the foregoing, Buyer and Seller may agree to adjust the amount of the Seller Loan at or prior to Closing. The provisions of this Section shall survive the Closing.

1.4 <u>Assumed Loans</u>. Buyer shall assume:

(a) <u>HOME Loan</u>. That certain loan from the County of Orange (the "County"), or its affiliates, encumbering the Property in the original principal amount of Five Hundred Thousand Dollars (\$500,000), plus accrued interest in the current estimated amount of Two Hundred Twenty Six Thousand Three Hundred Twenty Eight Dollars (\$226,328), for a total outstanding amount of Seven Hundred Twenty Six Thousand Three Hundred Twenty Eight Dollars (\$726,328), pursuant to a Loan Agreement dated November 24, 2003, between the County and Seller (the "HOME Loan").

(b) <u>OCDA Loan</u>. That certain loan from the Orange County Development Agency ("OCDA"), or its affiliates, encumbering the Property in the original principal amount of Two Million Four Hundred Twenty Three Thousand Three Hundred Eighty Four Dollars (\$2,423,384), plus accrued interest in the current estimated amount of One Million One Hundred Sixteen Thousand Three Hundred Eighty Two Dollars (\$1,116,382), for a total outstanding amount of Three Million Five Hundred Thirty Nine Thousand Seven Hundred Sixty Nine Dollars (\$3,539,769), pursuant to a Loan Agreement dated November 24, 2003, between County and Seller (the "OCDA Loan").

(c) <u>City Loan</u>. That certain loan from the City of San Clemente ("City"), or its affiliates, encumbering the Property in the original principal amount of Six Hundred

Thirteen Thousand Eighty Dollars (\$613,080), pursuant to an Affordable Housing Project Agreement (Phase 1) dated June 6, 2001, between the City and Seller (the "City Loan").

II

ESCROW

2.1 <u>Escrow Holder</u>. Closing of the sale of the Property shall take place through an escrow ("<u>Escrow</u>") to be established with First American Title Insurance Company ("Escrow Holder"), 18500 Von Karman Avenue, Suite 600, Irvine, CA 92612, Attention: Ryan Hahn. Escrow shall be deemed open upon delivery of a fully executed copy of this Agreement to Escrow Holder. Escrow Holder shall immediately notify Buyer and Seller of the date of Opening of Escrow and the escrow number.

2.2 <u>Closing Date</u>. The "Closing Date" shall be on or before June 17, 2024. The terms "Close of Escrow" and "Closing" shall mean the date the Grant Deed (hereafter defined) is either filed for record or, subject to the following sentence, sent for recording in the Official Records of Orange County, California ("Official Records"). The parties agree to effectuate the Closing through a so-called "gap" closing and Seller agrees to provide a customary gap indemnity to the Title Company (as hereinafter defined).

2.3 <u>Escrow Instructions</u>. Articles 1, 2, 4, 5, 7 and 8 also constitute escrow instructions to Escrow Holder. Additionally, Buyer and Seller agree to execute any form of escrow instructions as the Escrow Holder customarily requires as escrow holder in real property escrows administered by it. In the event of a conflict between any such additional terms and provisions of this Agreement, this Agreement shall supersede and be controlling.

2.4 <u>Conveyance of Title</u>. At the Closing Date, Seller shall deliver to Escrow Holder a duly executed and acknowledged Grant Deed ("Grant Deed") in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period. The Grant Deed shall convey all of Seller's interest in fee title to the Real Property to Buyer at Closing. Escrow Holder is hereby instructed to record the Grant Deed in the Official Records, if and when Escrow Holder holds instruments and funds accruing to Buyer and Seller and can obtain for Buyer a CLTA standard coverage owner's policy of title insurance without endorsement issued by First American Title Insurance Company ("Title Company") with liability in an amount equal to the Purchase Price for the Property, showing the Real Property vested in Buyer, subject only to:

- (a) The Seller Deed of Trust;
- (b) Non-delinquent real property taxes and assessments;
- (c) Title exceptions approved by Buyer pursuant to Section 3.3 below;

(d) Title exceptions, if any, resulting from documents being recorded or delivered through Escrow (including, without limitation, any documents related to or arising from any new financing arranged by Buyer);

(e) Rights of tenants in possession, as tenants only;

(f) Any of the labor, service, supply and maintenance contracts referenced in Section 1.1(d) above;

(g) Discrepancies, conflicts in boundary lines, shortages in area, encroachments and any state of facts shown on the survey delivered to the Title Company;

(h) The standard preprinted exceptions and exclusions to the CLTA standard coverage owner's policy of title insurance;

(i) The Deed of Trust, Assignment of Rents and Security Agreement recorded on December 10, 2003, in the Official Records as Instrument Number No. 2003001466420 (as amended or modified from time to time, the "HOME Deed of Trust") relating to the HOME Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(j) The Deed of Trust, Assignment of Rents and Security Agreement recorded on December 10, 2003, in the Official Records as Instrument Number No. 2003001466415 (as amended or modified from time to time, the "OCDA Deed of Trust") relating to the OCDA Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(k) The Deed of Trust and Request for Special Notice recorded on March 28, 2002, in the Official Records as Instrument Number No. 20020259158 (as amended or modified from time to time, the "City Deed of Trust") relating to the City Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(1) The Master Declaration of Covenants, Conditions, Restrictions and Reservation of Easements for Talega recorded on September 24, 1999, in the Official Records as Instrument Number 19990686286 (as amended or modified from time to time, the "Master Declaration") that Buyer has agreed to assume;

(m) The Talega Lifestyle Fee Agreement recorded on August 1, 2001, in the Official Records as Instrument Number 20010525826 (as amended or modified from time to time, the "Talega Lifestyle Fee Agreement") that Buyer has agreed to assume;

(n) The Regulatory Agreement and Declaration of Covenants and Restrictions recoded on August 1, 2001, in the Official Records as Instrument Number 20010525827 (as amended or modified from time to time, the "City Regulatory Agreement") that Buyer has agreed to assume;

(o) The Regulatory Agreement and Declaration of Restrictive Covenants recorded on November 29, 2001, in the Official Records as Instrument Number 20010854462 (as amended or modified from time to time, the "Bond Regulatory Agreement") that Buyer has agreed to assume;

(p) The Regulatory Agreement and Declaration of Restrictive Covenants recorded on December 10, 2003, in the Official Records as Instrument Number 2003001466414 (as amended or modified from time to time, the "OCDA Regulatory"

Agreement") relating to the OCDA Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(q) The Regulatory Agreement and Declaration of Restrictive Covenants recorded on December 10, 2003, in the Official Records as Instrument Number 2003001466419 (as amended or modified from time to time, the "HOME Regulatory Agreement") relating to the HOME Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(r) The Regulatory Agreement recorded on February 11, 2004, in the Official Records as Instrument Number 2004000106936 (as amended or modified from time to time, the "TCAC Regulatory Agreement") that Buyer has agreed to assume; and

(s) The Declaration of Development Covenants, Conditions and Restrictions Concerning Lot 1 of Tract 13898 recorded on August 1, 2001, in the Official Records as Instrument Number 20010525822 (as amended or modified from time to time, the "Talega Declaration") that Buyer has agreed to assume.

2.5 <u>Additional Closing Obligations of Seller</u>. On or before 12:00 noon on the business day preceding the Closing Date, Seller shall deliver to Escrow Holder (unless indicated to be delivered directly to Buyer) copies of the following documents and other items:

(a) Two (2) duplicate copies of a Bill of Sale conveying the Personal Property related to its Real Property to Buyer in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period, duly executed by such Seller;

(b) Two (2) duplicate copies of an Assignment of Leases by Seller to Buyer (the "Assignment of Leases") in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period, duly executed by such Seller;

(c) A Certificate of Non-Foreign Status (the "Non-Foreign Affidavit") duly executed by such Seller in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period and a California Franchise Tax Board Real Estate Withholding Certificate duly executed by such Seller;

(d) One (1) original of an assignment and assumption of the HOME Loan and HOME Deed of Trust ("Assignment of HOME Loan Documents") in a form approved by Buyer, Seller and the County duly executed and acknowledged by the Seller;

(e) One (1) original of an assignment and assumption of the HOME Regulatory Agreement ("Assignment of HOME Regulatory Agreement") in a form approved by Buyer, Seller and the County duly executed and acknowledged by Seller;

(f) One (1) original of an assignment and assumption of the OCDA Loan and OCDA Deed of Trust ("Assignment of OCDA Loan Documents") in a form approved by Buyer, Seller and OCDA duly executed and acknowledged by the Seller;

(g) One (1) original of an assignment and assumption of the OCDA Regulatory Agreement in a form approved by Buyer, Seller and OCDA duly executed and acknowledged by Seller ("Assignment of OCDA Regulatory Agreement");

(h) One (1) original of an assignment and assumption of the City Loan and City Deed of Trust ("Assignment of City Loan Documents") in a form approved by Buyer, Seller and City duly executed and acknowledged by the Seller;

(i) One (1) original of an assignment and assumption of the City Regulatory Agreement ("Assignment of City Regulatory Agreement") in a form approved by Buyer, Seller and City duly executed and acknowledged by Seller;

(j) One (1) original of an assignment and assumption of the TCAC Regulatory Agreement ("Assignment of TCAC Regulatory Agreement") in a form approved by Buyer and Seller duly executed and acknowledged by Seller;

(k) One (1) original of an assignment and assumption of the Bond Regulatory Agreement ("Assignment of Bond Regulatory Agreement") in a form approved by Buyer and Seller duly executed and acknowledged by Seller;

(1) Originals or copies of all assumable contracts relating to the Property which Buyer is to assume pursuant to Section 3.2 hereof (to be delivered to Buyer at the Property);

(m) An assignment to Buyer of Seller's right, title and interest in and to (i) all contracts which Buyer is to assume pursuant to Section 3.2 below; (ii) all assignable permits, licenses and certificates of occupancy relating to the Property; (iii) all trade names, logos, signs, trademarks, telephone listings and numbers and similar items in Seller's possession included within the Property; and (iv) all warranties and guaranties then in effect, if any, with respect to the Improvements in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period ("Assignment of Contracts"), duly executed by such Seller;

(n) All soils, seismic, geologic, drainage, toxic waste, and environmental reports, surveys, "as-built" plans and specifications, working drawings, grading plans, elevations and similar information with respect to its Real Property heretofore obtained by such Seller which such Seller has in its possession and/or control to the extent such items have not been delivered previously by such Seller to Buyer pursuant to Section 3.1 below (to be delivered to Buyer at the Property);

(o) All keys to the Improvements which such Seller or such Seller's agents have in their possession (to be delivered to Buyer at the Property);

(p) The original executed copy of each lease of the Property (to the extent that Seller has the original leases in its possession or control), including any amendments thereto (to be delivered to Buyer at the Property);

(q) Two (2) duplicate copies of the Closing Statement (as hereinafter defined) duly executed by such Seller;

(r) A notice to each of the tenants of the Property in compliance with the provisions of California Civil Code Section 1950.5(h) (to be delivered to each tenant outside of Escrow); and

(s) Any other documents, instruments and records required to be delivered to Buyer under the terms of this Agreement which have not been previously delivered.

2.6 <u>Closing Obligations of Buyer</u>. On or before 12:00 noon on the business day preceding the Closing Date, Buyer shall deliver to Escrow Holder copies of the following documents and other items:

(a) One (1) original of all documents required by HCD in connection with the Buyer's HCD Loan (collectively, the "HCD Documents");

(b) One (1) original of the Assignment of HOME Loan Documents, duly executed and acknowledged by Buyer;

(c) One (1) original of the Assignment of HOME Regulatory Agreement, duly executed and acknowledged by Buyer;

(d) One (1) original of the Assignment of OCDA Loan Documents, duly executed and acknowledged by Buyer;

(e) One (1) original of the Assignment of OCDA Regulatory Agreement, duly executed and acknowledged by Buyer;

(f) One (1) original of the Assignment of City Loan Documents, duly executed and acknowledged by Buyer;

(g) One (1) original of the Assignment of City Regulatory Agreement, duly executed and acknowledged by Buyer;

(h) One (1) original of the Assignment of TCAC Regulatory Agreement, duly executed and acknowledged by Buyer;

(i) One (1) original of the Assignment of Bond Regulatory Agreement, duly executed and acknowledged by Buyer;

(j) One (1) original of the Note, duly executed by Buyer;

(k) One (1) original of the Seller Deed of Trust, duly executed and acknowledged by Buyer;

(l) Two (2) duplicate copies of the Assignment of Leases, duly executed by Buyer;

(m) Two (2) duplicate copies of the Assignment of Contracts, duly executed by Buyer;

(n) Two (2) duplicate copies of the Closing Statement, duly executed by Buyer;

(o) Evidence of the existence, organizational authority of Buyer and of the authority of persons executing documents on behalf of Buyer reasonably satisfactory to Escrow Holder and the Title Company; and

(p) Any other documents, instruments or funds required to be delivered by Buyer under the terms of this Agreement or otherwise required by Escrow Holder or Title Company in order to close Escrow which have not previously been delivered.

2.7 <u>Delivery of Documents by Escrow Holder</u>. All of the items listed in Section 2.5 above shall be delivered to Buyer through Escrow on the Closing Date (unless indicated to the contrary), except that Escrow Holder is hereby instructed to record the original Grant Deed, Seller Deed of Trust, HCD Documents (as applicable), Assignment of HOME Loan Documents, Assignment of HOME Regulatory Agreement, Assignment of OCDA Loan Documents, Assignment of OCDA Regulatory Agreement, Assignment of City Loan Documents, Assignment of Bond Regulatory Agreement (collectively, the "Recordable Documents") in the Official Records upon Close of Escrow in the order approved by Buyer and Seller and to deliver the original Recordable Documents to Seller. Escrow Holder shall deliver to Seller one (1) duplicate copy of all of the items listed in Section 2.6 on the Close of Escrow and conformed copies of the Recordable Documents.

III

INSPECTIONS, REVIEW

3.1 Access to Books and Records; Items to be Inspected and Reviewed by Buyer. Within five (5) days following the execution of this Agreement by Buyer and Seller, Seller shall make available to Buyer for Buyer's inspection at the Property or at Seller's property manager's offices during normal business hours upon not less than forty-eight (48) hours' prior notice, copies of all books, records, reports, surveys, bills, labor, service, supply and maintenance contracts, copies of all leases, rental applications and lease files, books and records of Seller relating to the Property in Seller's possession which are reasonably requested by Buyer. Seller has not made any independent investigation to determine the truth or accuracy of any such documents and materials and shall have no liability to Buyer for any inaccuracy, misrepresentation or omission in such documents or materials. Notwithstanding anything to the contrary contained herein, Seller shall have no obligation to deliver to Buyer (i) those documents, reports, presentations, summaries and the like prepared for any of Seller's partners or investors in connection with its consideration of the acquisition or sale of the Property, (ii) Seller's attorney-client privileged materials, (iii) appraisals, or (iv) any information which is subject to a confidentiality agreement between Seller and a third party. In addition, prior to the expiration of the Due Diligence Period (as hereinafter defined), Seller shall deliver to Buyer a lead-based paint disclosure statement in the form required by applicable law and a natural hazards disclosure statement in the form prescribed by California Civil Code § 1103.2 prepared by a licensed professional described in California Civil Code § 1103.4(c).

3.2 Physical Inspection. From and after the date hereof through for a period of thirty (30) days (the "Due Diligence Period"), Seller shall permit Buyer, its engineers, analysts, contractors, agents and consultants, at the sole cost and expense of Buyer, to conduct physical inspections of the Property and Improvements, including the structural, electrical and mechanical aspects of the Improvements to the Property, supports, site work, foundations, soil, subsurface soils, drainage, seismic and other geological and topographical matters, location of asbestos, leadbased paint, mold, fungi, toxic substances, hazardous materials or wastes, if any, and any other investigations as Buyer deems prudent with respect to the physical condition of the Property. Such investigations may be made by Buyer and/or its agents during any normal business hours. Buyer shall also have the right to investigate all matters relating to the zoning, use and compliance with other applicable laws which relate to the use and occupancy of the Property. Seller shall reasonably cooperate with Buyer in completing such inspections and special investigations at no cost or expense to Seller. Such inspections and investigations shall be conducted only upon no less than forty-eight (48) hours' notice to Seller and shall be conducted at such times and in such a manner as to minimize any disruption to tenants upon the Property. Seller shall have the right, but not the obligation, to accompany Buyer during such investigations and/or inspections. Notwithstanding the foregoing, Buyer shall not perform any invasive or destructive testing of the Property without the prior written consent of Seller, which consent may be withheld in Seller's sole and absolute discretion. Buyer shall repair any and all damage to the Property or to any tenants' property caused by such inspections or investigations in a timely manner and shall indemnify, defend (with counsel acceptable to Seller) and hold harmless Seller and its partners, affiliates, officers, employees, agents and consultants (the "Indemnified Parties") from and against any liability whatsoever sought by the Indemnified Parties arising from or related to Buyer's physical inspection of the Property or access to the Property granted hereunder to Buyer, its engineers, analysts, contractors, agents and consultants, including any such liability to which the negligence of the indemnified parties may have contributed but excepting any liability to the extent arising from the gross negligence or willful misconduct of the indemnified parties. As a condition to allowing Buyer, or its engineers, analysts, contractors, agents or consultants access to the Property, Buyer shall obtain and deliver to Seller a certificate of insurance evidencing liability insurance policy in an amount not less than TWO MILLION DOLLARS (\$2,000,000.00) and naming Seller as an additional insured. Prior to the expiration of the Due Diligence Period, if, in Buyer's sole and absolute discretion, Buyer is dissatisfied with the results of its review of the documents related to the Property or the inspections or investigations of the Property, then Buyer may terminate this Agreement by giving written notice of termination to Seller and Escrow Holder. In addition, Buyer shall, not later than the expiration of the Due Diligence Period, advise Seller in writing of the labor, service, supply and maintenance contracts which it elects to assume; provided, however, that Buyer shall be obligated to assume those contracts which cannot be terminated by Seller without payment of a fee or penalty. If Buyer notifies Seller of its disapproval of its due diligence investigation of the Property on or before the expiration of the Due Diligence Period, this Agreement and the escrow established hereunder shall terminate.

3.3 <u>Preliminary Title Report</u>. Within five (5) days of the Effective Date, Buyer shall obtain from Title Company a preliminary title report describing the state of title of the Property (the "Preliminary Title Report"). Buyer shall notify Seller in writing ("Buyer's Objection Notice") of any objections Buyer may have to title exceptions contained in the Preliminary Title Report not later than seven (7) business days prior to the expiration of the Due Diligence Period ("Title Review Period"). In the event Buyer fails to deliver Buyer's Objection Notice prior to the

expiration of the Title Review Period, Buyer shall conclusively be deemed to have approved the state of title of the Property. Seller shall have a period of five (5) business days after receipt of Buyer's Objection Notice in which to deliver written notice to Buyer ("Seller's Notice") of Seller's election to either (i) agree to remove the objectionable items prior to the Close of Escrow, or (ii) decline to remove any such title exceptions and terminate Escrow and this Agreement. If Seller fails to deliver Seller's notice within said five (5) business day period, Seller shall be deemed to have elected to decline to remove such title exceptions and terminate Escrow and this Agreement. If Seller notifies or is deemed to have notified Buyer of its election to terminate Escrow rather than remove the objectionable items, Buyer shall have the right, by written notice delivered to Seller not later than the expiration of the Due Diligence Period to agree to accept the Property subject to the objectionable items, in which event Seller's election to terminate the Escrow shall be of no effect, and Buyer shall take title at the Close of Escrow subject to such objectionable items without any adjustment to or credit against the Purchase Price. Notwithstanding the foregoing, except as set forth herein, Seller shall, on or before the Close of Escrow, remove (i) all deeds of trust and financing statements, including, without limitation, that certain Department of Housing and Community Development Multifamily Housing Program NOFA of February 23, 2001 Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing (Permanent Financing) recorded on December 10, 2003, in the Official Records as Instrument Number 2003001466405 (collectively, "Monetary Liens"), and (ii) that certain Department of Housing and Community Development Multifamily Housing Program (NOFA 2/23/01) recorded on December 10, 2003, in the Official Records as Instrument Number 200300146640, as encumbrances to title, without the necessity for Buyer to object thereto. Upon the issuance of any amendment or supplement to the Preliminary Title Report which has additional exceptions for matters not shown on the original Preliminary Title Report, the foregoing right of review and approval shall also apply to said amendment or supplement (provided that the period for Buyer to review such amendment or supplement shall be the later of the expiration of the Title Review Period or five (5) days from receipt of the amendment or supplement). If Buyer properly notifies Seller of its disapproval of the state of title to the Property on or before the expiration of the Title Review Period (or such longer period as may be provided in the immediately sentence), this Agreement and the escrow established hereunder shall terminate.

IV

CONDITIONS TO AGREEMENT

4.1 <u>Buyer's Conditions Precedent</u>. Buyer's obligation to purchase the Property shall be conditioned upon the fulfillment of the following conditions precedent, all of which shall be satisfied or waived in writing pursuant to Section 4.4 below prior to the Close of Escrow except as indicated otherwise:

(a) Seller has delivered to Escrow Holder all of the documents required to be delivered under Section 2.5.

(b) The due performance by Seller of all material obligations to be performed by Seller hereunder and the truth in all material respects of the representations and warranties made by Seller pursuant to Section 6.4. (c) No order or injunction of any court or administrative agency of competent jurisdiction nor any statute, rule, regulation or executive order promulgated by any governmental authority of competent jurisdiction shall be in effect as of the Close of Escrow which restrains or prohibits the transfer of the Property or the consummation of any other transaction contemplated hereby.

(d) The Department of Housing and Community Development, a public agency of the State of California ("HCD") has approved of a new loan to Buyer (the "Buyer's HCD Loan").

(e) The County has approved of the transfer of the Property to Buyer and the Assignment of the HOME Loan Documents, the Assignment of HOME Regulatory Agreement and the Assignment of the Bond Regulatory Agreement.

(f) OCDA has approved of the transfer of the Property to Buyer and the Assignment of the OCDA Loan Documents and Assignment of OCDA Regulatory Agreement.

(g) The City has approved of the transfer of the Property to Buyer and the Assignment of the City Loan Documents and Assignment of City Regulatory Agreement.

(h) TCAC has approved of the transfer of the Property to Buyer and the Assignment of TCAC Regulatory Agreement.

4.2 <u>Seller's Conditions Precedent</u>. Seller's obligation to convey the Property to Buyer shall be conditioned upon the satisfaction or written waiver in writing, in whole or in part by Seller of the following conditions precedent:

(a) Buyer has delivered to Escrow Holder all of the documents required to be delivered under Section 2.6.

(b) The due performance by Buyer of all material obligations to be performed by Buyer hereunder and the truth in all material respects of the representations and warranties made by Buyer pursuant to Section 6.5.

(c) No order or injunction of any court or administrative agency of competent jurisdiction nor any statute, rule, regulation or executive order promulgated by any governmental authority of competent jurisdiction shall be in effect as of the Close of Escrow which restrains or prohibits the transfer of the Property or the consummation of any other transaction contemplated hereby.

(d) HCD has approved the Buyer's HCD Loan.

(e) The County has approved of the transfer of the Property to Buyer and the Assignment of the HOME Loan Documents, the Assignment of HOME Regulatory Agreement and the Assignment of the Bond Regulatory Agreement, and the release of Seller from liability under the HOME Loan Documents, the HOME Regulatory Agreement and the Bond Regulatory Agreement for matters first arising after the Closing Date.

(f) OCDA has approved of the transfer of the Property to Buyer and the Assignment of the OCDA Loan Documents and the Assignment of the OCDA Regulatory Agreement, and the release of Seller from liability under the OCDA Loan Documents and the OCDA Regulatory Agreement for matters first arising after the Closing Date.

(g) The City has approved of the transfer of the Property to Buyer and the Assignment of the City Loan Documents and the Assignment of the City Regulatory Agreement, and the release of Seller from liability under the City Loan Documents and the City Regulatory Agreement for matters first arising after the Closing Date.

(h) TCAC has approved of the transfer of the Property to Buyer and the Assignment of the TCAC Regulatory Agreement, and the release of Seller from liability under the TCAC City Regulatory Agreement for matters first arising after the Closing Date.

4.3 <u>Satisfaction of Conditions</u>. Where satisfaction of any of the foregoing conditions requires action by Buyer or Seller, each party shall use its diligent best efforts, in good faith, and at its own cost, to satisfy such condition; provided, however, that in no event shall Seller be required to expend money related to Buyer's financing or in order to satisfy any condition of Buyer.

4.4 <u>Waiver</u>. Buyer may at any time or times, at its election, waive any of the conditions set forth in Section 4.1 above to its obligations hereunder, but any such waiver shall be effective only if contained in a writing signed by Buyer and delivered to Seller. Seller may at any time or times, at its election, waive any of the conditions set forth in Section 4.2 above to its obligations hereunder, but any such waiver shall be effective only if contained in a writing signed by Seller and delivered to Buyer.

4.5 Termination. In the event each of the conditions set forth in Section 4.1 is not fulfilled within the time provided in Section 4.1 or waived by Buyer pursuant to Section 4.4, Buyer may, at its option, terminate this Agreement and the Escrow opened hereunder, thereby releasing the parties from further obligations hereunder, and all documents delivered by Buyer to Seller or Escrow Holder shall be returned to Buyer and all documents delivered by Seller to Buyer or Escrow Holder returned to Seller. In the event that the conditions set forth in Section 4.2 are not fulfilled or waived by Seller pursuant to Section 4.4 prior to the Closing Date, Seller may, at its option, terminate this Agreement and the Escrow opened hereunder, thereby releasing the parties from further obligations hereunder, and all documents delivered by Seller to Buyer or Escrow Holder shall be returned immediately to Seller. In the event this Agreement terminates for any reason pursuant to this Section 4.5, within five (5) days of the date of termination, Buyer shall deliver to Seller all due diligence materials delivered to Buyer pursuant to Section 3.1 above as well as true, correct and complete copies of all reports, studies, documents or surveys generated in connection with Buyer's due diligence investigation of the Property without representation or warranty as to the accuracy or completeness of such documents.

V

PRORATIONS, POSSESSION AND DEPOSITS

5.1 <u>Proration of Taxes</u>. Real and personal property taxes and assessments for the Property shall be prorated by the parties based on a 365-day year as of 11:59 p.m. on the Closing Date with Seller paying such taxes on the day of Closing. All tax prorations shall be based upon the latest available tax statement. If the tax statements for the fiscal tax year during which Escrow closes do not become available until after the Closing Date, the parties shall re-prorate said taxes and assessments outside of Escrow following the Closing Date when such tax statements become available. Seller shall be responsible for and shall pay or reimburse Buyer upon demand for any real or personal property taxes payable following the Closing Date with respect to its Real Property applicable to any period of time prior to the Closing Date as a result of any change in the tax assessment by reason of reassessment, changes in use of the Property, changes in ownership, errors by the Assessor or otherwise. Any tax refunds which relate to any period of time prior to Close of Escrow shall accrue to the benefit of Seller.

5.2 <u>Proration of Rents</u>. Rents of the tenants under the leases shall be prorated on a 365day year as of 11:59 p.m. in on the Closing Date based on rents actually collected with Seller receiving the rents on the day of Closing. Any such rents collected after the Closing Date by Buyer which are attributable to the period prior to the Closing Date shall be paid to Seller upon collection. Rents collected after the Closing Date from tenants whose rental was delinquent at Closing shall be deemed to first apply to costs of collecting such rents, second to rentals which were delinquent at Closing and third to rents due at the time of payment accruing after the Closing Date. Buyer shall have no obligation to commence litigation or to collect rents or to terminate the tenant's right to occupancy based upon tenant's failure to pay rentals which were delinquent at Closing; however, Buyer shall use reasonable efforts to collect such delinquent rents and shall reasonably and in good faith cooperate with Seller's attempts to collect such rents at no cost or expense to Buyer.

5.3 <u>Tenants' Deposits</u>. The security deposits and other tenant deposits (to the extent not previously applied by Seller) shall be credited to Buyer at the Closing. From and after the Closing, Buyer hereby assumes the obligations of the landlord under the leases with respect to the security deposits and any other tenant deposits. This provision shall survive the closing.

5.4 <u>Utilities</u>. Seller shall use its best efforts to have utility meters read as of the date that Escrow closes and shall be responsible for all utility services to the Property until the Closing Date. In the event Seller is unable to have the utility meters read as of the date Escrow closes, Buyer and Seller shall jointly prepare and deposit an estimated utility statement based upon the average daily usage on the most recent utility bill preceding the Close of Escrow with Seller paying the utility costs for the day of Closing. Escrow Holder shall initially prorate utilities based upon such estimated utility statements and the parties shall subsequently prorate utilities based upon the actual utility usage upon receipt of such utility statements.

5.5 <u>Maintenance Contracts</u>. Seller shall be responsible for payment of all maintenance services, such as janitorial services, guard services and similar services through the Closing. Buyer shall be responsible for such services thereafter for any such contracts which it assumes pursuant to Section 3.2 above.

5.6 <u>Method of Adjustment</u>. The obligations of Sections 5.1, 5.2, 5.3, 5.4, and 5.5 shall be handled by Buyer and Seller outside of Escrow once a final accounting of all figures have been obtained. Seller and Buyer shall complete all adjustments for items to be prorated pursuant to this Article V within the earlier of ninety (90) days after the Closing Date or promptly after the receipt of actual bills or verifiable information relating to the item to be adjusted.

5.7 <u>Insurance</u>. Seller shall cause its policies of insurance for the Property to be terminated effective immediately after the Closing Date and Buyer shall be responsible for obtaining its own insurance.

5.8 <u>Possession</u>. Subject to the rights of tenants in possession, Buyer shall be entitled to possession of the Property on the Closing Date.

5.9 <u>Closing Costs</u>. Buyer shall pay all costs in connection with the closing of this transaction, including the premium for a policy of title insurance, any county or city documentary or transfer taxes payable on account of the conveyance of the Real Property, the escrow fees which may be charged by Escrow Holder, the cost of any survey required by the Title Company to issue its title policy, the cost of recording the Recordable Documents, any and all costs and expenses associated with or arising from Buyer's new financing.

5.10 <u>Closing Statement</u>. No later than three (3) business days prior to the Closing Date, Escrow Holder shall prepare for approval by Buyer and Seller a closing statement ("Closing Statement") on Escrow Holder's standard form indicating, among other things, the Replacement Reserve Credit, the Short Term Work Credit and Escrow Holder's estimate of all closing costs and prorations made pursuant to this Agreement. Buyer and Seller shall assist Escrow Holder in determining the amount of all prorations.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

BREACH BY SELLER. IN THE EVENT THAT SELLER BREACHES ANY 5.11 OF ITS OBLIGATIONS UNDER THIS AGREEMENT WHETHER PRIOR TO OR AFTER THE CLOSE OF ESCROW, BUYER'S SOLE REMEDY SHALL BE THE RIGHT TO SUE FOR ACTUAL DAMAGES (NOT TO EXCEED \$25,000), AND THIS AGREEMENT SHALL TERMINATE AND, EXCEPT AS SPECIFICALLY PROVIDED FOR ELSEWHERE IN THIS AGREEMENT, NEITHER PARTY SHALL HAVE ANY FURTHER OBLIGATION OR LIABILITY TO THE OTHER AT LAW OR IN EQUITY. IN NO EVENT SHALL SELLER EVER BE LIABLE TO BUYER UNDER ANY STATUTORY, COMMON LAW, EQUITABLE OR OTHER THEORY OF LAW EITHER PRIOR TO OR FOLLOWING THE CLOSING FOR ANY LOST RENTS, PROFITS, **BUSINESS OPPORTUNITIES OR ANY FORM OF CONSEQUENTIAL OR PUNITIVE** DAMAGES IN CONNECTION WITH ANY CLAIM, LIABILITY, DEMAND OR CAUSE OF ACTION IN ANY WAY OR MANNER RELATING TO THE PROPERTY. THE CONDITION OF THE PROPERTY, THIS AGREEMENT, OR ANY TRANSACTION OR MATTER BETWEEN THE PARTIES CONTEMPLATED HEREUNDER. **BUYER** HEREBY WAIVES, RELEASES AND FOREVER RELINQUISHES ANY RIGHT BUYER MAY NOW HAVE OR MAY LATER ACOUIRE TO SEEK SPECIFIC PERFORMANCE, TO PLACE ANY LIENS OR ENCUMBRANCES, INCLUDING A NOTICE OF LIS PENDENS, ON THE PROPERTY.

Buyer's Initials

Seller's Initials

a. LIOUIDATED DAMAGES. IN THE EVENT THAT CLOSING FAILS TO OCCUR DUE TO A DEFAULT OF BUYER, THE DAMAGES THAT SELLER WILL INCUR BY REASON THEREOF ARE AND WILL BE IMPRACTICAL AND EXTREMELY DIFFICULT TO ESTABLISH. BUYER AND SELLER, IN A **REASONABLE EFFORT TO ASCERTAIN WHAT SELLER'S DAMAGES WOULD BE** IN THE EVENT OF EITHER SUCH A DEFAULT BY BUYER, HAVE AGREED THAT SUCH DAMAGES SHALL BE IN AN AMOUNT EQUAL TO \$25,000, AND THAT SUCH AMOUNT SHALL BE DELIVERED TO SELLER UPON SUCH DEFAULT BY BUYER OR FAILURE OF CONDITION AND RETAINED BY SELLER AS LIOUIDATED DAMAGES, WHICH DAMAGES SHALL BE SELLER'S SOLE AND EXCLUSIVE **REMEDY AT LAW OR IN EOUITY IN THE EVENT OF AND FOR SUCH DEFAULT BY** WITHOUT LIMITING THE FOREGOING PROVISIONS OF THIS BUYER. PARAGRAPH, SELLER WAIVES ANY AND ALL RIGHTS WHICH SELLER **OTHERWISE WOULD HAVE HAD UNDER CALIFORNIA CIVIL CODE SECTION** 3389 TO SPECIFICALLY ENFORCE THIS AGREEMENT. SELLER AND BUYER ACKNOWLEDGE AND AGREE THAT THEY HAVE READ AND UNDERSTAND THE PROVISIONS OF THIS PARAGRAPH AND BY THEIR INITIALS AGREE TO BE **BOUND BY ITS TERMS.**

Buyer's Initials

Seller's Initials

AS-IS PURCHASE

6.1 Buyer acknowledges and agrees that it has been provided the Disclaimer. opportunity to thoroughly inspect, investigate and exercise due diligence, and the opportunity to fully and independently become familiar with, and fully satisfy itself regarding, any and all matters relating to the Property. Except as expressly provided in Section 6.4 of this Agreement, Seller has not made, does not make and specifically negates and disclaims any representations, warranties, promises, covenants, agreements or guaranties of any kind or character whatsoever, whether express or implied or arising by operation of law, oral or written, past, present or future, of, as to, concerning or with respect to (i) the value of the Property; (ii) the income to be derived from the Property; (iii) the suitability of the Property for any and all activities and uses which Buyer may conduct thereon, including, without limitation, the possibilities for future development of the Property; (iv) the habitability, merchantability, marketability, profitability or fitness for a particular purpose of the Property; (v) the manner, quality, state of repair or lack of repair of the Property; (vi) the nature, quality or condition of the Property, including, without limitation, the water, soil and geology; (vii) the compliance of or by the Property or its operation with any laws, rules, ordinances or regulations of any applicable governmental authority or body including, without limitation, Title III of the Americans With Disabilities Act of 1990; (viii) the manner or quality of the construction or materials incorporated into the Property; (ix) compliance with any environmental protection, pollution or land use laws, rules, regulations, orders or requirements including, without limitation, the Federal Water Pollution Control Act, the Federal Resource Conservation and Recovery Act, the U.S. Environmental Protection Agency Regulations at 40 C.F.R., Part 261, the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, the Resource Conservation and Recovery Act of 1976, the Clean Water Act, the Safe Drinking Water Act, the Hazardous Materials Transportation Act, the Toxic Substance Control Act, and regulations promulgated under any of the foregoing (collectively, "Environmental Laws"); (x) the presence or absence of hazardous materials at, on, under, or adjacent to the Property including, but not limited to, petroleum products, asbestos, lead-based paint, mold or fungi, (xi) the content, completeness or accuracy of any items delivered or made available to Buyer pursuant to Section 3.1; (xii) the conformity of the Improvements to any plans or specifications for the Property, including any plans and specifications that may have been or may be provided to Buyer; (xiii) the conformity of the Property to past, current or future applicable zoning or building requirements; (xiv) deficiency of any undershoring; (xv) deficiency of any drainage; (xvi) the fact that all or a portion of the Property may be located on or near an earthquake fault line; (xvii) the existence of vested land use, zoning or building entitlements affecting the Property; and (xviii) any matters arising from or relating to Seller's alleged superior knowledge or breach of any duty to disclose. Seller is not liable or bound in any manner by any oral or written statements, representations or information pertaining to the Property, or the operation thereof, furnished by any real estate broker, agent, employee, servant or other person. Except as provided in Section 6.4, Buyer further acknowledges and agrees that to the maximum extent permitted by law, the sale of the Property as provided for herein is made on an "AS-IS" "WHERE IS" condition and basis as of the Closing Date with all faults, and that Seller has no obligations to make repairs, replacements or improvements, whether before or after the Close of Escrow. Buyer represents, warrants and covenants to Seller, which representation, warranty, and covenant will survive the Close of Escrow and not be merged with the deed, that, except for the representations and warranties set forth in Section 6.4 of this Agreement, Buyer will solely rely upon Buyer's own

investigation of the Property and all other matters relating to the Property, and not on any information provided or to be provided by Seller or anyone acting on Seller's behalf.

6.2 <u>Release and Waiver</u>.

Except with respect to the representations and warranties of Seller set forth (a) in Section 6.4 of this Agreement, Buyer hereby fully and forever releases, acquits and discharges Seller of and from, and hereby fully and forever waives any and all claims, actions, causes of action, suits, proceedings, demands, rights, damages, costs, expenses or other compensation whatsoever, whether known or unknown, direct or indirect, foreseeable or unforeseeable, absolute or contingent, that Buyer now has or may have or which may arise in the future arising out of, directly or indirectly, or in any way connected with the Property or any other matters relating to the Property, including, without limitation: (A) any condition of environmental contamination or pollution at the Property, however and whenever occurring (including, without limitation, the contamination or pollution of any soils, subsoil media, surface waters or groundwaters at the Property) including, but not limited to, petroleum products, asbestos, lead-based paint, mold or fungi; (B) to the extent not already included in (A) above, the prior, present or future existence, release or discharge, or threatened release, of any Hazardous Materials at the Property, however and whenever occurring; (C) the violation of, or non-compliance with, any applicable law now or hereafter in effect, however or whenever occurring; (D) the condition of the soil at the Property; (E) the condition of Improvements including, without limitation, the structural integrity and seismic compliance of such Improvements; (F) the inaccuracy, unreliability, or incompleteness of, or any defect or mistake in, any items delivered or made available to Buyer pursuant to Section 3.1; (G) to the extent not already covered by any of the foregoing clauses (A) through (F), above, the use, maintenance, development, construction, ownership or operation of the Property by Seller or any of Seller's predecessor(s)-in-interest in the Property; (H) any matters arising from or relating to the leases; or (G) any matters based on Seller's alleged superior knowledge or failure to close.

(b) Without limiting the scope or generality of the foregoing release and waiver provisions, those provisions shall specifically include and cover (1) any claim for or right to indemnification, contribution or other compensation based on or arising under any Environmental Law now or hereafter in effect, and (2) any claim for or based on trespass, nuisance, waste, negligence, negligence per se, strict liability, ultrahazardous activity, indemnification, contribution of other theory arising under the common law of the State of California (or any other applicable jurisdiction) or arising, under any applicable law now or hereafter in effect.

6.3 <u>Waiver of Civil Code § 1542</u>. Buyer hereby acknowledges that it has read and is familiar with the provisions of California Civil Code § 1542 ("Section 1542"), which are set forth below:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING

THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR."

Buyer hereby waives the provisions of Section 1542, and of any statute, principle of common law or case law which would limit the scope of the foregoing waiver and release, in connection with matters which are the subject of the foregoing waiver and release.

Buyer's Initials

6.4 <u>Representations of Seller</u>. As of the date hereof and again as of Closing, Seller represents and warrants to Buyer, as to itself and with respect to its Real Property only, as follows:

(a) Except as disclosed to Buyer in writing prior to Close of Escrow, Seller has not received written notice of, and otherwise has no actual knowledge of, any defaults by any of the parties to any of the leases, contracts or agreements respecting the Property to be assigned to Buyer at Closing.

(b) Except as disclosed to Buyer pursuant to Section 3.1, Seller has not received written notice of (i) any litigation (other than unlawful detainer actions) or condemnation, environmental, zoning or other land-use regulation proceedings, either instituted or planned to be instituted, which would materially and adversely affect the ownership, use, operation or value of the Property, or (ii) any special assessment proceedings affecting the Property. Seller shall notify Buyer promptly of any such litigation or proceedings of which Seller becomes aware.

(c) Seller has not received written notice that removal or other remedial action with respect to Hazardous Materials in, on, under or about the Property is required by any governmental authority having jurisdiction over the Property. For purposes of this Agreement, the term "Hazardous Materials" shall mean any toxic or hazardous waste, material or substance, including, without limitation, asbestos, petroleum, petroleum products, underground storage tanks now or previously containing any other Hazardous Materials, substances defined as "hazardous substances", "hazardous waste" or "toxic substances" in any Environmental Law.

(d) Seller is a limited partnership duly organized and validly existing under the laws of the State of California; this Agreement and all documents executed by Seller which are to be delivered to Buyer at the Closing are and at the time of Closing will be duly executed and delivered by Seller, are and at the time of Closing will be legal, valid and binding obligations of Seller enforceable against Seller in accordance with their respective terms and do not and at the time of Closing will not violate any provision of any agreement or judicial order to which Seller or the Property is subject. Seller has obtained all necessary authorizations, approvals and consents to the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby.

(e) Seller has not granted any option or right of first refusal or first opportunity to any party to acquire any interest in the Property.

Seller is in compliance with the requirements of Executive Order (f) No. 13224, 66 Fed. Reg. 49079 (Sept. 25, 2001) (the "Order") and other similar requirements contained in the rules and regulations of the Office of Foreign Assets Control, Department of the Treasury ("OFAC") and in any enabling legislation or other Executive Orders or regulations in respect thereof (the Order and such other rules, regulations, legislation or orders are collectively hereinafter referred to as the "Orders"). Seller (i) is not listed on the Specially Designated Nationals and Blocked Persons List maintained by OFAC pursuant to the Order and/or on any other list of terrorists or terrorist organizations maintained pursuant to any of the rules and regulations of OFAC or pursuant to any other applicable Orders (such lists are collectively referred to as the "Lists"), (ii) is not a Person (as defined in the Order) who has been determined by competent authority to be subject to the prohibitions contained in the Orders, or (iii) is not owned or controlled by (including without limitation by virtue of such person being a director or owning voting shares or interests), or acts for or on behalf of, any person on the Lists or any other person who has been determined by competent authority to be subject to the prohibitions contained in the Orders. Seller is not acting, directly or indirectly for, or on behalf of, any person, group, entity or nation named by any Executive Order (including the September 24, 2001, Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism) or the United States Treasury Department as a terrorist, "Specially Designated National and Blocked Person," or other banned or blocked person, entity, or nation pursuant to any Law that is enforced or administered by the Office of Foreign Assets Control, and is not engaging in the transactions contemplated herein, directly or indirectly, on behalf of, or instigating or facilitating the transactions contemplated herein, directly or indirectly, on behalf of, any such person, group, entity or nation.

(g) Seller has received no written notice from any governmental authority with jurisdiction over the Property of any current violation by the Property of any laws, ordinances or regulations applicable to the Property, and any such past written notices of past violations have been cured in accordance with law. Seller shall immediately provide Buyer with a copy of any such written notices received after the Effective Date.

For purposes of this Agreement, whenever the phrase "to Seller's actual knowledge" or words of similar import are used, they shall be deemed to refer to the actual knowledge, without obligation for inquiry or investigation, of Michael Massie. If Seller becomes aware of any fact or circumstance which would materially change or render materially incorrect, in whole or in part, any representation or warranty made by Seller pursuant to this Section 6.4, whether as of the date given or at any time thereafter through the Closing Date, and whether or not such representation or warranty was based upon Seller's knowledge as of a certain date, Seller will give prompt written notice of such changed fact or circumstance to Buyer. To the extent Buyer has or acquires knowledge prior to the Closing Date that any of Seller's representations and warranties are inaccurate, untrue or incorrect in any way, such representations and warranties shall be deemed modified to reflect Buyer's knowledge. Buyer shall be deemed to know a representation or warranty is untrue, inaccurate or incorrect if this Agreement or any due diligence items made available to Buyer pursuant to Section 3.1 contain information which is inconsistent with such representation or warranty. Upon receipt by Buyer of Seller's notice of such material change in fact or circumstance, or upon Buyer's obtaining knowledge of the material change in fact or circumstance, Buyer may, as Buyer's sole and exclusive remedy, either (i) terminate this Agreement or (ii) waive the breach, representation or warranty and proceed to Closing, and Seller shall have no further obligation or liability to Buyer with respect thereto. Buyer shall make such election within five (5) days of Buyer's receipt of notice or knowledge of such changed fact or circumstance. The representations and warranties set forth herein shall expire and be of no further force or effect six (6) months following the Closing Date; provided that Buyer shall have commenced an action for breach of the representation or warranty prior to the expiration of such six (6) month period. In no event shall Buyer be permitted to bring an action for breach of a representation or warranty unless the damage to Buyer on account of such breach equal or exceeds Ten Thousand Dollars (\$10,000.00) in the aggregate and in no event shall the maximum total liability for breaches of representations or warranties exceed Twenty-Five Thousand Dollars (\$25,000.00) in the aggregate.

6.5 <u>Buyer's Representations and Warranties</u>. As of the date hereof and as of the Closing Date, Buyer represents and warrants to Seller as follows:

(a) Buyer is a limited partnership, duly organized and validly existing under the laws of the State of California; this Agreement and all documents executed by Buyer which are to be delivered to Seller at the Closing are and at the time of Closing will be duly executed and delivered by Buyer, are and at the time of Closing will be legal, valid and binding obligations of Buyer enforceable against Buyer in accordance with their respective terms and do not and at the time of Closing will not violate any provision of any agreement or judicial order to which Buyer or the Property is subject. Buyer has obtained all necessary authorizations, approvals and consents to the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby.

(b)Buyer is in compliance with the requirements of the Orders and other similar requirements contained in the rules and regulations of OFAC. Buyer (i) is not listed on the Specially Designated Nationals and Blocked Persons List maintained by OFAC pursuant to the Order and/or on any other maintained pursuant to any of the rules and regulations of OFAC or pursuant to any other Lists, (ii) is not a Person (as defined in the Order) who has been determined by competent authority to be subject to the prohibitions contained in the Orders, or (iii) is not owned or controlled by (including without limitation by virtue of such person being a director or owning voting shares or interests), or acts for or on behalf of, any person on the Lists or any other person who has been determined by competent authority to be subject to the prohibitions contained in the Orders. Buyer is not acting, directly or indirectly for, or on behalf of, any person, group, entity or nation named by any Executive Order (including the September 24, 2001, Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism) or the United States Treasury Department as a terrorist, "Specially Designated National and Blocked Person," or other banned or blocked person, entity, or nation pursuant to any Law that is enforced or administered by the Office of Foreign Assets Control, and is not engaging in the transactions contemplated herein, directly or indirectly, on behalf of, or instigating or facilitating the transactions contemplated herein, directly or indirectly, on behalf of, any such person, group, entity or nation.

6.6 <u>Seller's Covenants Pending Closing</u>. During the term of this Agreement, Seller covenants and agrees as follows:

(a) <u>Contracts</u>. Seller will not enter into any agreement or contract, affecting a period after the Closing Date that cannot be terminated on thirty (30) days' notice without the prior written consent of Buyer, which consent may not unreasonably be withheld, conditioned or delayed. The failure of Buyer to respond to a request for approval of an agreement or contract within three (3) business days of Seller's request for approval shall be deemed to constitute Buyer's approval thereof. Seller will notify Buyer in writing of any agreements or contracts it enters into prior to Closing.

(b) <u>Leases</u>. From and after the expiration of the Due Diligence Period (assuming Buyer elects to proceed with the purchase), without the prior written consent of Buyer not to be unreasonably withheld, conditioned or delayed, none of the existing tenant leases will be amended to reduce the rents or other charges thereunder or will be renewed for rents or other charges either of which are less than those payable for similar apartment units in the Property; and no new lease or extension to an existing lease will be made unless it is for a term of not more than one year and at rent not less than that charged for similar apartment units in the Property. Buyer's failure to approve or disapprove the amendment, renewal or new lease within three (3) days of Seller's request shall be deemed to constitute Buyer's approval thereof.

(c) <u>Insurance</u>. Seller will maintain in effect all insurance policies now maintained on the Property, up to and including the Closing Date.

(d) <u>Operation and Condition Pending Closing</u>. Seller will continue to manage, operate and maintain the Property in the same manner as it has before the execution of this Agreement.

VII DAMAGE, DESTRUCTION AND CONDEMNATION

7.1 <u>Risk of Physical Loss</u>. Buyer shall be obligated to acquire the Property pursuant to this Agreement, notwithstanding that the Property is damaged by fire or other casualty prior to the Closing Date, provided that (i) if the cost to repair the Property, in the reasonable judgment of Buyer, can be completed for less than two percent (2%) of the Purchase Price, (ii) the cost to repair such damage is covered by insurance maintained by or for Seller, (iii) Seller assigns to Buyer all insurance proceeds and/or rights to receive insurance proceeds by reason of such damage through Escrow on the Close of Escrow, other than proceeds expended in restoration and repair by Seller, and (iv) Seller credits to the account of Buyer in Escrow, the amount of any deductible under Seller's insurance (not to exceed the cost of repair). Any such casualty is hereinafter referred to as a "non-material insured casualty".

In the event of casualty damage to the Property, other than a non-material insured casualty, Buyer may, at its option, either terminate this Agreement by giving written notice of such termination to Seller within ten (10) days of Buyer's receipt of written notice of such casualty from Seller, or elect to proceed with its purchase of the Property, in which event Seller shall transfer and assign to Buyer all insurance proceeds and all rights to receive insurance proceeds by reason of such damage through Escrow at its Close, other than proceeds expended in restoration and repair by Seller and rental loss proceeds applied to rents accruing through the Close of Escrow and shall credit Buyer's account in Escrow the amount of any deductible under Seller's insurance (not to exceed the cost of repair). If the right to receive any such insurance proceeds to be assigned to Buyer is not assignable by Seller to Buyer, Buyer may nevertheless elect to close the Escrow, in which event Seller shall promptly deliver to Buyer the proceeds of any such insurance received by it following the Close of Escrow, except to the extent such proceeds are in reimbursement for restoration and repair costs incurred or to be incurred by Seller prior to the Close of Escrow and/or rental loss proceeds for rents accruing prior to the Close of Escrow. In the event that such damage shall occur and Buyer elects not to purchase the Property, then this Agreement shall be terminated and Buyer shall be entitled to the return of all funds and documents deposited hereunder.

7.2 <u>Condemnation</u>. In the event that, prior to the Close of Escrow, any governmental entity shall commence any actions of eminent domain or similar type proceedings to take any portion of the Property, Buyer shall have the option either to (i) elect not to acquire the Property, or (ii) complete the acquisition of the Property, in which case Buyer shall be entitled to all the proceeds of such taking.

VIII

MISCELLANEOUS

8.1 <u>Attorneys' Fees</u>. In the event of any action between Buyer and Seller seeking enforcement of any of the terms and conditions to this Agreement, or otherwise in connection with the Property, the prevailing party in such action shall be awarded, in addition to damages, injunctive or other relief, its reasonable costs and expenses, including but not limited to reasonable attorneys' fees, court costs and expert witness fees.

8.2 <u>Notices</u>. All notices under this Agreement shall be effective upon (i) personal delivery to Buyer or Seller, as the case may be, or (ii) telecopier transmission with a hard copy deposited in overnight mail (Express Mail), or (iii) one (1) business day after deposit with an overnight courier service (e.g., Federal Express), (iv) upon delivery by electronic mail; or (v) three (3) business days after deposit in the United States mail, registered, certified, postage fully prepaid and addressed to the respective parties as follows:

| To Seller: | JAMBOREE-TAL HOUSING, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attn: Michael Massie Facsimile: (949) 214-2361 Email: mmassie@jamboreehousing.com |
|-----------------|--|
| With a copy to: | Rutan & Tucker, LLP 18575 Jamboree Road, Suite 900 Irvine, CA 92612 Attn: Patrick D. McCalla |

| | Facsimile: (714) 546-9035 Email: pmccalla@rutan.com |
|-----------------|---|
| To Buyer: | AMISTAD HOUSING PARTNERS I LP c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attn: Michael Massie Facsimile: (949) 214-2361 Email: mmassie@jamboreehousing.com |
| With a copy to: | Rutan & Tucker, LLP 18575 Jamboree Road, Suite 900 Irvine, CA 92612 Attn: Patrick D. McCalla Facsimile: (714) 546-9035 Email: pmccalla@rutan.com |

or to such other address as the parties may from time to time designate in writing.

8.3 <u>Entire Agreement</u>. This Agreement and the items incorporated herein contain all the agreements of the parties hereto with respect to the matters contained herein; and no prior agreement or understanding pertaining to any such matter (including, without limitation, the Letter of Intent) shall be effective for any purpose. No provisions of this Agreement may be amended or modified in any manner whatsoever except by an agreement in writing signed by duly authorized officers or representatives of each of the parties hereto.

8.4 <u>Successors</u>. The terms, covenants and conditions of this Agreement shall be binding upon and shall inure to the benefit of the heirs, executors, administrators and assigns of the respective parties hereto.

8.5 <u>Assignment</u>. Buyer may assign this Agreement and its rights hereunder to an entity in which Buyer owns a majority of the interests in profits and losses without the necessity of acquiring consent from Seller provided such party is designated not later than ten (10) days prior to the Close of Escrow. No other assignment may occur without the consent of Seller, which consent may be withheld in Seller's sole and absolute discretion. In no instance shall an assignment by either Buyer or Seller relieve the assigning party of its obligations hereunder.

8.6 <u>Choice of Laws</u>. This Agreement shall be governed by the laws of the State of California and any questions arising hereunder shall be construed or determined according to such law.

8.7 <u>Headings</u>. Headings at the beginning of each numbered Article and Section of this Agreement are solely for the convenience of the parties and are not a part of this Agreement.

8.8 <u>Survival</u>. This Agreement and all covenants, representations and warranties contained herein shall survive the close of this transaction and this Agreement shall remain a binding contract between the parties hereto.

8.9 <u>Time</u>. Time is of the essence of this Agreement, it being understood that each date set forth herein and the obligations of the parties to be satisfied by such date have been the subject of specific negotiation by the parties. Except as expressly provided for in this Agreement, the time for performance of any obligation or taking any action under this Agreement will be deemed to expire at 5:00 p.m. (California time) on the last day of the applicable time period provided for in this Agreement. If the time for performance of any obligation or taking any action under this Agreement expires on a Saturday, Sunday or California or Federal holiday, the time for performance or taking such action will be extended to the next succeeding day which is not a Saturday, Sunday or California or Federal holiday. The term "business days" shall not include any day which is a Saturday, Sunday or California or Federal holiday.

8.10 <u>Counterparts</u>. This Agreement may be signed by the parties in different counterparts and the signature pages combined to create a document binding on all parties.

8.11 <u>Brokerage Commissions</u>. Each party represents and warrants that it has retained no brokers or finders to represent its interests in connection with this transaction. Each party agrees to indemnify and hold the other harmless from and against all liabilities, costs, damages and expenses, including, without limitation, reasonable attorneys' fees, resulting from any claims or fees or commissions, based upon agreements by it, if any, to pay any broker's commission and/or finder's fee.

8.12 Information Report. Escrow Holder shall file and Buyer and Seller agree to cooperate with Escrow Holder and with each other in completing any report ("Information Report") and/or other information required to be delivered to the Internal Revenue Service pursuant to Internal Revenue Code Section 6045(e) regarding the real estate sales transaction contemplated by this Agreement, including, without limitation, Internal Revenue Service Form 1099-B as such may be hereinafter modified or amended by the Internal Revenue Service, or as may be required pursuant to any regulation now or hereinafter promulgated by the Treasury Department with respect thereto. Buyer and Seller also agree that Buyer and Seller, their respective employees and attorneys, and Escrow Holder and its employees may disclose to the Internal Revenue Service, whether pursuant to such Information Report or otherwise, any information regarding this Agreement or the transaction contemplated herein as such party reasonably deems to be required to be disclosed to the Internal Revenue Service by such party pursuant to Internal Revenue Code Section 6045(e), and further agree that neither Buyer nor Seller shall seek to hold any such party liable for the disclosure to the Internal Revenue Service of any such information.

8.13 <u>Cooperation in 1031 Exchange</u>. The parties each agree to cooperate with the other (at no cost or liability to the agreeing party) in effecting for the benefit of either party a concurrent, delayed or reverse like-kind exchange of real property pursuant to Section 1031 of the United States Internal Revenue Code and similar provisions of applicable state law; provided, however, that (i) neither party shall be obligated to accept title, or to otherwise become part of the chain of title, to an exchange property of the other party; and (ii) the Closing Date shall not be delayed as the result of such exchange (except as provided in Section 2.1). This Agreement and the parties' obligations hereunder are not subject to or conditioned upon ability to consummate an exchange.

Each party's responsibility for reviewing exchange documents shall be limited to determining whether the terms and conditions of such exchange documents are such that they are in compliance with the foregoing provisions. Each party shall be responsible for the payment of all deposits and other costs required to be paid by the "buyer" pursuant to the exchange documents and for making all determinations as to the legal sufficiency or other consideration including, but not limited to, tax considerations, relating to such exchange documents. In so cooperating in any exchange transaction arranged by a party, the non-exchanging party shall in no event be responsible for, or in any way warrant, the tax consequences of the exchange transaction.

8.14 <u>No Third Party Beneficiary</u>. No term or provision of this Agreement is intended to, or shall be, for the benefit of any person not a party hereto and no such person shall have any right or cause of action hereunder.

8.15 <u>Limited Liability</u>. The obligations of Seller under this Agreement and under all of the documents referenced herein are intended to be binding only on the assets of Seller and shall not be personally binding upon, nor shall any resort be had to, the private properties of any member of Seller or any trustee, officer, director, employee or affiliate of Seller or any of its members.

8.16 <u>Jury Waiver</u>. TO THE EXTENT PERMITTED BY LAW, BUYER AND SELLER EACH WAIVES RIGHT TO A JURY IN ANY LITIGATION IN CONNECTION WITH THIS AGREEMENT, OR THE PROPERTY, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. BUYER AND SELLER EACH ACKNOWLEDGES THAT THIS WAIVER HAS BEEN FREELY GIVEN AFTER CONSULTATION BY IT WITH COMPETENT COUNSEL. THE PROVISIONS OF THIS SECTION 8.17 SHALL SURVIVE CLOSING OR ANY EARLY TERMINATION OF THIS AGREEMENT.

8.17 Arbitration. If a dispute arises between Seller and Buyer arising out of or relating to this Agreement or the Property, such dispute shall be determined and resolved by a single arbitrator appointed and acting pursuant to Code of Civil Procedure section 1280, et seq. and the Commercial Arbitration Rules of the American Arbitration Association ("AAA"). Any arbitrator selected under this Section shall be knowledgeable of the subject matter of the dispute. A qualified retired judge with at least five (5) years arbitration experience shall be selected as the arbitrator through panels maintained by the American Arbitration Association, any court in which the property is located or any private organization providing such services. The parties shall have the full right of discovery allowed under Code of Civil Procedure section 1283.05, and the arbitrator may, in the arbitrator's discretion, allow additional discovery. The fees charged by the arbitrator shall be paid equally by Seller and Buyer unless the arbitrator awards such fees to the prevailing party. The venue for the arbitration shall be in Orange County, California. The decision of the arbitrator shall be final and non-appealable and judgment may be entered on the award in accordance with applicable law. Nothing contained herein shall be construed to preclude a party from commencing an action to obtain provisional remedies; however, after granting the provisional relief, the action shall be stayed pending completion of the arbitration proceeding.

8.18 <u>Acceptance of Grant Deed</u>. The acceptance of the Grant Deed by the Buyer shall be deemed full compliance by the Seller of all the Seller's obligations under this Agreement except for those obligations of the Seller which are specifically stated to survive the close of this transaction.

8.19 <u>Bulk Sale Requirements</u>. Seller and Buyer hereby acknowledge and agree that the parties do not intend to comply with and have waived the provisions of any statutory bulk sale requirements applicable to the transaction to be effected by this Agreement, and to rely upon the adjustment provisions of this Agreement to address any matters that would otherwise be subject to such bulk sale requirements.

8.20 <u>Incorporation</u>. The exhibits attached to this Agreement are incorporated herein and made a part hereof.

[Signatures appear on the following page.]

IN WITNESS WHEREOF, Buyer and Seller have executed this Agreement as of the date first above written.

BUYER:

AMISTAD HOUSING PARTNERS I LP, a California limited partnership

- By: JHC-AMISTAD LLC, a California limited liability company, its general partner
 - By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By: ______ Name: Michael Massie Title: Executive Vice President and Chief Development Officer

SELLER:

JAMBOREE-TAL HOUSING, L.P., a California limited partnership

By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its general partner

By: _

Name: Michael Massie Title: Executive Vice President and Chief Development Officer

ACCEPTANCE BY ESCROW HOLDER

The undersigned hereby acknowledges that it has received originally executed counterparts or a fully executed original of the foregoing Purchase and Sale Agreement and Escrow Instructions and agrees to act as Escrow Holder thereunder and to be bound by and perform the terms thereof as such terms apply to Escrow Holder.

Dated: _____, 2024

FIRST AMERICAN TITLE INSURANCE COMPANY

By: _______Ryan Hahn, Senior Escrow Officer

EXHIBIT "A"

LEGAL DESCRIPTION OF REAL PROPERTY

Real property in the City of San Clemente, County of Orange, State of California, described as follows:

LOT 1 OF TRACT NO. 13898, AS SHOWN ON A MAP FILED IN <u>BOOK 817, PAGES 36 TO 40</u> INCLUSIVE OF MISCELLANEOUS MAPS, RECORDS OF ORANGE COUNTY, CALIFORNIA.

EXCEPTING THEREFROM ALL OIL, GAS, AND MINERAL RIGHTS, AS RESERVED BY TALEGA ASSOCIATES, L.L.C., A DELAWARE LIMITED LIABILITY COMPANY, RECORDED AUGUST 8, 2000 AS INSTRUMENT NO. 20000414218 OF OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM ALL OIL, OIL RIGHTS, MINERALS, MINERAL RIGHTS, NATURAL GAS, NATURAL GAS RIGHTS AND OTHER HYDROCARBONS BY WHATSOEVER NAME KNOWN, GEOTHERMAL STEAM, AND ALL PRODUCTS DERIVED FROM ANY OF THE FOREGOING, THAT MAY BE WITHIN OR UNDER THE PARCEL OF LAND HEREINABOVE DESCRIBED, TOGETHER WITH THE PERPETUAL RIGHT OF DRILLING, MINING, EXPLORING AND OPERATING THEREFOR, AND STORING IN AND REMOVING THE SAME FROM SAID LAND OR ANY OTHER LAND, INCLUDING THE RIGHT TO WHIPSTOCK OR DIRECTIONALLY DRILL AND MINE FROM LANDS OTHER THAN THOSE HEREINABOVE DESCRIBED, OIL OR GAS WELLS, TUNNELS AND SHAFTS INTO, THROUGH OR ACROSS THE SUBSURFACE OF THE LAND HEREINABOVE DESCRIBED, AND TO BOTTOM SUCH WHIPSTOCKED OR DIRECTIONALLY DRILLED WELLS, TUNNELS AND SHAFTS UNDER AND BENEATH OR BEYOND THE EXTERIOR LIMITS THEREOF, AND TO REDRILL, RETUNNEL, EQUIP, MAINTAIN, REPAIR, DEEPEN AND OPERATE ANY SUCH WELLS OR MINES, WITHOUT, HOWEVER, THE RIGHT TO DRILL, MINE, STORE, EXPLORE AND OPERATE THROUGH THE SURFACE OR THE UPPER 500 FEET OF THE SUBSURFACE OF THE LAND HEREINABOVE DESCRIBED, AS RESERVED BY TALEGA ASSOCIATES, LLC, IN THE DEED RECORDED AUGUST 01, 2001 AS INSTRUMENT NO. <u>20010525823</u> OF OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM ALL WATER AND WATER RIGHTS, IF ANY, INCLUDING WITHIN AND UNDERLYING SAID LAND, AS RESERVED IN THE SAME DEED.

APN: 701-041-39

EXHIBIT "B"

<u>NOTE</u>

[see attached]

PROMISSORY NOTE SECURED BY SUBORDINATED DEED OF TRUST

\$[2,774,095]

Irvine, California , 2024

FOR VALUE RECEIVED, the undersigned, AMISTAD HOUSING PARTNERS I LP, a California limited partnership ("Borrower") hereby promises to pay to the order of Jamboree-Tal Housing, L.P., a California limited partnership ("Lender"), whose address is 17701 Cowan Ave., Suite 200, Irvine, California 92614 or at such other place as Lender may from time to time designate in writing, the principal sum of \$[2,774,095] or such lesser amount as may be advanced to or for the account of Borrower pursuant to this Promissory Note ("Note), together with all accrued interest on the unpaid balance this Note, and all other amounts now or hereafter owing by Borrower to Lender under this Note. The indebtedness evidenced by this Note is referred to as the "Loan". The Loan is being made by Lender to Borrower in connection with that certain Purchase and Sale Agreement and Escrow Instructions, dated as of , 2024 (the "Purchase Agreement"), pursuant to which Borrower is simultaneously acquiring from Lender certain real property more particularly described in such Purchase Agreement (the "Property"). Borrower shall pay interest at the rates, in the amounts and at the times hereinafter provided. The obligation of Borrower to Lender is set forth in and subject to the terms of this Note and that certain Subordinated Deed of Trust dated as of even date herewith executed by Borrower for the purpose of securing this Note, as the same shall be amended from time to time (the "Subordinated Deed of Trust"). Capitalized terms not otherwise defined herein shall have the meanings set forth Borrower's Limited Partnership Agreement, dated as of August 3, 2023 (the "Partnership Agreement").

1. <u>Principal and Interest</u>. From and after the date hereof through and including the Maturity Date (as hereinafter defined), simple interest shall accrue on the unpaid principal amount outstanding from time to time under this Note at an annual rate equal to [___]% [Insert Term AFR for month of closing] per annum (the "Loan Rate").

2. <u>Payments.</u> Payments under this Note shall be made annually in arrears in immediately available funds for the previous calendar year on the 1st day of May, commencing with May 1 in the year immediately following the year in which the Property is initially occupied in an amount equal to 100% of Residual Receipts for the corresponding year. If the due date for any payment is not a business day, such due date shall be deemed for all purposes to fall on the next business day. As used in this Note, the term "business day" shall mean a day other than a Saturday, Sunday or a day on which banking institutions in the State of California are authorized or required by law to close.

As used herein, the term "Residual Receipts" shall mean Annual Revenue less the sum of:

- (A) Operating Expenses;
- (B) Debt Service;
- (C) Partnership Related Fees/Expenses;

(D) payment of unpaid tax credit adjustment amounts or reimbursement of tax credit adjustment amounts paid by the administrative and/or managing general partners and/or the guarantors to the Property pursuant to the Partnership Agreement, if any;

(E) repayment of loans, if any, made by the partner(s) of Borrower;

(F) Deferred Developer Fee for the Property which remains unpaid; and

(G) repayment of outstanding development and operating loans and/or contributions for capital expenses for which no revenues from operation of the Improvements are available, if any, made by the managing general partner and/or the guarantor to the Property.

"Annual Revenue" shall mean all gross income and all revenues from the operation of the Property in a calendar year received by Borrower determined on the basis of generally accepted accounting principles applied on a consistent basis. Notwithstanding the foregoing, Annual Revenue shall not include the following items: (a) security deposits from tenants (except when applied by Borrower to rent or other amounts owing by tenants); (b) capital contributions to Borrower by its members, partners or shareholders (including capital contributions required to pay any Deferred Developer Fee); (c) condemnation or insurance proceeds; or (d) proceeds of any loan.

"Operating Expenses" shall mean actual costs, fees and expenses attributable to the operation, maintenance, and management of the Property in a calendar year, including, without limitation, painting, cleaning, repairs, alterations, landscaping; utilities, refuse removal, certificates, permits and licenses, sewer charges, taxes, filing fees, assessments, insurance, security, advertising and promotion, janitorial services, cleaning and building supplies, purchase, repair, servicing and installation of appliances, equipment, fixtures and furnishings which are not paid from the capital replacement reserve, fees and expenses of property management and common area expenses, fees and expenses of accountants, attorneys and other professionals, the cost of social services for the tenant households and other actual, reasonable and customary operating costs which are directly incurred and paid by Borrower, but which are not paid from reserve accounts. Operating Expenses shall include costs, fees and expenses paid for the operation of Borrower, including administrative, accounting and legal fees and expenses.

"Debt Service" shall mean payments made in a calendar year pursuant to the financing obtained for the construction, ownership, and operation of the Property permitted pursuant to the Loan Documents, and those certain loans in effect as of the date hereof, as evidenced by the (i) First Deed of Trust (as defined below), (ii) the HCD Deed of Trust (as defined below), (iii) the HOME Deed of Trust, (iv) the OCDA Deed of Trust, (v) the City Deed of Trust, and (vi) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the then-outstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate.

"Partnership Related Fees/Expenses" shall mean the fees and expenses of Borrower payable to the partners in Borrower or affiliates thereof pursuant to the Partnership Agreement. In the event insufficient Annual Revenues exist to provide for payment of all or part of the Partnership Related Fees/Expenses the unpaid balance will accrue.

"Deferred Developer Fee" means the portion of the Developer Fee (as defined in the [Partnership Agreement]) to be paid from cash flow from Annual Revenue.

3. <u>Maturity</u>. The entire unpaid principal balance of this Note, together with all accrued and unpaid interest, shall be due and payable on _____, 20_ [*Insert 58 years from date of Closing*] (the "Maturity Date").

4. <u>Prepayment</u>. The principal of this Note may be voluntarily prepaid at any time, in its entirety or in any partial amount, without penalty.

5. <u>Events of Default</u>. The occurrence of any of the following, whatever the reason therefor, shall constitute an "Event of Default" hereunder:

a. Borrower fails to make any regularly scheduled payment of principal or interest hereunder, to the extent there are Residual Receipts (a "Required Payment"), within thirty (30) days of receipt by Borrower of written notice of such failure; or

b. Borrower fails to pay any other amount owing to Lender under this Note, or fails to perform any other obligation under this Note, the breach of which can be cured by the payment of money, within thirty (30) days after receipt by Borrower of written notice thereof; or

c. Borrower fails to perform any obligation under this Note (other than obligations the breach of which can be cured by the payment of money) within thirty (30) days after notice; *provided, however*, that, if cure cannot reasonably be effected within such thirty (30) day period, there shall be no Event of Default under this subparagraph (c) so long as Borrower promptly commences cure and thereafter diligently prosecutes such cure to completion; or

d. Borrower is dissolved, liquidated or terminated, or all or substantially all of the assets of Borrower are sold or otherwise transferred without Lender's prior consent; or

e. Borrower is the subject of an order for relief by a bankruptcy court, or is unable or admits its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors; or Borrower applies for or consents to the appointment of any rehabilitator or similar officer for it or any part of its property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of Borrower and the appointment continues undischarged or unstayed for one hundred twenty (120) days; or Borrower institutes or consents to any bankruptcy, insolvency, reorganization, arrangement, readjustment or debt, dissolution, custodianship, conservatorship, liquidation, rehabilitation or similar proceeding relating to it or any part of its property; or any similar proceeding is instituted without the consent of Borrower, as the case may be, and continues undismissed or unstayed for one hundred twenty (120) days; or attachment or execution, or similar process is issued or levied against any property of Borrower and is not released, vacated or fully bonded within one hundred twenty (120) days after its issue or levy.

Notwithstanding anything to the contrary contained herein, prior to declaring any default or taking any remedy permitted hereunder or under applicable law based upon an alleged default under this Note, Lender shall give a written notice to the limited partner of Borrower at the notice address set forth herein and the limited partner shall have a period of not less than thirty (30) days to cure such alleged default; provided, however, if in order to cure such default the limited partner reasonably believes that it must remove the general partners of Borrower (or one of them) pursuant to the provisions contained in the Partnership Agreement of Borrower, the limited partner shall so notify Lender and so long as the limited partner is diligently attempting to so remove such general partner(s), the limited partner shall have until the date thirty (30) days after the effective date of the removal of such general partner(s) to cure such default.

6. <u>Limitation on Remedies</u>. Notwithstanding anything to the contrary contained herein or in any other document evidencing or securing the Note, during any time prior to the expiration of the [Compliance Period (as defined in the Partnership Agreement)] in which Lender, or an affiliate of Lender, acts as general partner of Borrower, Lender shall have no right to declare a default hereunder or thereunder, initiate an arbitration pursuant to this Note, obtain a receiver or exercise any other remedy hereunder or under any other document evidencing or securing this Note without first obtaining the prior written consent of the limited partner of Borrower; provided, however, that Lender may bring an action for specific performance to compel any payments due Lender under this Note.

7. <u>Acceleration and Other Remedies</u>. Upon the occurrence of any Event of Default, the holder of this Note may, at its option, declare the outstanding principal balance of this Note, together with all accrued interest thereon, to be immediately due and payable. All rights and remedies provided to the holder of this Note are cumulative and shall be in addition to all other rights and remedies of the holder under other documents, at law or in equity, and all such rights and remedies may be exercised singly, successively and/or concurrently. Failure to exercise any right or remedy shall not be deemed a waiver of such right or remedy.

8. <u>Due on Sale; Due on Encumbrance</u>. Notwithstanding anything to the contrary contained herein, Lender may declare all obligations hereunder immediately due and payable if Borrower, without Lender's prior written consent, sells, conveys, leases with an option to purchase or further encumbers any interest in or any part of the Property. Any such sale, conveyance, transfer, pledge, lease or encumbrance made without Lender's prior written consent shall be void. Borrower shall not, without the prior written consent of Lender, assign the rents from the Property and any such assignment without the express written consent of Lender shall be null and void.

9. <u>General Partner Change</u>. The withdrawal, removal, and/or replacement of a general partner of the Partnership pursuant to the terms of the Partnership Agreement shall not constitute a default under this Note, and any such actions shall not accelerate the maturity of the Loan, provided that any required substitute general partner is reasonably acceptable to Lender and is selected with reasonable promptness.

10. <u>Purchase Rights</u>. The execution and delivery of the purchase option and right of first refusal agreement described in the Partnership Agreement shall not constitute a default under this Note or accelerate the maturity of the Loan. Any requisite consent of Lender to (a) the exercise of said purchase option and right of first refusal agreement by the Managing General Partner and to (b) the assumption without penalty of loan obligations by the Managing General Partner and the release of Borrower from such obligations, shall not be unreasonably withheld. Subject to any such consent requirement, the exercise of rights under such agreement shall not constitute a default or accelerate maturity of the Loan.

11. <u>Lender Approvals, Etc</u>. In any approval, consent, or other determination by Lender required under this Note, Lender shall act reasonably and in good faith.

12. <u>Application of Payments</u>. All payments made hereunder shall be applied first to interest due and then in reduction of the principal balance. If more than one annual payment

remains unpaid, amounts received shall be applied to such payments in reverse chronological order.

13. <u>Expenses/Attorneys' Fees</u>. If this Note is not paid when due, whether at each scheduled payment date, at maturity or by acceleration, Borrower promises to pay all costs of collection, including, but not limited to, reasonable attorneys' fees and all expenses incurred in connection with the protection or realization of the collateral securing the payment hereof or enforcement of any guarantee, incurred by Lender on account of such collection, whether or not suit is filed hereon.

14. <u>Waivers</u>. Except as expressly set forth in this Note, Borrower hereby waives presentment, demand, protest, notice of dishonor, notice of protest and all other notices and demands of every kind that would otherwise be available in connection with this Note.

15. <u>Notice</u>. All notices and other communications provided for in this Note shall be in writing and be delivered by messenger, overnight air courier or registered or certified U.S. mail with return receipt requested to the appropriate party at its address as follows:

| If to Lender: | Jamboree-Tal Housing, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Michael Massie |
|-----------------|---|
| If to Borrower: | Amistad Housing Partners I LP c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Michael Massie |

Addresses for notice may be changed from time to time by written notice to the other party. Communications by mail shall be effective upon the earlier of (a) 96 hours after deposit in the U.S. mail with postage prepaid, (b) one day after deposit with a reputable national overnight courier service, or (c) actual receipt, as indicated by the return receipt; and communications by messenger shall be effective when delivered.

16. <u>Miscellaneous</u>. Time is of the essence hereof. All payments hereunder shall be made in lawful money of the United States of America. This Note shall be governed by the laws of the State of California. Neither this Note nor any term hereof may be waived, amended, discharged, modified, changed or terminated orally; nor shall any waiver of any provision hereof be effective except by an instrument in writing signed by Borrower and Lender. No delay or omission on the part of Lender in exercising any right hereunder shall operate as a waiver of such right or of any other right under this Note.

17. <u>Subordination</u>. The indebtedness evidenced by this Note is and shall be subordinate in right of payment to the prior payment in full of all amounts then due and payable (including, but not limited to, all amounts due and payable by virtue of any default or acceleration or upon maturity) with respect to (i) that certain _______, dated as of the date hereof, by Borrower in favor of First American Title Insurance Company, a California corporation, as trustee, for the benefit of _______, a ______, as beneficiary (the "First Deed of Trust"), (ii) that deed of trust relating to the loan from the Department of Housing

and Community Development, a public agency of the State of California, or its affiliates, encumbering the Property (collectively, the "HCD Deed of Trust"), (iii) that deed of trust relating to the loan from the County of Orange, or its affiliates, encumbering the Property (collectively, the "HOME Deed of Trust"), (iv) that deed of trust relating to the loan from the Orange County Development Agency, or its affiliates, encumbering the Property (collectively, the "OCDA Deed of Trust"), (v) that deed of trust relating to the loan from the City of San Clemente, or its affiliates, encumbering the Property (collectively, the "City Deed of Trust"), and (vi) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the then-outstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate. Lender further agrees that upon the maturity of any debt or obligation senior in priority to the Subordinated Deed of Trust, Lender shall subordinate this Note and the Subordinated Deed of Trust to any new debt which is used to partially or fully repay said maturing senior debt or obligation.

18. <u>Non-Recourse</u>. Neither Borrower nor Borrower's officers, partners, directors, employees or agents nor their respective members, managers, officers, partners, shareholders, directors, employees or agents shall have personal liability for the payment or performance of any of Borrower's obligations under this Note or the Subordinated Deed of Trust. Lender's sole recourse shall be to realize against the collateral described in the Subordinated Deed of Trust.

Executed this ____ day of _____ 2024.

BORROWER:

AMISTAD HOUSING PARTNERS I LP, a California limited partnership

- By: JHC-AMISTAD LLC, a California limited liability company, its general partner
 - By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By:

Name: Michael Massie Title: Executive Vice President and Chief Development Officer

EXHIBIT "C"

DEED OF TRUST

[see attached]

RECORDING REQUESTED BY AND WHEN RECORDED, RETURN TO:

Jamboree-Tal Housing, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Asset Management

SUBORDINATED DEED OF TRUST

THIS SUBORDINATED DEED OF TRUST ("Subordinated Deed of Trust") is made on this ______day of ______, 2024 by ______, a _____, as trustor ("Borrower") to First American Title Insurance Company ("Trustee"), for the benefit of Jamboree-Tal Housing, L.P., a California limited liability company, as beneficiary ("Lender").

Borrower, in consideration of the indebtedness recited and the trust created in this Subordinated Deed of Trust, irrevocably grants and conveys to Trustee, in trust, with power of sale certain real property owned in fee by Borrower and located in the County of Orange, State of California ("Property"). The Property is legally described in <u>"Exhibit A"</u> attached hereto and made a part hereof.

Together with all of the improvements now or hereafter erected on said property, and all easements, rights, appurtenances, rents (subject, however, to the rights and authorities given in this Subordinated Deed of Trust to Lender to collect and apply such rents), royalties, mineral, oil and gas rights and profits, water, water rights, and water stock, and all fixtures now or hereafter attached to said property, all of which including replacements and additions thereto, shall be deemed to be and remain a part of the property covered by this Subordinated Deed of Trust; and all of the foregoing, together with said property are referred to in this Subordinated Deed of Trust as the "Property";

To secure for the benefit of Lender the repayment of the indebtedness evidenced by Borrower's Promissory Note Secured by Subordinated Deed of Trust of even date herewith (the "Note") in the principal sum not to exceed \$[2,774,095] and the performance of the covenants and agreements of Borrower contained in this Subordinated Deed of Trust and in the Note.

Borrower covenants and agrees as follows:

1. PAYMENT OF PRINCIPAL AND INTEREST. Borrower shall pay when due the interest and the principal of the indebtedness evidenced by the Note from the sources and pursuant to the terms of the Note.

2. COMPLIANCE WITH SENIOR DEED OF TRUST. Borrower covenants and agrees to comply with the terms and conditions of any senior deeds of trust recorded against the Property (collectively, the "Senior Encumbrances").

3. CHARGES AND LIENS. Borrower shall pay all taxes, assessments and other charges, fines and impositions attributable to the Property which may attain a priority over this Subordinated Deed of Trust. Borrower shall promptly furnish to Lender all notices of amounts

due under this paragraph and in the event Borrower shall make payment directly, Borrower shall promptly furnish to Lender receipts evidencing such payments. Borrower shall promptly discharge any lien which has priority over this Subordinated Deed of Trust; provided that Borrower shall not be required to discharge any such lien so long as Borrower shall agree in writing to the payment of the obligation secured by such lien in a manner acceptable to Lender, or shall in good faith contest such lien by, or defend enforcement of such lien in, legal proceedings which operate to prevent the enforcement of the lien or forfeiture of the Property or any part thereof. Borrower shall not allow the attachment of any subordinate lien or other encumbrance on the Property without the prior written consent of Lender.

Borrower shall promptly pay when due all installments or payments required by Lender and comply with all obligations of any deed of trust that is prior to this Subordinated Deed of Trust.

4. INSURANCE PROCEEDS. Any insurance proceeds received as a result of damage to the Property from fire or other calamity shall be first expended as required by any Senior Encumbrances.

5. PROTECTION OF LENDER'S SECURITY. If Borrower fails to perform the covenants and agreements contained in this Subordinated Deed of Trust, or if any action or proceeding is commenced which materially affects Lender's interest in the Property, including, but not limited to, default on or foreclosure of a prior deed of trust, eminent domain, insolvency, code enforcement, or arrangement or proceedings involving a bankrupt or decedent, then Lender at Lender's option, upon notice to Borrower, may make such appearances, disburse such sums and take such action as is necessary to protect Lender's interest including, but not limited to, disbursement of reasonable attorney's fees and entry upon the Property to make repairs, but shall have no right of reimbursement.

6. INSPECTION. Lender may make or cause to be made reasonable entries upon and inspections of the Property, provided that Lender shall give Borrower notice prior to any such inspection specifying reasonable cause therefor related to Lender's interest in the Property.

7. CONDEMNATION. Subject to the rights of the holder of a prior deed of trust, the proceeds of any award or claim for damages, direct or consequential, in connection with any condemnation or other taking of the Property, or part thereof, or for conveyance in lieu of condemnation, are hereby assigned and shall be paid to Lender.

In the event of a total taking of the Property, the proceeds shall be treated as if received from a sale of the Property. The proceeds shall be applied to the sums to be repaid in the amount and manner described in the Senior Encumbrances and the documents evidencing the indebtedness secured thereby. Any part of the proceeds remaining after these amounts have been paid shall be paid to Borrower. In the event of a partial taking of the Property, unless Borrower and Lender agree in writing, the proceeds shall be treated as if received from a sale of that portion of the Property which is taken in the condemnation. The percentage value of the portion taken, as compared to the full value of the entire Property, shall be determined by dividing the condemnation proceeds by the fair market value of the entire Property just prior to the taking. This percentage value, once determined, shall be used in the following manner to allocate the condemnation proceeds:

a. First, to the payment of any amount owing under the Senior Encumbrances and the documents evidencing the indebtedness secured thereby;

- b. Second, to payment of the Note;
- c. Third, to Borrower.

8. BORROWER NOT RELEASED. Extension of the time for payment or modification of amortization of the sums secured by this Subordinated Deed of Trust granted by Lender to any successor in interest of Borrower shall not operate to release, in any manner, the liability of the original Borrower and Borrower's successors in interest. Lender shall not be required to commence proceedings against such successor or refuse to extend time for payment or modification of amortization of the sums secured by this Subordinated Deed of Trust by reason of any demand made by the original Borrower and Borrower and Borrower's successors in interest.

9. FORBEARANCE BY LENDER NOT A WAIVER. Any forbearance by Lender in exercising any right or remedy hereunder, otherwise afforded by applicable law, shall not be a waiver of or preclude the exercise of any such right or remedy. The procurement of insurance or the payment of taxes or other liens or charges by Lender shall not be a waiver of Lender's right to accelerate the maturity of the indebtedness secured by this Subordinated Deed of Trust.

10. REMEDIES CUMULATIVE. All remedies provided in this Subordinated Deed of Trust are distinct and cumulative to any other right or remedy under this Subordinated Deed of Trust or afforded by law or equity, and may be exercised concurrently, independently or successively.

11. SUCCESSORS AND ASSIGNS BOUND; JOINT AND SEVERAL LIABILITY; CAPTIONS. The covenants and agreements contained in this Subordinated Deed of Trust shall bind, and the rights hereunder shall inure to, the respective successors and assigns of Lender and Borrower, subject to the provisions of Section 15 hereof. All covenants and agreements of Borrower shall be joint and several. The captions and headings of the Sections of this Subordinated Deed of Trust are for convenience only and are not to be used to interpret or define the provisions hereof.

12. NOTICE. Except for any notice required under applicable law to be given in another manner, (a) any notice to Borrower provided for in this Subordinated Deed of Trust shall be given by facsimile, a nationally recognized overnight courier service, or by mailing such notice by certified mail addressed to Borrower at 17701 Cowan Ave., Suite 200, Irvine, California 92614, Attention: President, or at such other address as Borrower may designate by notice to Lender as provided in this Section 12, and (b) any notice to Lender shall be given by facsimile, a nationally recognized overnight courier service, or by mailing such notice by certified mail, return receipt requested, to Lender's address stated in this Subordinated Deed of Trust or to such other address as Lender may designate by notice to Borrower as provided in this Section 12. Any notice provided for in this Subordinated Deed of Trust shall be deemed to have been given to Borrower or Lender upon receipt or refusal of delivery by the addressee.

13. GOVERNING LAW; SEVERABILITY. This Subordinated Deed of Trust shall be governed by the laws of the State of California. In the event that any provision or clause of this Subordinated Deed of Trust or the Note as incorporated herein conflicts with applicable law, such conflict shall not affect other provisions of this Subordinated Deed of Trust or the Note which can be given effect without the conflicting provision, and to this end the provisions of the Subordinated Deed of Trust and Note are declared to be severable.

14. BORROWER'S COPY. Borrower shall be furnished a conformed copy of the Note and of this Subordinated Deed of Trust at the time of execution or after recordation hereof.

15. TRANSFER OF THE PROPERTY. If all or any of the Property or an interest in it is sold or transferred by Borrower, excluding permitted transfers pursuant to this Subordinated Deed of Trust, the Note or the Partnership Agreement (as defined in the Note), all the sums secured by this Subordinated Deed of Trust shall be immediately due and payable. In the event of such an acceleration, Lender shall mail Borrower notice of the acceleration in accordance with Section 12 hereof. Such notice shall provide a period of not less than 30 days from the date the notice is mailed within which Borrower may pay the sums declared due. If Borrower fails to pay such sums prior to the expiration of such period, Lender may, without further notice or demand on Borrower, invoke any remedies permitted by Section 16 hereof.

ACCELERATION; REMEDIES Except as provided in Section 15 hereof, upon 16. Borrower's breach of any covenant or agreement of Borrower in this Subordinated Deed of Trust or the Note as incorporated by reference in this Subordinated Deed of Trust, including the covenants to pay, when due, any sums secured by this Subordinated Deed of Trust, Lender, prior to acceleration, shall mail notice to Borrower as provided in Section 12 hereof specifying: (a) the breach; (b) the action required to cure such breach; (c) a date by which such breach must be cured, which date shall be extended by Lender so long as Borrower or investor limited partner of Borrower is diligently prosecuting a cure of such breach; and (d) that failure to cure such breach on or before the date specified in the notice may result in acceleration of the sums secured by this Subordinated Deed of Trust and sale of the Property. If the breach is not cured on or before the date specified in the notice, Lender, at its option, may declare all of the sums determined by this Subordinated Deed of Trust to be immediately due and payable without further demand and may invoke the power of sale and any other remedies permitted by applicable law. Lender shall be entitled to collect all reasonable costs and expenses incurred in pursuing the remedies provided in this Section 16, including, but not limited to, reasonable attorneys' fees.

Notwithstanding anything to the contrary contained in this Subordinated Deed of Trust, the Note, or any other document evidencing or securing the Note, during the time prior to the Maturity Date (as defined in the Note) that Lender or any affiliate of Lender acts as a general partner of Borrower, Lender shall have no right to declare a default hereunder or thereunder, foreclose on the Project, accelerate the maturity of any amounts due, obtain a receiver or exercise any other remedy hereunder or under any other document evidencing or securing said note without first obtaining the prior written consent of Borrower, the investor limited partner of Borrower, and the beneficiary of the Senior Encumbrances; provided however, that Lender may bring an action for specific performance to compel any payments due Borrower pursuant to Borrower's Partnership Agreement.

Subject to the terms and conditions provided herein, if Lender invokes the power of sale, Lender shall have appraised, or shall cause Trustee to have appraised, the Property to determine its fair market value. Lender shall also execute or cause Trustee to execute a written notice of the occurrence of an event of default and of Lender's election to cause the Property to be sold, and shall cause such notice to be recorded in each county in which the Property or some part thereof is located. Lender or Trustee shall mail copies of such notice in a manner prescribed by applicable law. After the lapse of such time as may be required by applicable law, Trustee, without demand on Borrower, shall sell the Property at public auction to the highest bidder at the time and place and under the terms designated in the notice of sale in one or more parcels and in such order as Trustee may determine. Trustee may postpone sale of all or any parcel of the Property by public announcement at the time and place of any previously scheduled sale. Lender or Lender's designee may purchase the Property at any sale.

Trustee shall deliver to the purchaser Trustee's deed conveying the Property so sold without any covenant or warranty, expressed or implied. The recitals in the Trustee's deed shall be prime facie evidence of the truth of the statements made therein. Trustee shall apply the proceeds of the sale in the following order: (a) to all reasonable costs and expenses of the sale, including, but not limited to, reasonable Trustee and attorneys' fees and costs of title evidence; (b) to all sums secured by the lien of some other or later security instrument, as evidenced by a note, by and between Borrower and Borrower's lender, that is either prior in lien priority to the priority of this Subordinated Deed of Trust or to which Lender has subordinated its lien rights and, thereby, placed itself in a subordinated position, (c) to all sums secured by this Subordinated Deed of Trust, as evidenced by the Note; and (d) the excess, if any, to the person or persons legally entitled thereto.

BORROWER'S RIGHT TO REINSTATE. Notwithstanding Lender's acceleration 17. of the sums secured by this Subordinated Deed of Trust, Borrower shall have the right to have any proceedings begun by Lender to enforce covenants of this Subordinated Deed of Trust relating to sums due and payable by Borrower discontinued at any time prior to five days before the Property is scheduled for sale pursuant to the power of sale contained in this Subordinated Deed of Trust or at any time prior to entry of a judgment enforcing this Subordinated Deed of Trust if: (a) Borrower pays Lender all sums which would be then due under this Subordinated Deed of Trust and the Note, had no acceleration occurred; (b) Borrower pays all reasonable expenses incurred by Lender and Trustee in enforcing the covenants and agreements of Borrower contained in this Subordinated Deed of Trust and in enforcing Lender's and Trustee's remedies as provided in Section 16 hereof, including, but not limited to reasonable attorneys' fees; (c) Borrower takes such actions as Lender may reasonably require to assure that the lien of this Subordinated Deed of Trust, Lender's interest in the Property, and Borrower's obligation to pay the sums secured by this Subordinated Deed of Trust shall continue unimpaired. Upon such payment and cure by Borrower, this Subordinated Deed of Trust and the obligations secured hereby shall remain in full force and effect as if no acceleration had occurred.

18. APPOINTMENT OF RECEIVER; LENDER IN POSSESSION. Upon acceleration under Section 16 hereof or abandonment of the Property, Lender, in person, by agent or by judicially appointed receiver, shall be entitled to enter upon, take possession of, rent and manage the Property and to collect the rents of the Property. All rents collected by Lender or by a judicially appointed receiver shall be applied first to payment of the cost of management of the Property and collection of rents, including, but not limited to, receiver's fees, premiums on receiver's bonds and reasonable attorneys' fees, and then to the sums secured by this Subordinated Deed of Trust. Lender and the receiver shall be liable to account for those rents actually received and expenditures actually incurred.

19. RECONVEYANCE. Upon payment of all sums evidenced by the Note and secured by this Subordinated Deed of Trust, Trustee shall reconvey the Property without warranty and without charge to the person or persons legally entitled thereto. Such person or persons shall pay all costs of recordation, if any.

20. SUBSTITUTE TRUSTEE. Lender, at Lender's option, may from time to time remove Trustee and appoint a successor trustee to act as Trustee under this Subordinated Deed of Trust. Without conveyance of the Property, the successor trustee shall succeed to all the title,

power and duties conferred upon the Trustee in this Subordinated Deed of Trust and by applicable law.

21. REQUEST FOR NOTICES. Borrower requests that copies of the notice of default and notice of sale be sent to Borrower's address which is the Property address.

22. STATEMENT OF OBLIGATION. Lender may collect a fee not to exceed \$15.00 for furnishing the statement of obligation as provided by Section 2943 of the Civil Code of the State of California.

23. SUBORDINATION ACKNOWLEDGMENT. Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is subordinate in right of payment to the prior payment in full of all amounts then due and payable (including, but not limited to, all amounts due and payable by virtue of any default or acceleration or upon maturity) with respect to (i) that certain , dated as of the date hereof, by Borrower in favor of First American Title Insurance Company, a California corporation, as trustee, for the benefit of __, as beneficiary (the "First Deed of Trust"), (ii) that а deed of trust relating to the loan from the Department of Housing and Community Development, a public agency of the State of California, or its affiliates, encumbering the Property (collectively, the "HCD Deed of Trust"), (iii) that deed of trust relating to the loan from the County of Orange, or its affiliates, encumbering the Property (collectively, the "HOME Deed of Trust"), (iv) that deed of trust relating to the loan from the Orange County Development Agency, or its affiliates, encumbering the Property (collectively, the "OCDA Deed of Trust"), (v) that deed of trust relating to the loan from the City of San Clemente, or its affiliates, encumbering the Property (collectively, the "City Deed of Trust"), and (vi) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the thenoutstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate. Without limiting the generality of the foregoing, Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is also subordinate to the extended use agreement required to be executed by Borrower pursuant to Section 42(h)(6)(B) of the Internal Revenue Code, for purposes of the low-income housing tax credits to be allocated to Borrower. In addition, Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is further subordinate to Section 42(h)(6)(e)(ii) of the Internal Revenue Code, which prohibits the eviction or termination of a tenancy, other than for good cause, of an existing tenant of any low-income housing tax credit unit or any increase in the gross rent with respect to such unit, not otherwise permitted under Section 42, for a period of three (3) years after the date the Property is acquired by Lender through foreclosure or instrument in lieu of foreclosure.

24. NON-RECOURSE. This is a non-recourse Subordinated Deed of Trust. Neither Borrower, nor any of its partners shall have any personal liability for the payment of any portion of the indebtedness evidenced by the Note or performance of Borrower's obligations under this Subordinated Deed of Trust. In the event of a default by Borrower under the terms of the Note or this Subordinated Deed of Trust, the Beneficiary's sole remedy shall be limited to exercising its rights under the Note and this Subordinated Deed of Trust, including foreclosure and the exercise of the power of sale or other rights granted under the Note and this Subordinated Deed of Trust, but shall not include a right to proceed directly against Borrower, or any of its partners, or the right to obtain a deficiency judgment after foreclosure against Borrower or any of its partners.

[SIGNATURES ON FOLLOWING PAGE.]

IN WITNESS THEREOF. Borrower has executed this Subordinated Deed of Trust on the day and year first above written.

BORROWER:

AMISTAD HOUSING PARTNERS I LP, a California limited partnership

- By: JHC-AMISTAD LLC, a California limited liability company, its general partner
 - By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By: _____ Name: Michael Massie Title: Executive Vice President and Chief Development Officer

A Notary Public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.

State of California County of _____

Ś

On _____, before me, _____ (insert name and title of the officer)

Notary Public, personally appeared

who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf of which the person(s) acted, executed the instrument.

I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.

WITNESS my hand and official seal.

Signature

(Seal)

EXHIBIT "A"

LEGAL DESCRIPTION OF THE PROPERTY

Real property in the City of San Clemente, County of Orange, State of California, described as follows:

LOT 1 OF TRACT NO. 13898, AS SHOWN ON A MAP FILED IN BOOK 817, PAGES 36 TO 40 INCLUSIVE OF MISCELLANEOUS MAPS, RECORDS OF ORANGE COUNTY, CALIFORNIA.

EXCEPTING THEREFROM ALL OIL, GAS, AND MINERAL RIGHTS, AS RESERVED BY TALEGA ASSOCIATES, L.L.C., A DELAWARE LIMITED LIABILITY COMPANY, RECORDED AUGUST 8, 2000 AS INSTRUMENT NO. 20000414218 OF OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM ALL OIL, OIL RIGHTS, MINERALS, MINERAL RIGHTS, NATURAL GAS, NATURAL GAS RIGHTS AND OTHER HYDROCARBONS BY WHATSOEVER NAME KNOWN, GEOTHERMAL STEAM, AND ALL PRODUCTS DERIVED FROM ANY OF THE FOREGOING, THAT MAY BE WITHIN OR UNDER THE PARCEL OF LAND HEREINABOVE DESCRIBED, TOGETHER WITH THE PERPETUAL RIGHT OF DRILLING, MINING, EXPLORING AND OPERATING THEREFOR, AND STORING IN AND REMOVING THE SAME FROM SAID LAND OR ANY OTHER LAND, INCLUDING THE RIGHT TO WHIPSTOCK OR DIRECTIONALLY DRILL AND MINE FROM LANDS OTHER THAN THOSE HEREINABOVE DESCRIBED, OIL OR GAS WELLS, TUNNELS AND SHAFTS INTO, THROUGH OR ACROSS THE SUBSURFACE OF THE LAND HEREINABOVE DESCRIBED, AND TO BOTTOM SUCH WHIPSTOCKED OR DIRECTIONALLY DRILLED WELLS, TUNNELS AND SHAFTS UNDER AND BENEATH OR BEYOND THE EXTERIOR LIMITS THEREOF, AND TO REDRILL, RETUNNEL, EQUIP, MAINTAIN, REPAIR, DEEPEN AND OPERATE ANY SUCH WELLS OR MINES, WITHOUT, HOWEVER, THE RIGHT TO DRILL, MINE, STORE, EXPLORE AND OPERATE THROUGH THE SURFACE OR THE UPPER 500 FEET OF THE SUBSURFACE OF THE LAND HEREINABOVE DESCRIBED, AS RESERVED BY TALEGA ASSOCIATES, LLC, IN THE DEED RECORDED AUGUST 01, 2001 AS INSTRUMENT NO. 20010525823 OF OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM ALL WATER AND WATER RIGHTS, IF ANY, INCLUDING WITHIN AND UNDERLYING SAID LAND, AS RESERVED IN THE SAME DEED.

APN: 701-041-39

ATTACHMENT Q

PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS

by and between

JAMBOREE-TAL HOUSING, L.P., a California limited partnership

("Seller")

and

AMISTAD HOUSING PARTNERS I LP, a California limited partnership

("Buyer")

TABLE OF CONTENTS

Page

| I | AGDI | EEMENT TO PURCHASE | <u>1 « C</u> |
|-----|----------------|--|--------------|
| 1 | 1.1 | The Property | |
| | 1.1 | Purchase Price | |
| | 1.2 | | |
| | - | Carryback Financing | |
| | 1.4 | Assumed Loans | <i>L</i> |
| II | ESCR | .OW | 3 |
| | 2.1 | Escrow Holder | 3 |
| | 2.2 | Closing Date | 3 |
| | 2.3 | Escrow Instructions | |
| | 2.4 | Conveyance of Title | 3 |
| | 2.5 | Additional Closing Obligations of Seller | |
| | 2.6 | Closing Obligations of Buyer | |
| | 2.7 | Delivery of Documents by Escrow Holder | |
| | 2.1 | Derivery of Documents by Escrow Holder | ·····) |
| III | INSPI | ECTIONS, REVIEW | 9 |
| | 3.1 | Access to Books and Records; Items to be Inspected and Reviewed by | |
| | | Buyer | |
| | 3.2 | Physical Inspection | |
| | 3.3 | Preliminary Title Report | 10 |
| IV | CONI | DITIONS TO AGREEMENT | 11 |
| | 4.1 | Buyer's Conditions Precedent | |
| | 4.2 | Seller's Conditions Precedent | |
| | 4.3 | Satisfaction of Conditions | |
| | 4.4 | Waiver | |
| | 4.5 | Termination | |
| | | | |
| V | | RATIONS, POSSESSION AND DEPOSITS | |
| | 5.1 | Proration of Taxes | |
| | 5.2 | Proration of Rents | 14 |
| | 5.3 | Tenants' Deposits | 14 |
| | 5.4 | Utilities | 14 |
| | 5.5 | Maintenance Contracts | 14 |
| | 5.6 | Method of Adjustment | 14 |
| | 5.7 | Insurance | 14 |
| | 5.8 | Possession | 15 |
| | 5.9 | Closing Costs | 15 |
| | 5.10 | Closing Statement | |
| | 5.11 | BREACH BY SELLER | |
| VI | AS-IS PURCHASE | | 17 |
| V I | 6.1 | Disclaimer | |
| | 6.1 6.2 | Release and Waiver | |
| | - | | |
| | 6.3 | Waiver of Civil Code § 1542 | 18 |

Page

| | 6.4 | Representations of Seller | . 19 |
|------|--|--|------|
| | 6.5 | Buyer's Representations and Warranties | |
| | 6.6 | Seller's Covenants Pending Closing | |
| VII | /II DAMAGE, DESTRUCTION AND CONDEMNATION | | 22 |
| | 7.1 | Risk of Physical Loss | |
| | 7.2 | Condemnation | |
| VIII | MISC | ELLANEOUS | 23 |
| | 8.1 | Attorneys' Fees | |
| | 8.2 | Notices | |
| | 8.3 | Entire Agreement | |
| | 8.4 | Successors | |
| | 8.5 | Assignment | 24 |
| | 8.6 | Choice of Laws | |
| | 8.7 | Headings | 24 |
| | 8.8 | Survival | 25 |
| | 8.9 | Time | 25 |
| | 8.10 | Counterparts | 25 |
| | 8.11 | Brokerage Commissions | 25 |
| | 8.12 | Information Report | |
| | 8.13 | Cooperation in 1031 Exchange | |
| | 8.14 | No Third Party Beneficiary | |
| | 8.15 | Limited Liability | |
| | 8.16 | Jury Waiver | |
| | 8.17 | Arbitration | 26 |
| | 8.18 | Acceptance of Grant Deed | 27 |
| | 8.19 | Bulk Sale Requirements | |
| | 8.20 | Incorporation | |
| | | - | |

PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS

THIS PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS ("Agreement") is made as of this _____ day of _____, 2024 (the "Effective Date") by and between JAMBOREE-TAL HOUSING II, L.P., a California limited partnership ("Seller"), and AMISTAD HOUSING PARTNERS II LP, a California limited partnership ("Buyer").

$\underline{R} \, \underline{E} \, \underline{C} \, \underline{I} \, \underline{T} \, \underline{A} \, \underline{L} \, \underline{S}$

A. Seller is the owner of that certain real property (the "Real Property") located in the City of San Clemente (the "City"), County of Orange, State of California, legally described on <u>Exhibit "A"</u>, which Real Property is improved with a 62 unit affordable housing development project.

B. Buyer wishes to purchase and Seller wishes to sell the Real Property, including the buildings, parking areas, improvements and fixtures now or hereafter located thereon (the "Improvements"), on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of \$100.00 paid by Buyer to Seller and in consideration of the mutual covenants, promises and undertakings set forth herein, Buyer and Seller hereby agree as follows:

I AGREEMENT TO PURCHASE

1.1 <u>The Property</u>. Subject to all of the terms, conditions and provisions of this Agreement, and for the consideration herein set forth, Seller hereby agrees to sell and Buyer hereby agrees to buy the following:

(a) Seller's fee estate in and to the Real Property, together with the Improvements and all tenements, hereditaments and appurtenances thereto;

(b) All furniture, personal property, machinery, apparatus and equipment which is owned by Seller and is located on the Real Property (the "Personal Property");

(c) All signs, logos, trade names or styles relating to the Real Property, if any, owned by Seller and to be assigned to Buyer pursuant to Section 2.5(k) hereof;

(d) To the extent assignable, all labor, service, supply and maintenance contracts relating to the Improvements or the Real Property which are to be assumed by Buyer pursuant to Section 3.2 below;

(e) All leases, lease deposits, and prepaid rentals related to the Real Property or the Improvements; and

(f) Any and all transferable guaranties and warranties covering the Real Property, the Improvements or any part thereof, if any, owned by Seller and which are

assignable by Seller, in each case, excluding any cash and any Personal Property of third parties.

The Real Property and the other property described in Paragraphs (a) through (f) above are hereinafter collectively referred to as the "Property".

1.2 <u>Purchase Price</u>. The purchase price which Seller agrees to accept and Buyer agrees to pay for the Property is the sum of NINE MILLION FIFTY THOUSAND DOLLARS (\$9,050,000) ("Purchase Price"). At Closing, Buyer shall receive a credit against the Purchase Price in the amount of \$239,677 for the existing replacement and operating reserves (the "Reserve Credit").

1.3 <u>Carryback Financing</u>. In connection with the sale of the Property, Seller has agreed to make a loan to Buyer in the current estimated amount of One Million One Hundred Eighty Five Thousand Five Hundred Eleven Dollars (\$1,185,511) (the "Seller Loan"), which comprises the sum of the Purchase Price adjusted for closing costs and prorations, less the amounts assumed by Buyer pursuant to Section 1.4 below. To evidence the Loan, Buyer shall execute and deliver a Promissory Note Secured by Subordinated Deed of Trust in the form of <u>Exhibit "B"</u> attached hereto (the "Note") and a Subordinated Deed of Trust in the form of <u>Exhibit "C"</u> attached hereto (the "Seller Deed of Trust"). The Seller Deed of Trust shall be recorded against the Real Property and against Buyer's leasehold interest in certain other real property at Closing. Notwithstanding the foregoing, Buyer and Seller may agree to adjust the amount of the Seller Loan at or prior to Closing. The provisions of this Section shall survive the Closing.

1.4 <u>Assumed Loans</u>. Buyer shall assume:

(a) <u>OCDA Loan</u>. That certain loan from the Orange County Development Agency ("OCDA"), or its affiliates, encumbering the Property in the original principal amount of Seven Hundred Twenty Eight Thousand One Hundred Seventy Six Dollars (\$728,176), plus accrued interest in the current estimated amount of Three Hundred Forty Five Thousand Seven Hundred Sixty Seven Dollars (\$345,767), for a total outstanding amount of One Million Seventy Three Thousand Nine Hundred Forty Dollars (\$1,073,940), pursuant to a Loan Agreement dated June 11, 2004, between County and Seller (the "OCDA Loan").

(b) <u>City Loan</u>. That certain loan from the City of San Clemente ("City"), or its affiliates, encumbering the Property in the original principal amount of Three Hundred Fifty One Thousand Five Hundred Forty Dollars (\$351,540), pursuant to an Affordable Housing Project Agreement (Phase 2) dated June 6, 2001, between the City and Seller (the "City Loan").

Π

ESCROW

2.1 <u>Escrow Holder</u>. Closing of the sale of the Property shall take place through an escrow ("<u>Escrow</u>") to be established with First American Title Insurance Company ("Escrow Holder"), 18500 Von Karman Avenue, Suite 600, Irvine, CA 92612, Attention: Ryan Hahn. Escrow shall be deemed open upon delivery of a fully executed copy of this Agreement to Escrow

Holder. Escrow Holder shall immediately notify Buyer and Seller of the date of Opening of Escrow and the escrow number.

2.2 <u>Closing Date</u>. The "Closing Date" shall be on or before June 17, 2024. The terms "Close of Escrow" and "Closing" shall mean the date the Grant Deed (hereafter defined) is either filed for record or, subject to the following sentence, sent for recording in the Official Records of Orange County, California ("Official Records"). The parties agree to effectuate the Closing through a so-called "gap" closing and Seller agrees to provide a customary gap indemnity to the Title Company (as hereinafter defined).

2.3 <u>Escrow Instructions</u>. Articles 1, 2, 4, 5, 7 and 8 also constitute escrow instructions to Escrow Holder. Additionally, Buyer and Seller agree to execute any form of escrow instructions as the Escrow Holder customarily requires as escrow holder in real property escrows administered by it. In the event of a conflict between any such additional terms and provisions of this Agreement, this Agreement shall supersede and be controlling.

2.4 <u>Conveyance of Title</u>. At the Closing Date, Seller shall deliver to Escrow Holder a duly executed and acknowledged Grant Deed ("Grant Deed") in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period. The Grant Deed shall convey all of Seller's interest in fee title to the Real Property to Buyer at Closing. Escrow Holder is hereby instructed to record the Grant Deed in the Official Records, if and when Escrow Holder holds instruments and funds accruing to Buyer and Seller and can obtain for Buyer a CLTA standard coverage owner's policy of title insurance without endorsement issued by First American Title Insurance Company ("Title Company") with liability in an amount equal to the Purchase Price for the Property, showing the Real Property vested in Buyer, subject only to:

- (a) The Seller Deed of Trust;
- (b) Non-delinquent real property taxes and assessments;
- (c) Title exceptions approved by Buyer pursuant to Section 3.3 below;

(d) Title exceptions, if any, resulting from documents being recorded or delivered through Escrow (including, without limitation, any documents related to or arising from any new financing arranged by Buyer);

(e) Rights of tenants in possession, as tenants only;

(f) Any of the labor, service, supply and maintenance contracts referenced in Section 1.1(d) above;

(g) Discrepancies, conflicts in boundary lines, shortages in area, encroachments and any state of facts shown on the survey delivered to the Title Company;

(h) The standard preprinted exceptions and exclusions to the CLTA standard coverage owner's policy of title insurance;

(i) The Deed of Trust, Assignment of Rents and Security Agreement recorded on July 24, 2002, in the Official Records as Instrument Number No. 20020612337 (as amended or modified from time to time, the "OCDA Deed of Trust") relating to the OCDA Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(j) The Deed of Trust recorded on March 31, 2003, in the Official Records as Instrument Number No. 2003000347377 (as amended or modified from time to time, the "City Deed of Trust") relating to the City Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(k) The Master Declaration of Covenants, Conditions, Restrictions and Reservation of Easements for Talega recorded on September 24, 1999, in the Official Records as Instrument Number 19990686286 (as amended or modified from time to time, the "Master Declaration") that Buyer has agreed to assume;

(1) The Talega Lifestyle Fee Agreement recorded June 28, 2002, in the Official Records as Instrument Number 20020548429 (as amended or modified from time to time, the "Talega Lifestyle Fee Agreement") that Buyer has agreed to assume;

(m) The Regulatory Agreement and Declaration of Covenants and Restrictions recoded on June 28, 2002, in the Official Records as Instrument Number 20020548433 (as amended or modified from time to time, the "City Regulatory Agreement") that Buyer has agreed to assume;

(n) The Regulatory Agreement and Declaration of Restrictive Covenants recorded on July 24, 2002, in the Official Records as Instrument Number 20020612336 (as amended or modified from time to time, the "Bond Regulatory Agreement") that Buyer has agreed to assume;

(o) The Regulatory Agreement and Declaration of Restrictive Covenants recorded on June 21, 2004, in the Official Records as Instrument Number 2004000555674 (as amended or modified from time to time, the "OCDA Regulatory Agreement") relating to the OCDA Loan that Buyer has agreed to assume, which will not be paid off and released as of the Closing Date;

(p) That certain Regulatory Agreement recorded on December 02, 2004, in the Official Records as Instrument Number 2004001071788 (as amended or modified from time to time, the "TCAC Regulatory Agreement") that Buyer has agreed to assume; and

(q) The Declaration of Development Covenants, Conditions and Restrictions Concerning Lot 2 of Tract 13898 recorded on June 28, 2002, in the Official Records as Instrument Number 20020548427 (as amended or modified from time to time, the "Talega Declaration") that Buyer has agreed to assume.

2.5 <u>Additional Closing Obligations of Seller</u>. On or before 12:00 noon on the business day preceding the Closing Date, Seller shall deliver to Escrow Holder (unless indicated to be delivered directly to Buyer) copies of the following documents and other items:

(a) Two (2) duplicate copies of a Bill of Sale conveying the Personal Property related to its Real Property to Buyer in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period, duly executed by such Seller;

(b) Two (2) duplicate copies of an Assignment of Leases by Seller to Buyer (the "Assignment of Leases") in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period, duly executed by such Seller;

(c) A Certificate of Non-Foreign Status (the "Non-Foreign Affidavit") duly executed by such Seller in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period and a California Franchise Tax Board Real Estate Withholding Certificate duly executed by such Seller;

(d) One (1) original of an assignment and assumption of the OCDA Loan and OCDA Deed of Trust ("Assignment of OCDA Loan Documents") in a form approved by Buyer, Seller and OCDA duly executed and acknowledged by the Seller;

(e) One (1) original of an assignment and assumption of the OCDA Regulatory Agreement in a form approved by Buyer, Seller and OCDA duly executed and acknowledged by Seller ("Assignment of OCDA Regulatory Agreement");

(f) One (1) original of an assignment and assumption of the City Loan and City Deed of Trust ("Assignment of City Loan Documents") in a form approved by Buyer, Seller and City duly executed and acknowledged by the Seller;

(g) One (1) original of an assignment and assumption of the City Regulatory Agreement ("Assignment of City Regulatory Agreement") in a form approved by Buyer, Seller and City duly executed and acknowledged by Seller;

(h) One (1) original of an assignment and assumption of the TCAC Regulatory Agreement ("Assignment of TCAC Regulatory Agreement") in a form approved by Buyer and Seller duly executed and acknowledged by Seller;

(i) One (1) original of an assignment and assumption of the Bond Regulatory Agreement ("Assignment of Bond Regulatory Agreement") in a form approved by Buyer and Seller duly executed and acknowledged by Seller;

(j) Originals or copies of all assumable contracts relating to the Property which Buyer is to assume pursuant to Section 3.2 hereof (to be delivered to Buyer at the Property);

(k) An assignment to Buyer of Seller's right, title and interest in and to (i) all contracts which Buyer is to assume pursuant to Section 3.2 below; (ii) all assignable permits, licenses and certificates of occupancy relating to the Property; (iii) all trade names, logos, signs, trademarks, telephone listings and numbers and similar items in Seller's possession included within the Property; and (iv) all warranties and guaranties then in effect, if any, with respect to the Improvements in a form reasonably approved by Buyer and Seller prior to the expiration of the Due Diligence Period ("Assignment of Contracts"), duly executed by such Seller;

(1) All soils, seismic, geologic, drainage, toxic waste, and environmental reports, surveys, "as-built" plans and specifications, working drawings, grading plans, elevations and similar information with respect to its Real Property heretofore obtained by such Seller which such Seller has in its possession and/or control to the extent such items have not been delivered previously by such Seller to Buyer pursuant to Section 3.1 below (to be delivered to Buyer at the Property);

(m) All keys to the Improvements which such Seller or such Seller's agents have in their possession (to be delivered to Buyer at the Property);

(n) The original executed copy of each lease of the Property (to the extent that Seller has the original leases in its possession or control), including any amendments thereto (to be delivered to Buyer at the Property);

(o) Two (2) duplicate copies of the Closing Statement (as hereinafter defined) duly executed by such Seller;

(p) A notice to each of the tenants of the Property in compliance with the provisions of California Civil Code Section 1950.5(h) (to be delivered to each tenant outside of Escrow); and

(q) Any other documents, instruments and records required to be delivered to Buyer under the terms of this Agreement which have not been previously delivered.

2.6 <u>Closing Obligations of Buyer</u>. On or before 12:00 noon on the business day preceding the Closing Date, Buyer shall deliver to Escrow Holder copies of the following documents and other items:

(a) One (1) original of all documents required by HCD in connection with the Buyer's HCD Loan (collectively, the "HCD Documents");

(b) One (1) original of the Assignment of OCDA Loan Documents, duly executed and acknowledged by Buyer;

(c) One (1) original of the Assignment of OCDA Regulatory Agreement, duly executed and acknowledged by Buyer;

(d) One (1) original of the Assignment of City Loan Documents, duly executed and acknowledged by Buyer;

(e) One (1) original of the Assignment of City Regulatory Agreement, duly executed and acknowledged by Buyer;

(f) One (1) original of the Assignment of TCAC Regulatory Agreement, duly executed and acknowledged by Buyer;

(g) One (1) original of the Assignment of Bond Regulatory Agreement, duly executed and acknowledged by Buyer;

(h) One (1) original of the Note, duly executed by Buyer;

(i) One (1) original of the Seller Deed of Trust, duly executed and acknowledged by Buyer;

(j) Two (2) duplicate copies of the Assignment of Leases, duly executed by Buyer;

(k) Two (2) duplicate copies of the Assignment of Contracts, duly executed by Buyer;

(1) Two (2) duplicate copies of the Closing Statement, duly executed by Buyer;

(m) Evidence of the existence, organizational authority of Buyer and of the authority of persons executing documents on behalf of Buyer reasonably satisfactory to Escrow Holder and the Title Company; and

(n) Any other documents, instruments or funds required to be delivered by Buyer under the terms of this Agreement or otherwise required by Escrow Holder or Title Company in order to close Escrow which have not previously been delivered.

2.7 <u>Delivery of Documents by Escrow Holder</u>. All of the items listed in Section 2.5 above shall be delivered to Buyer through Escrow on the Closing Date (unless indicated to the contrary), except that Escrow Holder is hereby instructed to record the original Grant Deed, Seller Deed of Trust, HCD Documents (as applicable), Assignment of OCDA Loan Documents, Assignment of OCDA Regulatory Agreement, Assignment of City Loan Documents, Assignment of City Regulatory Agreement, Assignment of the TCAC Regulatory Agreement and Assignment of Bond Regulatory Agreement (collectively, the "Recordable Documents") in the Official Records upon Close of Escrow in the order approved by Buyer and Seller and to deliver the original Recordable Documents to Seller. Escrow Holder shall deliver to Seller one (1) duplicate copy of all of the items listed in Section 2.6 on the Close of Escrow and conformed copies of the Recordable Documents.

III INSPECTIONS, REVIEW

3.1 Access to Books and Records; Items to be Inspected and Reviewed by Buyer. Within five (5) days following the execution of this Agreement by Buyer and Seller, Seller shall make available to Buyer for Buyer's inspection at the Property or at Seller's property manager's offices during normal business hours upon not less than forty-eight (48) hours' prior notice, copies of all books, records, reports, surveys, bills, labor, service, supply and maintenance contracts, copies of all leases, rental applications and lease files, books and records of Seller relating to the Property in Seller's possession which are reasonably requested by Buyer. Seller has not made any independent investigation to determine the truth or accuracy of any such documents and materials and shall have no liability to Buyer for any inaccuracy, misrepresentation or omission in such documents or materials. Notwithstanding anything to the contrary contained herein, Seller shall have no obligation to deliver to Buyer (i) those documents, reports, presentations, summaries and the like prepared for any of Seller's partners or investors in connection with its consideration of the acquisition or sale of the Property, (ii) Seller's attorney-client privileged materials, (iii) appraisals, or (iv) any information which is subject to a confidentiality agreement between Seller and a third party. In addition, prior to the expiration of the Due Diligence Period (as hereinafter defined), Seller shall deliver to Buyer a lead-based paint disclosure statement in the form required by applicable law and a natural hazards disclosure statement in the form prescribed by California Civil Code § 1103.2 prepared by a licensed professional described in California Civil Code § 1103.4(c).

3.2 Physical Inspection. From and after the date hereof through for a period of thirty (30) days (the "Due Diligence Period"), Seller shall permit Buyer, its engineers, analysts, contractors, agents and consultants, at the sole cost and expense of Buyer, to conduct physical inspections of the Property and Improvements, including the structural, electrical and mechanical aspects of the Improvements to the Property, supports, site work, foundations, soil, subsurface soils, drainage, seismic and other geological and topographical matters, location of asbestos, leadbased paint, mold, fungi, toxic substances, hazardous materials or wastes, if any, and any other investigations as Buyer deems prudent with respect to the physical condition of the Property. Such investigations may be made by Buyer and/or its agents during any normal business hours. Buyer shall also have the right to investigate all matters relating to the zoning, use and compliance with other applicable laws which relate to the use and occupancy of the Property. Seller shall reasonably cooperate with Buyer in completing such inspections and special investigations at no cost or expense to Seller. Such inspections and investigations shall be conducted only upon no less than forty-eight (48) hours' notice to Seller and shall be conducted at such times and in such a manner as to minimize any disruption to tenants upon the Property. Seller shall have the right, but not the obligation, to accompany Buyer during such investigations and/or inspections. Notwithstanding the foregoing, Buyer shall not perform any invasive or destructive testing of the Property without the prior written consent of Seller, which consent may be withheld in Seller's sole and absolute discretion. Buyer shall repair any and all damage to the Property or to any tenants' property caused by such inspections or investigations in a timely manner and shall indemnify, defend (with counsel acceptable to Seller) and hold harmless Seller and its partners, affiliates, officers, employees, agents and consultants (the "Indemnified Parties") from and against any liability whatsoever sought by the Indemnified Parties arising from or related to Buyer's physical inspection of the Property or access to the Property granted hereunder to Buyer, its engineers, analysts, contractors, agents and consultants, including any such liability to which the negligence of the indemnified parties may have contributed but excepting any liability to the extent arising from the gross negligence or willful misconduct of the indemnified parties. As a condition to allowing Buyer, or its engineers, analysts, contractors, agents or consultants access to the Property, Buyer shall obtain and deliver to Seller a certificate of insurance evidencing liability insurance policy in an amount not less than TWO MILLION DOLLARS (\$2,000,000.00) and naming Seller as an additional insured. Prior to the expiration of the Due Diligence Period, if, in Buyer's sole and absolute discretion, Buyer is dissatisfied with the results of its review of the documents related to the Property or the inspections or investigations of the Property, then Buyer may terminate this Agreement by giving written notice of termination to Seller and Escrow Holder. In addition, Buyer shall, not later than the expiration of the Due Diligence Period, advise Seller in writing of the labor, service, supply and maintenance contracts which it elects to assume; provided, however, that Buyer shall be obligated to assume those contracts which cannot be terminated by Seller without payment of a fee or penalty. If Buyer notifies Seller of its disapproval of its due diligence investigation of the Property on or before the expiration of the Due Diligence Period, this Agreement and the escrow established hereunder shall terminate.

Preliminary Title Report. Within five (5) days of the Effective Date, Buyer shall 3.3 obtain from Title Company a preliminary title report describing the state of title of the Property (the "Preliminary Title Report"). Buyer shall notify Seller in writing ("Buyer's Objection Notice") of any objections Buyer may have to title exceptions contained in the Preliminary Title Report not later than seven (7) business days prior to the expiration of the Due Diligence Period ("Title Review Period"). In the event Buyer fails to deliver Buyer's Objection Notice prior to the expiration of the Title Review Period, Buyer shall conclusively be deemed to have approved the state of title of the Property. Seller shall have a period of five (5) business days after receipt of Buyer's Objection Notice in which to deliver written notice to Buyer ("Seller's Notice") of Seller's election to either (i) agree to remove the objectionable items prior to the Close of Escrow, or (ii) decline to remove any such title exceptions and terminate Escrow and this Agreement. If Seller fails to deliver Seller's notice within said five (5) business day period, Seller shall be deemed to have elected to decline to remove such title exceptions and terminate Escrow and this Agreement. If Seller notifies or is deemed to have notified Buyer of its election to terminate Escrow rather than remove the objectionable items, Buyer shall have the right, by written notice delivered to Seller not later than the expiration of the Due Diligence Period to agree to accept the Property subject to the objectionable items, in which event Seller's election to terminate the Escrow shall be of no effect, and Buyer shall take title at the Close of Escrow subject to such objectionable items without any adjustment to or credit against the Purchase Price. Notwithstanding the foregoing, except as set forth herein, Seller shall, on or before the Close of Escrow, remove (i) all deeds of trust and financing statements, including, without limitation, that certain Department of Housing and Community Development Multifamily Housing Program NOFA of October 25, 2001 Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing (Permanent Financing) recorded on June 21, 2004, in the Official Records as Instrument Number 2004000555668 (collectively, "Monetary Liens"), and (ii) that certain Department of Housing and Community Development Multifamily Housing Program (NOFA 10/25/01) recorded on June 21, 2004, in the Official Records as Instrument Number 2004000555667, as encumbrances to title, without the necessity for Buyer to object thereto. Upon the issuance of any amendment or supplement to the Preliminary Title Report which has additional exceptions for matters not shown on the original Preliminary Title Report, the foregoing right of review and approval shall also apply to said amendment or supplement (provided that the period for Buyer to review such amendment or supplement shall be the later of the expiration of the Title Review Period or five (5) days from receipt of the amendment or supplement). If Buyer properly notifies Seller of its disapproval of the state of title to the Property on or before the expiration of the Title Review Period (or such longer period as may be provided in the immediately sentence), this Agreement and the escrow established hereunder shall terminate.

IV

CONDITIONS TO AGREEMENT

4.1 <u>Buyer's Conditions Precedent</u>. Buyer's obligation to purchase the Property shall be conditioned upon the fulfillment of the following conditions precedent, all of which shall be satisfied or waived in writing pursuant to Section 4.4 below prior to the Close of Escrow except as indicated otherwise:

(a) Seller has delivered to Escrow Holder all of the documents required to be delivered under Section 2.5.

(b) The due performance by Seller of all material obligations to be performed by Seller hereunder and the truth in all material respects of the representations and warranties made by Seller pursuant to Section 6.4.

(c) No order or injunction of any court or administrative agency of competent jurisdiction nor any statute, rule, regulation or executive order promulgated by any governmental authority of competent jurisdiction shall be in effect as of the Close of Escrow which restrains or prohibits the transfer of the Property or the consummation of any other transaction contemplated hereby.

(d) The Department of Housing and Community Development, a public agency of the State of California ("HCD") has approved of a new loan to Buyer (the "Buyer's HCD Loan").

(e) OCDA has approved of the transfer of the Property to Buyer and the Assignment of the OCDA Loan Documents and Assignment of OCDA Regulatory Agreement.

(f) The City has approved of the transfer of the Property to Buyer and the Assignment of the City Loan Documents and Assignment of City Regulatory Agreement.

(g) TCAC has approved of the transfer of the Property to Buyer and the Assignment of TCAC Regulatory Agreement.

4.2 <u>Seller's Conditions Precedent</u>. Seller's obligation to convey the Property to Buyer shall be conditioned upon the satisfaction or written waiver in writing, in whole or in part by Seller of the following conditions precedent:

(a) Buyer has delivered to Escrow Holder all of the documents required to be delivered under Section 2.6.

(b) The due performance by Buyer of all material obligations to be performed by Buyer hereunder and the truth in all material respects of the representations and warranties made by Buyer pursuant to Section 6.5.

(c) No order or injunction of any court or administrative agency of competent jurisdiction nor any statute, rule, regulation or executive order promulgated by any governmental authority of competent jurisdiction shall be in effect as of the Close of Escrow which restrains or prohibits the transfer of the Property or the consummation of any other transaction contemplated hereby.

(d) HCD has approved the Buyer's HCD Loan.

(e) OCDA has approved of the transfer of the Property to Buyer and the Assignment of the OCDA Loan Documents and the Assignment of the OCDA Regulatory

Agreement, and the release of Seller from liability under the OCDA Loan Documents and the OCDA Regulatory Agreement for matters first arising after the Closing Date.

(f) The City has approved of the transfer of the Property to Buyer and the Assignment of the City Loan Documents and the Assignment of the City Regulatory Agreement, and the release of Seller from liability under the City Loan Documents and the City Regulatory Agreement for matters first arising after the Closing Date.

(g) TCAC has approved of the transfer of the Property to Buyer and the Assignment of the TCAC Regulatory Agreement, and the release of Seller from liability under the TCAC City Regulatory Agreement for matters first arising after the Closing Date.

4.3 <u>Satisfaction of Conditions</u>. Where satisfaction of any of the foregoing conditions requires action by Buyer or Seller, each party shall use its diligent best efforts, in good faith, and at its own cost, to satisfy such condition; provided, however, that in no event shall Seller be required to expend money related to Buyer's financing or in order to satisfy any condition of Buyer.

4.4 <u>Waiver</u>. Buyer may at any time or times, at its election, waive any of the conditions set forth in Section 4.1 above to its obligations hereunder, but any such waiver shall be effective only if contained in a writing signed by Buyer and delivered to Seller. Seller may at any time or times, at its election, waive any of the conditions set forth in Section 4.2 above to its obligations hereunder, but any such waiver shall be effective only if contained in a writing signed by Seller and delivered to Buyer.

4.5 Termination. In the event each of the conditions set forth in Section 4.1 is not fulfilled within the time provided in Section 4.1 or waived by Buyer pursuant to Section 4.4, Buyer may, at its option, terminate this Agreement and the Escrow opened hereunder, thereby releasing the parties from further obligations hereunder, and all documents delivered by Buyer to Seller or Escrow Holder shall be returned to Buyer and all documents delivered by Seller to Buyer or Escrow Holder returned to Seller. In the event that the conditions set forth in Section 4.2 are not fulfilled or waived by Seller pursuant to Section 4.4 prior to the Closing Date, Seller may, at its option, terminate this Agreement and the Escrow opened hereunder, thereby releasing the parties from further obligations hereunder, and all documents delivered by Seller to Buyer or Escrow Holder shall be returned immediately to Seller. In the event this Agreement terminates for any reason pursuant to this Section 4.5, within five (5) days of the date of termination, Buyer shall deliver to Seller all due diligence materials delivered to Buyer pursuant to Section 3.1 above as well as true, correct and complete copies of all reports, studies, documents or surveys generated in connection with Buyer's due diligence investigation of the Property without representation or warranty as to the accuracy or completeness of such documents.

V

PRORATIONS, POSSESSION AND DEPOSITS

5.1 <u>Proration of Taxes</u>. Real and personal property taxes and assessments for the Property shall be prorated by the parties based on a 365-day year as of 11:59 p.m. on the Closing Date with Seller paying such taxes on the day of Closing. All tax prorations shall be based upon

the latest available tax statement. If the tax statements for the fiscal tax year during which Escrow closes do not become available until after the Closing Date, the parties shall re-prorate said taxes and assessments outside of Escrow following the Closing Date when such tax statements become available. Seller shall be responsible for and shall pay or reimburse Buyer upon demand for any real or personal property taxes payable following the Closing Date with respect to its Real Property applicable to any period of time prior to the Closing Date as a result of any change in the tax assessment by reason of reassessment, changes in use of the Property, changes in ownership, errors by the Assessor or otherwise. Any tax refunds which relate to any period of time prior to Close of Escrow shall accrue to the benefit of Seller.

5.2 <u>Proration of Rents</u>. Rents of the tenants under the leases shall be prorated on a 365day year as of 11:59 p.m. in on the Closing Date based on rents actually collected with Seller receiving the rents on the day of Closing. Any such rents collected after the Closing Date by Buyer which are attributable to the period prior to the Closing Date shall be paid to Seller upon collection. Rents collected after the Closing Date from tenants whose rental was delinquent at Closing shall be deemed to first apply to costs of collecting such rents, second to rentals which were delinquent at Closing and third to rents due at the time of payment accruing after the Closing Date. Buyer shall have no obligation to commence litigation or to collect rents or to terminate the tenant's right to occupancy based upon tenant's failure to pay rentals which were delinquent at Closing; however, Buyer shall use reasonable efforts to collect such delinquent rents and shall reasonably and in good faith cooperate with Seller's attempts to collect such rents at no cost or expense to Buyer.

5.3 <u>Tenants' Deposits</u>. The security deposits and other tenant deposits (to the extent not previously applied by Seller) shall be credited to Buyer at the Closing. From and after the Closing, Buyer hereby assumes the obligations of the landlord under the leases with respect to the security deposits and any other tenant deposits. This provision shall survive the closing.

5.4 <u>Utilities</u>. Seller shall use its best efforts to have utility meters read as of the date that Escrow closes and shall be responsible for all utility services to the Property until the Closing Date. In the event Seller is unable to have the utility meters read as of the date Escrow closes, Buyer and Seller shall jointly prepare and deposit an estimated utility statement based upon the average daily usage on the most recent utility bill preceding the Close of Escrow with Seller paying the utility costs for the day of Closing. Escrow Holder shall initially prorate utilities based upon such estimated utility statements and the parties shall subsequently prorate utilities based upon the actual utility usage upon receipt of such utility statements.

5.5 <u>Maintenance Contracts</u>. Seller shall be responsible for payment of all maintenance services, such as janitorial services, guard services and similar services through the Closing. Buyer shall be responsible for such services thereafter for any such contracts which it assumes pursuant to Section 3.2 above.

5.6 <u>Method of Adjustment</u>. The obligations of Sections 5.1, 5.2, 5.3, 5.4, and 5.5 shall be handled by Buyer and Seller outside of Escrow once a final accounting of all figures have been obtained. Seller and Buyer shall complete all adjustments for items to be prorated pursuant to this Article V within the earlier of ninety (90) days after the Closing Date or promptly after the receipt of actual bills or verifiable information relating to the item to be adjusted. 5.7 <u>Insurance</u>. Seller shall cause its policies of insurance for the Property to be terminated effective immediately after the Closing Date and Buyer shall be responsible for obtaining its own insurance.

5.8 <u>Possession</u>. Subject to the rights of tenants in possession, Buyer shall be entitled to possession of the Property on the Closing Date.

5.9 <u>Closing Costs</u>. Buyer shall pay all costs in connection with the closing of this transaction, including the premium for a policy of title insurance, any county or city documentary or transfer taxes payable on account of the conveyance of the Real Property, the escrow fees which may be charged by Escrow Holder, the cost of any survey required by the Title Company to issue its title policy, the cost of recording the Recordable Documents, any and all costs and expenses associated with or arising from Buyer's new financing.

5.10 <u>Closing Statement</u>. No later than three (3) business days prior to the Closing Date, Escrow Holder shall prepare for approval by Buyer and Seller a closing statement ("Closing Statement") on Escrow Holder's standard form indicating, among other things, the Replacement Reserve Credit, the Short Term Work Credit and Escrow Holder's estimate of all closing costs and prorations made pursuant to this Agreement. Buyer and Seller shall assist Escrow Holder in determining the amount of all prorations.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

BREACH BY SELLER. IN THE EVENT THAT SELLER BREACHES ANY 5.11 OF ITS OBLIGATIONS UNDER THIS AGREEMENT WHETHER PRIOR TO OR AFTER THE CLOSE OF ESCROW, BUYER'S SOLE REMEDY SHALL BE THE RIGHT TO SUE FOR ACTUAL DAMAGES (NOT TO EXCEED \$25,000), AND THIS AGREEMENT SHALL TERMINATE AND, EXCEPT AS SPECIFICALLY PROVIDED FOR ELSEWHERE IN THIS AGREEMENT, NEITHER PARTY SHALL HAVE ANY FURTHER OBLIGATION OR LIABILITY TO THE OTHER AT LAW OR IN EQUITY. IN NO EVENT SHALL SELLER EVER BE LIABLE TO BUYER UNDER ANY STATUTORY, COMMON LAW, EQUITABLE OR OTHER THEORY OF LAW EITHER PRIOR TO OR FOLLOWING THE CLOSING FOR ANY LOST RENTS, PROFITS, **BUSINESS OPPORTUNITIES OR ANY FORM OF CONSEQUENTIAL OR PUNITIVE** DAMAGES IN CONNECTION WITH ANY CLAIM, LIABILITY, DEMAND OR CAUSE OF ACTION IN ANY WAY OR MANNER RELATING TO THE PROPERTY, THE CONDITION OF THE PROPERTY, THIS AGREEMENT, OR ANY TRANSACTION OR MATTER BETWEEN THE PARTIES CONTEMPLATED HEREUNDER. BUYER HEREBY WAIVES, RELEASES AND FOREVER RELINQUISHES ANY RIGHT BUYER MAY NOW HAVE OR MAY LATER ACOUIRE TO SEEK SPECIFIC PERFORMANCE, TO PLACE ANY LIENS OR ENCUMBRANCES, INCLUDING A NOTICE OF LIS PENDENS. ON THE PROPERTY.

Buyer's Initials

Seller's Initials

a. LIOUIDATED DAMAGES. IN THE EVENT THAT CLOSING FAILS TO OCCUR DUE TO A DEFAULT OF BUYER, THE DAMAGES THAT SELLER WILL INCUR BY REASON THEREOF ARE AND WILL BE IMPRACTICAL AND EXTREMELY DIFFICULT TO ESTABLISH. BUYER AND SELLER, IN A **REASONABLE EFFORT TO ASCERTAIN WHAT SELLER'S DAMAGES WOULD BE** IN THE EVENT OF EITHER SUCH A DEFAULT BY BUYER, HAVE AGREED THAT SUCH DAMAGES SHALL BE IN AN AMOUNT EQUAL TO \$25,000, AND THAT SUCH AMOUNT SHALL BE DELIVERED TO SELLER UPON SUCH DEFAULT BY BUYER OR FAILURE OF CONDITION AND RETAINED BY SELLER AS LIOUIDATED DAMAGES, WHICH DAMAGES SHALL BE SELLER'S SOLE AND EXCLUSIVE **REMEDY AT LAW OR IN EOUITY IN THE EVENT OF AND FOR SUCH DEFAULT BY** WITHOUT LIMITING THE FOREGOING PROVISIONS OF THIS BUYER. PARAGRAPH, SELLER WAIVES ANY AND ALL RIGHTS WHICH SELLER **OTHERWISE WOULD HAVE HAD UNDER CALIFORNIA CIVIL CODE SECTION** 3389 TO SPECIFICALLY ENFORCE THIS AGREEMENT. SELLER AND BUYER ACKNOWLEDGE AND AGREE THAT THEY HAVE READ AND UNDERSTAND THE PROVISIONS OF THIS PARAGRAPH AND BY THEIR INITIALS AGREE TO BE **BOUND BY ITS TERMS.**

Buyer's Initials

Seller's Initials

AS-IS PURCHASE

6.1 Disclaimer. Buyer acknowledges and agrees that it has been provided the opportunity to thoroughly inspect, investigate and exercise due diligence, and the opportunity to fully and independently become familiar with, and fully satisfy itself regarding, any and all matters relating to the Property. Except as expressly provided in Section 6.4 of this Agreement, Seller has not made, does not make and specifically negates and disclaims any representations, warranties, promises, covenants, agreements or guaranties of any kind or character whatsoever, whether express or implied or arising by operation of law, oral or written, past, present or future, of, as to, concerning or with respect to (i) the value of the Property; (ii) the income to be derived from the Property; (iii) the suitability of the Property for any and all activities and uses which Buyer may conduct thereon, including, without limitation, the possibilities for future development of the Property; (iv) the habitability, merchantability, marketability, profitability or fitness for a particular purpose of the Property; (v) the manner, quality, state of repair or lack of repair of the Property; (vi) the nature, quality or condition of the Property, including, without limitation, the water, soil and geology; (vii) the compliance of or by the Property or its operation with any laws, rules, ordinances or regulations of any applicable governmental authority or body including, without limitation, Title III of the Americans With Disabilities Act of 1990; (viii) the manner or quality of the construction or materials incorporated into the Property; (ix) compliance with any environmental protection, pollution or land use laws, rules, regulations, orders or requirements including, without limitation, the Federal Water Pollution Control Act, the Federal Resource Conservation and Recovery Act, the U.S. Environmental Protection Agency Regulations at 40 C.F.R., Part 261, the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, the Resource Conservation and Recovery Act of 1976, the Clean Water Act, the Safe Drinking Water Act, the Hazardous Materials Transportation Act, the Toxic Substance Control Act, and regulations promulgated under any of the foregoing (collectively, "Environmental Laws"); (x) the presence or absence of hazardous materials at, on, under, or adjacent to the Property including, but not limited to, petroleum products, asbestos, lead-based paint, mold or fungi, (xi) the content, completeness or accuracy of any items delivered or made available to Buyer pursuant to Section 3.1; (xii) the conformity of the Improvements to any plans or specifications for the Property, including any plans and specifications that may have been or may be provided to Buyer; (xiii) the conformity of the Property to past, current or future applicable zoning or building requirements; (xiv) deficiency of any undershoring; (xv) deficiency of any drainage; (xvi) the fact that all or a portion of the Property may be located on or near an earthquake fault line; (xvii) the existence of vested land use, zoning or building entitlements affecting the Property; and (xviii) any matters arising from or relating to Seller's alleged superior knowledge or breach of any duty to disclose. Seller is not liable or bound in any manner by any oral or written statements, representations or information pertaining to the Property, or the operation thereof, furnished by any real estate broker, agent, employee, servant or other person. Except as provided in Section 6.4, Buyer further acknowledges and agrees that to the maximum extent permitted by law, the sale of the Property as provided for herein is made on an "AS-IS" "WHERE IS" condition and basis as of the Closing Date with all faults, and that Seller has no obligations to make repairs, replacements or improvements, whether before or after the Close of Escrow. Buyer represents, warrants and covenants to Seller, which representation, warranty, and covenant will survive the Close of Escrow and not be merged with the deed, that, except for the representations and warranties set forth in Section 6.4 of this Agreement, Buyer will solely rely upon Buyer's own investigation of the Property and all other matters relating to the Property, and not on any information provided or to be provided by Seller or anyone acting on Seller's behalf.

6.2 <u>Release and Waiver</u>.

Except with respect to the representations and warranties of Seller set forth (a) in Section 6.4 of this Agreement, Buyer hereby fully and forever releases, acquits and discharges Seller of and from, and hereby fully and forever waives any and all claims, actions, causes of action, suits, proceedings, demands, rights, damages, costs, expenses or other compensation whatsoever, whether known or unknown, direct or indirect, foreseeable or unforeseeable, absolute or contingent, that Buyer now has or may have or which may arise in the future arising out of, directly or indirectly, or in any way connected with the Property or any other matters relating to the Property, including, without limitation: (A) any condition of environmental contamination or pollution at the Property, however and whenever occurring (including, without limitation, the contamination or pollution of any soils, subsoil media, surface waters or groundwaters at the Property) including, but not limited to, petroleum products, asbestos, lead-based paint, mold or fungi; (B) to the extent not already included in (A) above, the prior, present or future existence, release or discharge, or threatened release, of any Hazardous Materials at the Property, however and whenever occurring; (C) the violation of, or non-compliance with, any applicable law now or hereafter in effect, however or whenever occurring; (D) the condition of the soil at the Property; (E) the condition of Improvements including, without limitation, the structural integrity and seismic compliance of such Improvements; (F) the inaccuracy, unreliability, or incompleteness of, or any defect or mistake in, any items delivered or made available to Buyer pursuant to Section 3.1; (G) to the extent not already covered by any of the foregoing clauses (A) through (F), above, the use, maintenance, development, construction, ownership or operation of the Property by Seller or any of Seller's predecessor(s)-in-interest in the Property; (H) any matters arising from or relating to the leases; or (G) any matters based on Seller's alleged superior knowledge or failure to close.

(b) Without limiting the scope or generality of the foregoing release and waiver provisions, those provisions shall specifically include and cover (1) any claim for or right to indemnification, contribution or other compensation based on or arising under any Environmental Law now or hereafter in effect, and (2) any claim for or based on trespass, nuisance, waste, negligence, negligence per se, strict liability, ultrahazardous activity, indemnification, contribution of other theory arising under the common law of the State of California (or any other applicable jurisdiction) or arising, under any applicable law now or hereafter in effect.

6.3 <u>Waiver of Civil Code § 1542</u>. Buyer hereby acknowledges that it has read and is familiar with the provisions of California Civil Code § 1542 ("Section 1542"), which are set forth below:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING

THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR."

Buyer hereby waives the provisions of Section 1542, and of any statute, principle of common law or case law which would limit the scope of the foregoing waiver and release, in connection with matters which are the subject of the foregoing waiver and release.

Buyer's Initials

6.4 <u>Representations of Seller</u>. As of the date hereof and again as of Closing, Seller represents and warrants to Buyer, as to itself and with respect to its Real Property only, as follows:

(a) Except as disclosed to Buyer in writing prior to Close of Escrow, Seller has not received written notice of, and otherwise has no actual knowledge of, any defaults by any of the parties to any of the leases, contracts or agreements respecting the Property to be assigned to Buyer at Closing.

(b) Except as disclosed to Buyer pursuant to Section 3.1, Seller has not received written notice of (i) any litigation (other than unlawful detainer actions) or condemnation, environmental, zoning or other land-use regulation proceedings, either instituted or planned to be instituted, which would materially and adversely affect the ownership, use, operation or value of the Property, or (ii) any special assessment proceedings affecting the Property. Seller shall notify Buyer promptly of any such litigation or proceedings of which Seller becomes aware.

(c) Seller has not received written notice that removal or other remedial action with respect to Hazardous Materials in, on, under or about the Property is required by any governmental authority having jurisdiction over the Property. For purposes of this Agreement, the term "Hazardous Materials" shall mean any toxic or hazardous waste, material or substance, including, without limitation, asbestos, petroleum, petroleum products, underground storage tanks now or previously containing any other Hazardous Materials, substances defined as "hazardous substances", "hazardous waste" or "toxic substances" in any Environmental Law.

(d) Seller is a limited partnership duly organized and validly existing under the laws of the State of California; this Agreement and all documents executed by Seller which are to be delivered to Buyer at the Closing are and at the time of Closing will be duly executed and delivered by Seller, are and at the time of Closing will be legal, valid and binding obligations of Seller enforceable against Seller in accordance with their respective terms and do not and at the time of Closing will not violate any provision of any agreement or judicial order to which Seller or the Property is subject. Seller has obtained all necessary authorizations, approvals and consents to the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby.

(e) Seller has not granted any option or right of first refusal or first opportunity to any party to acquire any interest in the Property.

Seller is in compliance with the requirements of Executive Order (f) No. 13224, 66 Fed. Reg. 49079 (Sept. 25, 2001) (the "Order") and other similar requirements contained in the rules and regulations of the Office of Foreign Assets Control, Department of the Treasury ("OFAC") and in any enabling legislation or other Executive Orders or regulations in respect thereof (the Order and such other rules, regulations, legislation or orders are collectively hereinafter referred to as the "Orders"). Seller (i) is not listed on the Specially Designated Nationals and Blocked Persons List maintained by OFAC pursuant to the Order and/or on any other list of terrorists or terrorist organizations maintained pursuant to any of the rules and regulations of OFAC or pursuant to any other applicable Orders (such lists are collectively referred to as the "Lists"), (ii) is not a Person (as defined in the Order) who has been determined by competent authority to be subject to the prohibitions contained in the Orders, or (iii) is not owned or controlled by (including without limitation by virtue of such person being a director or owning voting shares or interests), or acts for or on behalf of, any person on the Lists or any other person who has been determined by competent authority to be subject to the prohibitions contained in the Orders. Seller is not acting, directly or indirectly for, or on behalf of, any person, group, entity or nation named by any Executive Order (including the September 24, 2001, Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism) or the United States Treasury Department as a terrorist, "Specially Designated National and Blocked Person," or other banned or blocked person, entity, or nation pursuant to any Law that is enforced or administered by the Office of Foreign Assets Control, and is not engaging in the transactions contemplated herein, directly or indirectly, on behalf of, or instigating or facilitating the transactions contemplated herein, directly or indirectly, on behalf of, any such person, group, entity or nation.

(g) Seller has received no written notice from any governmental authority with jurisdiction over the Property of any current violation by the Property of any laws, ordinances or regulations applicable to the Property, and any such past written notices of past violations have been cured in accordance with law. Seller shall immediately provide Buyer with a copy of any such written notices received after the Effective Date.

For purposes of this Agreement, whenever the phrase "to Seller's actual knowledge" or words of similar import are used, they shall be deemed to refer to the actual knowledge, without obligation for inquiry or investigation, of Michael Massie. If Seller becomes aware of any fact or circumstance which would materially change or render materially incorrect, in whole or in part, any representation or warranty made by Seller pursuant to this Section 6.4, whether as of the date given or at any time thereafter through the Closing Date, and whether or not such representation or warranty was based upon Seller's knowledge as of a certain date, Seller will give prompt written notice of such changed fact or circumstance to Buyer. To the extent Buyer has or acquires knowledge prior to the Closing Date that any of Seller's representations and warranties are inaccurate, untrue or incorrect in any way, such representations and warranties shall be deemed modified to reflect Buyer's knowledge. Buyer shall be deemed to know a representation or warranty is untrue, inaccurate or incorrect if this Agreement or any due diligence items made available to Buyer pursuant to Section 3.1 contain information which is inconsistent with such representation or warranty. Upon receipt by Buyer of Seller's notice of such material change in fact or circumstance, or upon Buyer's obtaining knowledge of the material change in fact or circumstance, Buyer may, as Buyer's sole and exclusive remedy, either (i) terminate this Agreement or (ii) waive the breach, representation or warranty and proceed to Closing, and Seller shall have no further obligation or liability to Buyer with respect thereto. Buyer shall make such election within five (5) days of Buyer's receipt of notice or knowledge of such changed fact or circumstance. The representations and warranties set forth herein shall expire and be of no further force or effect six (6) months following the Closing Date; provided that Buyer shall have commenced an action for breach of the representation or warranty prior to the expiration of such six (6) month period. In no event shall Buyer be permitted to bring an action for breach of a representation or warranty unless the damage to Buyer on account of such breach equal or exceeds Ten Thousand Dollars (\$10,000.00) in the aggregate and in no event shall the maximum total liability for breaches of representations or warranties exceed Twenty-Five Thousand Dollars (\$25,000.00) in the aggregate.

6.5 <u>Buyer's Representations and Warranties</u>. As of the date hereof and as of the Closing Date, Buyer represents and warrants to Seller as follows:

(a) Buyer is a limited partnership, duly organized and validly existing under the laws of the State of California; this Agreement and all documents executed by Buyer which are to be delivered to Seller at the Closing are and at the time of Closing will be duly executed and delivered by Buyer, are and at the time of Closing will be legal, valid and binding obligations of Buyer enforceable against Buyer in accordance with their respective terms and do not and at the time of Closing will not violate any provision of any agreement or judicial order to which Buyer or the Property is subject. Buyer has obtained all necessary authorizations, approvals and consents to the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby.

(b)Buyer is in compliance with the requirements of the Orders and other similar requirements contained in the rules and regulations of OFAC. Buyer (i) is not listed on the Specially Designated Nationals and Blocked Persons List maintained by OFAC pursuant to the Order and/or on any other maintained pursuant to any of the rules and regulations of OFAC or pursuant to any other Lists, (ii) is not a Person (as defined in the Order) who has been determined by competent authority to be subject to the prohibitions contained in the Orders, or (iii) is not owned or controlled by (including without limitation by virtue of such person being a director or owning voting shares or interests), or acts for or on behalf of, any person on the Lists or any other person who has been determined by competent authority to be subject to the prohibitions contained in the Orders. Buyer is not acting, directly or indirectly for, or on behalf of, any person, group, entity or nation named by any Executive Order (including the September 24, 2001, Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism) or the United States Treasury Department as a terrorist, "Specially Designated National and Blocked Person," or other banned or blocked person, entity, or nation pursuant to any Law that is enforced or administered by the Office of Foreign Assets Control, and is not engaging in the transactions contemplated herein, directly or indirectly, on behalf of, or instigating or facilitating the transactions contemplated herein, directly or indirectly, on behalf of, any such person, group, entity or nation.

6.6 <u>Seller's Covenants Pending Closing</u>. During the term of this Agreement, Seller covenants and agrees as follows:

(a) <u>Contracts</u>. Seller will not enter into any agreement or contract, affecting a period after the Closing Date that cannot be terminated on thirty (30) days' notice without the prior written consent of Buyer, which consent may not unreasonably be withheld, conditioned or delayed. The failure of Buyer to respond to a request for approval of an agreement or contract within three (3) business days of Seller's request for approval shall be deemed to constitute Buyer's approval thereof. Seller will notify Buyer in writing of any agreements or contracts it enters into prior to Closing.

(b) <u>Leases</u>. From and after the expiration of the Due Diligence Period (assuming Buyer elects to proceed with the purchase), without the prior written consent of Buyer not to be unreasonably withheld, conditioned or delayed, none of the existing tenant leases will be amended to reduce the rents or other charges thereunder or will be renewed for rents or other charges either of which are less than those payable for similar apartment units in the Property; and no new lease or extension to an existing lease will be made unless it is for a term of not more than one year and at rent not less than that charged for similar apartment units in the Property. Buyer's failure to approve or disapprove the amendment, renewal or new lease within three (3) days of Seller's request shall be deemed to constitute Buyer's approval thereof.

(c) <u>Insurance</u>. Seller will maintain in effect all insurance policies now maintained on the Property, up to and including the Closing Date.

(d) <u>Operation and Condition Pending Closing</u>. Seller will continue to manage, operate and maintain the Property in the same manner as it has before the execution of this Agreement.

VII DAMAGE, DESTRUCTION AND CONDEMNATION

7.1 <u>Risk of Physical Loss</u>. Buyer shall be obligated to acquire the Property pursuant to this Agreement, notwithstanding that the Property is damaged by fire or other casualty prior to the Closing Date, provided that (i) if the cost to repair the Property, in the reasonable judgment of Buyer, can be completed for less than two percent (2%) of the Purchase Price, (ii) the cost to repair such damage is covered by insurance maintained by or for Seller, (iii) Seller assigns to Buyer all insurance proceeds and/or rights to receive insurance proceeds by reason of such damage through Escrow on the Close of Escrow, other than proceeds expended in restoration and repair by Seller, and (iv) Seller credits to the account of Buyer in Escrow, the amount of any deductible under Seller's insurance (not to exceed the cost of repair). Any such casualty is hereinafter referred to as a "non-material insured casualty".

In the event of casualty damage to the Property, other than a non-material insured casualty, Buyer may, at its option, either terminate this Agreement by giving written notice of such termination to Seller within ten (10) days of Buyer's receipt of written notice of such casualty from Seller, or elect to proceed with its purchase of the Property, in which event Seller shall transfer and assign to Buyer all insurance proceeds and all rights to receive insurance proceeds by reason of such damage through Escrow at its Close, other than proceeds expended in restoration and repair by Seller and rental loss proceeds applied to rents accruing through the Close of Escrow and shall credit Buyer's account in Escrow the amount of any deductible under Seller's insurance (not to exceed the cost of repair). If the right to receive any such insurance proceeds to be assigned to Buyer is not assignable by Seller to Buyer, Buyer may nevertheless elect to close the Escrow, in which event Seller shall promptly deliver to Buyer the proceeds of any such insurance received by it following the Close of Escrow, except to the extent such proceeds are in reimbursement for restoration and repair costs incurred or to be incurred by Seller prior to the Close of Escrow and/or rental loss proceeds for rents accruing prior to the Close of Escrow. In the event that such damage shall occur and Buyer elects not to purchase the Property, then this Agreement shall be terminated and Buyer shall be entitled to the return of all funds and documents deposited hereunder.

7.2 <u>Condemnation</u>. In the event that, prior to the Close of Escrow, any governmental entity shall commence any actions of eminent domain or similar type proceedings to take any portion of the Property, Buyer shall have the option either to (i) elect not to acquire the Property, or (ii) complete the acquisition of the Property, in which case Buyer shall be entitled to all the proceeds of such taking.

VIII

MISCELLANEOUS

8.1 <u>Attorneys' Fees</u>. In the event of any action between Buyer and Seller seeking enforcement of any of the terms and conditions to this Agreement, or otherwise in connection with the Property, the prevailing party in such action shall be awarded, in addition to damages, injunctive or other relief, its reasonable costs and expenses, including but not limited to reasonable attorneys' fees, court costs and expert witness fees.

8.2 <u>Notices</u>. All notices under this Agreement shall be effective upon (i) personal delivery to Buyer or Seller, as the case may be, or (ii) telecopier transmission with a hard copy deposited in overnight mail (Express Mail), or (iii) one (1) business day after deposit with an overnight courier service (e.g., Federal Express), (iv) upon delivery by electronic mail; or (v) three (3) business days after deposit in the United States mail, registered, certified, postage fully prepaid and addressed to the respective parties as follows:

| To Seller: | JAMBOREE-TAL HOUSING II, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attn: Michael Massie Facsimile: (949) 214-2361 Email: mmassie@jamboreehousing.com |
|-----------------|---|
| With a copy to: | Rutan & Tucker, LLP 18575 Jamboree Road, Suite 900 Irvine, CA 92612 Attn: Patrick D. McCalla |

| | Facsimile: (714) 546-9035 Email: pmccalla@rutan.com |
|-----------------|--|
| To Buyer: | AMISTAD HOUSING PARTNERS II LP c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attn: Michael Massie Facsimile: (949) 214-2361 Email: mmassie@jamboreehousing.com |
| With a copy to: | Rutan & Tucker, LLP 18575 Jamboree Road, Suite 900 Irvine, CA 92612 Attn: Patrick D. McCalla Facsimile: (714) 546-9035 Email: pmccalla@rutan.com |

or to such other address as the parties may from time to time designate in writing.

8.3 <u>Entire Agreement</u>. This Agreement and the items incorporated herein contain all the agreements of the parties hereto with respect to the matters contained herein; and no prior agreement or understanding pertaining to any such matter (including, without limitation, the Letter of Intent) shall be effective for any purpose. No provisions of this Agreement may be amended or modified in any manner whatsoever except by an agreement in writing signed by duly authorized officers or representatives of each of the parties hereto.

8.4 <u>Successors</u>. The terms, covenants and conditions of this Agreement shall be binding upon and shall inure to the benefit of the heirs, executors, administrators and assigns of the respective parties hereto.

8.5 <u>Assignment</u>. Buyer may assign this Agreement and its rights hereunder to an entity in which Buyer owns a majority of the interests in profits and losses without the necessity of acquiring consent from Seller provided such party is designated not later than ten (10) days prior to the Close of Escrow. No other assignment may occur without the consent of Seller, which consent may be withheld in Seller's sole and absolute discretion. In no instance shall an assignment by either Buyer or Seller relieve the assigning party of its obligations hereunder.

8.6 <u>Choice of Laws</u>. This Agreement shall be governed by the laws of the State of California and any questions arising hereunder shall be construed or determined according to such law.

8.7 <u>Headings</u>. Headings at the beginning of each numbered Article and Section of this Agreement are solely for the convenience of the parties and are not a part of this Agreement.

8.8 <u>Survival</u>. This Agreement and all covenants, representations and warranties contained herein shall survive the close of this transaction and this Agreement shall remain a binding contract between the parties hereto.

8.9 <u>Time</u>. Time is of the essence of this Agreement, it being understood that each date set forth herein and the obligations of the parties to be satisfied by such date have been the subject of specific negotiation by the parties. Except as expressly provided for in this Agreement, the time for performance of any obligation or taking any action under this Agreement will be deemed to expire at 5:00 p.m. (California time) on the last day of the applicable time period provided for in this Agreement. If the time for performance of any obligation or taking any action under this Agreement expires on a Saturday, Sunday or California or Federal holiday, the time for performance or taking such action will be extended to the next succeeding day which is not a Saturday, Sunday or California or Federal holiday. The term "business days" shall not include any day which is a Saturday, Sunday or California or Federal holiday.

8.10 <u>Counterparts</u>. This Agreement may be signed by the parties in different counterparts and the signature pages combined to create a document binding on all parties.

8.11 <u>Brokerage Commissions</u>. Each party represents and warrants that it has retained no brokers or finders to represent its interests in connection with this transaction. Each party agrees to indemnify and hold the other harmless from and against all liabilities, costs, damages and expenses, including, without limitation, reasonable attorneys' fees, resulting from any claims or fees or commissions, based upon agreements by it, if any, to pay any broker's commission and/or finder's fee.

8.12 Information Report. Escrow Holder shall file and Buyer and Seller agree to cooperate with Escrow Holder and with each other in completing any report ("Information Report") and/or other information required to be delivered to the Internal Revenue Service pursuant to Internal Revenue Code Section 6045(e) regarding the real estate sales transaction contemplated by this Agreement, including, without limitation, Internal Revenue Service Form 1099-B as such may be hereinafter modified or amended by the Internal Revenue Service, or as may be required pursuant to any regulation now or hereinafter promulgated by the Treasury Department with respect thereto. Buyer and Seller also agree that Buyer and Seller, their respective employees and attorneys, and Escrow Holder and its employees may disclose to the Internal Revenue Service, whether pursuant to such Information Report or otherwise, any information regarding this Agreement or the transaction contemplated herein as such party reasonably deems to be required to be disclosed to the Internal Revenue Service by such party pursuant to Internal Revenue Code Section 6045(e), and further agree that neither Buyer nor Seller shall seek to hold any such party liable for the disclosure to the Internal Revenue Service of any such information.

8.13 <u>Cooperation in 1031 Exchange</u>. The parties each agree to cooperate with the other (at no cost or liability to the agreeing party) in effecting for the benefit of either party a concurrent, delayed or reverse like-kind exchange of real property pursuant to Section 1031 of the United States Internal Revenue Code and similar provisions of applicable state law; provided, however, that (i) neither party shall be obligated to accept title, or to otherwise become part of the chain of title, to an exchange property of the other party; and (ii) the Closing Date shall not be delayed as the result of such exchange (except as provided in Section 2.1). This Agreement and the parties' obligations hereunder are not subject to or conditioned upon ability to consummate an exchange.

Each party's responsibility for reviewing exchange documents shall be limited to determining whether the terms and conditions of such exchange documents are such that they are in compliance with the foregoing provisions. Each party shall be responsible for the payment of all deposits and other costs required to be paid by the "buyer" pursuant to the exchange documents and for making all determinations as to the legal sufficiency or other consideration including, but not limited to, tax considerations, relating to such exchange documents. In so cooperating in any exchange transaction arranged by a party, the non-exchanging party shall in no event be responsible for, or in any way warrant, the tax consequences of the exchange transaction.

8.14 <u>No Third Party Beneficiary</u>. No term or provision of this Agreement is intended to, or shall be, for the benefit of any person not a party hereto and no such person shall have any right or cause of action hereunder.

8.15 <u>Limited Liability</u>. The obligations of Seller under this Agreement and under all of the documents referenced herein are intended to be binding only on the assets of Seller and shall not be personally binding upon, nor shall any resort be had to, the private properties of any member of Seller or any trustee, officer, director, employee or affiliate of Seller or any of its members.

8.16 <u>Jury Waiver</u>. TO THE EXTENT PERMITTED BY LAW, BUYER AND SELLER EACH WAIVES RIGHT TO A JURY IN ANY LITIGATION IN CONNECTION WITH THIS AGREEMENT, OR THE PROPERTY, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. BUYER AND SELLER EACH ACKNOWLEDGES THAT THIS WAIVER HAS BEEN FREELY GIVEN AFTER CONSULTATION BY IT WITH COMPETENT COUNSEL. THE PROVISIONS OF THIS SECTION 8.17 SHALL SURVIVE CLOSING OR ANY EARLY TERMINATION OF THIS AGREEMENT.

Arbitration. If a dispute arises between Seller and Buyer arising out of or relating 8.17 to this Agreement or the Property, such dispute shall be determined and resolved by a single arbitrator appointed and acting pursuant to Code of Civil Procedure section 1280, et seq. and the Commercial Arbitration Rules of the American Arbitration Association ("AAA"). Any arbitrator selected under this Section shall be knowledgeable of the subject matter of the dispute. A qualified retired judge with at least five (5) years arbitration experience shall be selected as the arbitrator through panels maintained by the American Arbitration Association, any court in which the property is located or any private organization providing such services. The parties shall have the full right of discovery allowed under Code of Civil Procedure section 1283.05, and the arbitrator may, in the arbitrator's discretion, allow additional discovery. The fees charged by the arbitrator shall be paid equally by Seller and Buyer unless the arbitrator awards such fees to the prevailing party. The venue for the arbitration shall be in Orange County, California. The decision of the arbitrator shall be final and non-appealable and judgment may be entered on the award in accordance with applicable law. Nothing contained herein shall be construed to preclude a party from commencing an action to obtain provisional remedies; however, after granting the provisional relief, the action shall be stayed pending completion of the arbitration proceeding.

8.18 <u>Acceptance of Grant Deed</u>. The acceptance of the Grant Deed by the Buyer shall be deemed full compliance by the Seller of all the Seller's obligations under this Agreement except for those obligations of the Seller which are specifically stated to survive the close of this transaction.

8.19 <u>Bulk Sale Requirements</u>. Seller and Buyer hereby acknowledge and agree that the parties do not intend to comply with and have waived the provisions of any statutory bulk sale requirements applicable to the transaction to be effected by this Agreement, and to rely upon the adjustment provisions of this Agreement to address any matters that would otherwise be subject to such bulk sale requirements.

8.20 <u>Incorporation</u>. The exhibits attached to this Agreement are incorporated herein and made a part hereof.

[Signatures appear on the following page.]

IN WITNESS WHEREOF, Buyer and Seller have executed this Agreement as of the date first above written.

BUYER:

AMISTAD HOUSING PARTNERS II LP, a California limited partnership

- By: JHC-AMISTAD II LLC, a California limited liability company, its general partner
 - By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By: ______ Name: Michael Massie Title: Executive Vice President and Chief Development Officer

SELLER:

JAMBOREE-TAL HOUSING II, L.P., a California limited partnership

By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its general partner

By:

Name: Michael Massie Title: Executive Vice President and Chief Development Officer

ACCEPTANCE BY ESCROW HOLDER

The undersigned hereby acknowledges that it has received originally executed counterparts or a fully executed original of the foregoing Purchase and Sale Agreement and Escrow Instructions and agrees to act as Escrow Holder thereunder and to be bound by and perform the terms thereof as such terms apply to Escrow Holder.

Dated: _____, 2024

FIRST AMERICAN TITLE INSURANCE COMPANY

By: _______Ryan Hahn, Senior Escrow Officer

EXHIBIT "A"

LEGAL DESCRIPTION OF REAL PROPERTY

Real property in the City of San Clemente, County of Orange, State of California, described as follows:

LOT 2 OF TRACT NO. 13898, AS SHOWN ON A MAP FILED IN <u>BOOK 817, PAGES 36 TO 40</u> INCLUSIVE OF MISCELLANEOUS MAPS, RECORDS OF ORANGE COUNTY, CALIFORNIA.

EXCEPTING THEREFROM ALL WATER AND WATER RIGHTS, IF ANY, INCLUDED WITHIN AND UNDERLYING THE DISTINCTIVE BORDER OF THIS TRACT MAP AS DEDICATED TO THE SANTA MARGARITA WATER DISTRICT ON THE MAP OF SAID TRACT.

ALSO EXCEPTING THEREFROM ANY AND ALL UNPROCESSED OIL, OIL RIGHTS, MINERALS, MINERAL RIGHTS, NATURAL GAS RIGHTS AND OTHER HYDROCARBONS BY WHATSOEVER NAME KNOWN, GEOTHERMAL STEAM AND ALL PRODUCTS DERIVED FROM ANY OF THE FOREGOING, THAT MAY BE WITHIN OR UNDER THE PROPERTY, TOGETHER WITH THE PERPETUAL RIGHT OF DRILLING, MINING, EXPLORING, AND OPERATING THEREFOR AND STORING IN AND REMOVING THE SAME FROM THE PROPERTY OR ANY OTHER LAND AS DESCRIBED THEREIN, BUT WITHOUT THE RIGHT TO ENTER UPON OR USE THE SURFACE OF THE PROPERTY TO DRILL, MINE, STORE, EXPLORE, OR OPERATE THROUGH THE SURFACE OR THE UPPER 500 FEET OF THE SUBSURFACE OF THE PROPERTY, RECORDED JUNE 28, 2002 AS INSTRUMENT NO. 20020548428.

APN: 701-041-38

EXHIBIT "B"

<u>NOTE</u>

[see attached]

PROMISSORY NOTE SECURED BY SUBORDINATED DEED OF TRUST

\$[____]

Irvine, California , 2024

FOR VALUE RECEIVED, the undersigned, AMISTAD HOUSING PARTNERS II LP, a California limited partnership ("Borrower") hereby promises to pay to the order of Jamboree-Tal Housing II, L.P., a California limited partnership ("Lender"), whose address is 17701 Cowan Ave., Suite 200, Irvine, California 92614 or at such other place as Lender may from time to time designate in writing, the principal sum of [\$1,185,511] or such lesser amount as may be advanced to or for the account of Borrower pursuant to this Promissory Note ("Note), together with all accrued interest on the unpaid balance this Note, and all other amounts now or hereafter owing by Borrower to Lender under this Note. The indebtedness evidenced by this Note is referred to as the "Loan". The Loan is being made by Lender to Borrower in connection with that certain Purchase and Sale Agreement and Escrow Instructions, dated as of , 2024 (the "Purchase Agreement"), pursuant to which Borrower is simultaneously acquiring from Lender certain real property more particularly described in such Purchase Agreement (the "Property"). Borrower shall pay interest at the rates, in the amounts and at the times hereinafter provided. The obligation of Borrower to Lender is set forth in and subject to the terms of this Note and that certain Subordinated Deed of Trust dated as of even date herewith executed by Borrower for the purpose of securing this Note, as the same shall be amended from time to time (the "Subordinated Deed of Trust"). Capitalized terms not otherwise defined herein shall have the meanings set forth Borrower's Limited Partnership Agreement, dated as of August 3, 2023 (the "Partnership Agreement").

1. <u>Principal and Interest</u>. From and after the date hereof through and including the Maturity Date (as hereinafter defined), simple interest shall accrue on the unpaid principal amount outstanding from time to time under this Note at an annual rate equal to [___]% [Insert Term AFR for month of closing] per annum (the "Loan Rate").

2. <u>Payments.</u> Payments under this Note shall be made annually in arrears in immediately available funds for the previous calendar year on the 1st day of May, commencing with May 1 in the year immediately following the year in which the Property is initially occupied in an amount equal to 100% of Residual Receipts for the corresponding year. If the due date for any payment is not a business day, such due date shall be deemed for all purposes to fall on the next business day. As used in this Note, the term "business day" shall mean a day other than a Saturday, Sunday or a day on which banking institutions in the State of California are authorized or required by law to close.

As used herein, the term "Residual Receipts" shall mean Annual Revenue less the sum of:

- (A) Operating Expenses;
- (B) Debt Service;
- (C) Partnership Related Fees/Expenses;

(D) payment of unpaid tax credit adjustment amounts or reimbursement of tax credit adjustment amounts paid by the administrative and/or managing general partners and/or the guarantors to the Property pursuant to the Partnership Agreement, if any;

(E) repayment of loans, if any, made by the partner(s) of Borrower;

(F) Deferred Developer Fee for the Property which remains unpaid; and

(G) repayment of outstanding development and operating loans and/or contributions for capital expenses for which no revenues from operation of the Improvements are available, if any, made by the managing general partner and/or the guarantor to the Property.

"Annual Revenue" shall mean all gross income and all revenues from the operation of the Property in a calendar year received by Borrower determined on the basis of generally accepted accounting principles applied on a consistent basis. Notwithstanding the foregoing, Annual Revenue shall not include the following items: (a) security deposits from tenants (except when applied by Borrower to rent or other amounts owing by tenants); (b) capital contributions to Borrower by its members, partners or shareholders (including capital contributions required to pay any Deferred Developer Fee); (c) condemnation or insurance proceeds; or (d) proceeds of any loan.

"Operating Expenses" shall mean actual costs, fees and expenses attributable to the operation, maintenance, and management of the Property in a calendar year, including, without limitation, painting, cleaning, repairs, alterations, landscaping; utilities, refuse removal, certificates, permits and licenses, sewer charges, taxes, filing fees, assessments, insurance, security, advertising and promotion, janitorial services, cleaning and building supplies, purchase, repair, servicing and installation of appliances, equipment, fixtures and furnishings which are not paid from the capital replacement reserve, fees and expenses of property management and common area expenses, fees and expenses of accountants, attorneys and other professionals, the cost of social services for the tenant households and other actual, reasonable and customary operating costs which are directly incurred and paid by Borrower, but which are not paid from reserve accounts. Operating Expenses shall include costs, fees and expenses paid for the operation of Borrower, including administrative, accounting and legal fees and expenses.

"Debt Service" shall mean payments made in a calendar year pursuant to the financing obtained for the construction, ownership, and operation of the Property permitted pursuant to the Loan Documents, and those certain loans in effect as of the date hereof, as evidenced by the (i) First Deed of Trust (as defined below), (ii) the HCD Deed of Trust (as defined below), (iii) the HOME Deed of Trust, (iv) the OCDA Deed of Trust, (v) the City Deed of Trust, and (vi) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the then-outstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate.

"Partnership Related Fees/Expenses" shall mean the fees and expenses of Borrower payable to the partners in Borrower or affiliates thereof pursuant to the Partnership Agreement. In the event insufficient Annual Revenues exist to provide for payment of all or part of the Partnership Related Fees/Expenses the unpaid balance will accrue.

"Deferred Developer Fee" means the portion of the Developer Fee (as defined in the [Partnership Agreement]) to be paid from cash flow from Annual Revenue.

3. <u>Maturity</u>. The entire unpaid principal balance of this Note, together with all accrued and unpaid interest, shall be due and payable on _____, 20_ [*Insert 58 years from date of Closing*] (the "Maturity Date").

4. <u>Prepayment</u>. The principal of this Note may be voluntarily prepaid at any time, in its entirety or in any partial amount, without penalty.

5. <u>Events of Default</u>. The occurrence of any of the following, whatever the reason therefor, shall constitute an "Event of Default" hereunder:

a. Borrower fails to make any regularly scheduled payment of principal or interest hereunder, to the extent there are Residual Receipts (a "Required Payment"), within thirty (30) days of receipt by Borrower of written notice of such failure; or

b. Borrower fails to pay any other amount owing to Lender under this Note, or fails to perform any other obligation under this Note, the breach of which can be cured by the payment of money, within thirty (30) days after receipt by Borrower of written notice thereof; or

c. Borrower fails to perform any obligation under this Note (other than obligations the breach of which can be cured by the payment of money) within thirty (30) days after notice; *provided, however*, that, if cure cannot reasonably be effected within such thirty (30) day period, there shall be no Event of Default under this subparagraph (c) so long as Borrower promptly commences cure and thereafter diligently prosecutes such cure to completion; or

d. Borrower is dissolved, liquidated or terminated, or all or substantially all of the assets of Borrower are sold or otherwise transferred without Lender's prior consent; or

e. Borrower is the subject of an order for relief by a bankruptcy court, or is unable or admits its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors; or Borrower applies for or consents to the appointment of any rehabilitator or similar officer for it or any part of its property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of Borrower and the appointment continues undischarged or unstayed for one hundred twenty (120) days; or Borrower institutes or consents to any bankruptcy, insolvency, reorganization, arrangement, readjustment or debt, dissolution, custodianship, conservatorship, liquidation, rehabilitation or similar proceeding relating to it or any part of its property; or any similar proceeding is instituted without the consent of Borrower, as the case may be, and continues undismissed or unstayed for one hundred twenty (120) days; or Borrower and is not released, vacated or fully bonded within one hundred twenty (120) days after its issue or levy.

Notwithstanding anything to the contrary contained herein, prior to declaring any default or taking any remedy permitted hereunder or under applicable law based upon an alleged default under this Note, Lender shall give a written notice to the limited partner of Borrower at the notice address set forth herein and the limited partner shall have a period of not less than thirty (30) days to cure such alleged default; provided, however, if in order to cure such default the limited partner reasonably believes that it must remove the general partners of Borrower (or one of them) pursuant to the provisions contained in the Partnership Agreement of Borrower, the limited partner shall so notify Lender and so long as the limited partner is diligently attempting to so remove such general partner(s), the limited partner shall have until the date thirty (30) days after the effective date of the removal of such general partner(s) to cure such default.

6. <u>Limitation on Remedies</u>. Notwithstanding anything to the contrary contained herein or in any other document evidencing or securing the Note, during any time prior to the expiration of the [Compliance Period (as defined in the Partnership Agreement)] in which Lender, or an affiliate of Lender, acts as general partner of Borrower, Lender shall have no right to declare a default hereunder or thereunder, initiate an arbitration pursuant to this Note, obtain a receiver or exercise any other remedy hereunder or under any other document evidencing or securing this Note without first obtaining the prior written consent of the limited partner of Borrower; provided, however, that Lender may bring an action for specific performance to compel any payments due Lender under this Note.

7. <u>Acceleration and Other Remedies</u>. Upon the occurrence of any Event of Default, the holder of this Note may, at its option, declare the outstanding principal balance of this Note, together with all accrued interest thereon, to be immediately due and payable. All rights and remedies provided to the holder of this Note are cumulative and shall be in addition to all other rights and remedies of the holder under other documents, at law or in equity, and all such rights and remedies may be exercised singly, successively and/or concurrently. Failure to exercise any right or remedy shall not be deemed a waiver of such right or remedy.

8. <u>Due on Sale; Due on Encumbrance</u>. Notwithstanding anything to the contrary contained herein, Lender may declare all obligations hereunder immediately due and payable if Borrower, without Lender's prior written consent, sells, conveys, leases with an option to purchase or further encumbers any interest in or any part of the Property. Any such sale, conveyance, transfer, pledge, lease or encumbrance made without Lender's prior written consent shall be void. Borrower shall not, without the prior written consent of Lender, assign the rents from the Property and any such assignment without the express written consent of Lender shall be null and void.

9. <u>General Partner Change</u>. The withdrawal, removal, and/or replacement of a general partner of the Partnership pursuant to the terms of the Partnership Agreement shall not constitute a default under this Note, and any such actions shall not accelerate the maturity of the Loan, provided that any required substitute general partner is reasonably acceptable to Lender and is selected with reasonable promptness.

10. <u>Purchase Rights</u>. The execution and delivery of the purchase option and right of first refusal agreement described in the Partnership Agreement shall not constitute a default under this Note or accelerate the maturity of the Loan. Any requisite consent of Lender to (a) the exercise of said purchase option and right of first refusal agreement by the Managing General Partner and to (b) the assumption without penalty of loan obligations by the Managing General Partner and the release of Borrower from such obligations, shall not be unreasonably withheld. Subject to any such consent requirement, the exercise of rights under such agreement shall not constitute a default or accelerate maturity of the Loan.

11. <u>Lender Approvals, Etc</u>. In any approval, consent, or other determination by Lender required under this Note, Lender shall act reasonably and in good faith.

12. <u>Application of Payments</u>. All payments made hereunder shall be applied first to interest due and then in reduction of the principal balance. If more than one annual payment

remains unpaid, amounts received shall be applied to such payments in reverse chronological order.

13. <u>Expenses/Attorneys' Fees</u>. If this Note is not paid when due, whether at each scheduled payment date, at maturity or by acceleration, Borrower promises to pay all costs of collection, including, but not limited to, reasonable attorneys' fees and all expenses incurred in connection with the protection or realization of the collateral securing the payment hereof or enforcement of any guarantee, incurred by Lender on account of such collection, whether or not suit is filed hereon.

14. <u>Waivers</u>. Except as expressly set forth in this Note, Borrower hereby waives presentment, demand, protest, notice of dishonor, notice of protest and all other notices and demands of every kind that would otherwise be available in connection with this Note.

15. <u>Notice</u>. All notices and other communications provided for in this Note shall be in writing and be delivered by messenger, overnight air courier or registered or certified U.S. mail with return receipt requested to the appropriate party at its address as follows:

| If to Lender: | Jamboree-Tal Housing II, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Michael Massie |
|-----------------|--|
| If to Borrower: | Amistad Housing Partners II LP c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Michael Massie |

Addresses for notice may be changed from time to time by written notice to the other party. Communications by mail shall be effective upon the earlier of (a) 96 hours after deposit in the U.S. mail with postage prepaid, (b) one day after deposit with a reputable national overnight courier service, or (c) actual receipt, as indicated by the return receipt; and communications by messenger shall be effective when delivered.

16. <u>Miscellaneous</u>. Time is of the essence hereof. All payments hereunder shall be made in lawful money of the United States of America. This Note shall be governed by the laws of the State of California. Neither this Note nor any term hereof may be waived, amended, discharged, modified, changed or terminated orally; nor shall any waiver of any provision hereof be effective except by an instrument in writing signed by Borrower and Lender. No delay or omission on the part of Lender in exercising any right hereunder shall operate as a waiver of such right or of any other right under this Note.

17. <u>Subordination</u>. The indebtedness evidenced by this Note is and shall be subordinate in right of payment to the prior payment in full of all amounts then due and payable (including, but not limited to, all amounts due and payable by virtue of any default or acceleration or upon maturity) with respect to (i) that certain _______, dated as of the date hereof, by Borrower in favor of First American Title Insurance Company, a California corporation, as trustee, for the benefit of _______, a ______, as beneficiary (the "First Deed of Trust"), (ii) that deed of trust relating to the loan from the Department of Housing

and Community Development, a public agency of the State of California, or its affiliates, encumbering the Property (collectively, the "HCD Deed of Trust"), (iii) that deed of trust relating to the loan from the County of Orange, or its affiliates, encumbering the Property (collectively, the "HOME Deed of Trust"), (iv) that deed of trust relating to the loan from the Orange County Development Agency, or its affiliates, encumbering the Property (collectively, the "OCDA Deed of Trust"), (v) that deed of trust relating to the loan from the City of San Clemente, or its affiliates, encumbering the Property (collectively, the "City Deed of Trust"), and (vi) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the then-outstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate. Lender further agrees that upon the maturity of any debt or obligation senior in priority to the Subordinated Deed of Trust, Lender shall subordinate this Note and the Subordinated Deed of Trust to any new debt which is used to partially or fully repay said maturing senior debt or obligation.

18. <u>Non-Recourse</u>. Neither Borrower nor Borrower's officers, partners, directors, employees or agents nor their respective members, managers, officers, partners, shareholders, directors, employees or agents shall have personal liability for the payment or performance of any of Borrower's obligations under this Note or the Subordinated Deed of Trust. Lender's sole recourse shall be to realize against the collateral described in the Subordinated Deed of Trust.

Executed this ____ day of _____ 2024.

BORROWER:

AMISTAD HOUSING PARTNERS II LP, a California limited partnership

- By: JHC-AMISTAD II LLC, a California limited liability company, its general partner
 - By: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By:

Name: Michael Massie Title: Executive Vice President and Chief Development Officer

EXHIBIT "C"

DEED OF TRUST

[see attached]

RECORDING REQUESTED BY AND WHEN RECORDED, RETURN TO:

Jamboree-Tal Housing II, L.P. c/o Jamboree Housing Corporation 17701 Cowan Ave., Suite 200 Irvine, CA 92614 Attention: Asset Management

SUBORDINATED DEED OF TRUST

THIS SUBORDINATED DEED OF TRUST ("Subordinated Deed of Trust") is made on this _____ day of ______, 2024 by ______, a _____, as trustor ("Borrower") to First American Title Insurance Company ("Trustee"), for the benefit of Jamboree-Tal Housing II, L.P., a California limited liability company, as beneficiary ("Lender").

Borrower, in consideration of the indebtedness recited and the trust created in this Subordinated Deed of Trust, irrevocably grants and conveys to Trustee, in trust, with power of sale certain real property owned in fee by Borrower and located in the County of Orange, State of California ("Property"). The Property is legally described in <u>"Exhibit A"</u> attached hereto and made a part hereof.

Together with all of the improvements now or hereafter erected on said property, and all easements, rights, appurtenances, rents (subject, however, to the rights and authorities given in this Subordinated Deed of Trust to Lender to collect and apply such rents), royalties, mineral, oil and gas rights and profits, water, water rights, and water stock, and all fixtures now or hereafter attached to said property, all of which including replacements and additions thereto, shall be deemed to be and remain a part of the property covered by this Subordinated Deed of Trust; and all of the foregoing, together with said property are referred to in this Subordinated Deed of Trust as the "Property";

To secure for the benefit of Lender the repayment of the indebtedness evidenced by Borrower's Promissory Note Secured by Subordinated Deed of Trust of even date herewith (the "Note") in the principal sum not to exceed \$[1,185,511] and the performance of the covenants and agreements of Borrower contained in this Subordinated Deed of Trust and in the Note.

Borrower covenants and agrees as follows:

1. PAYMENT OF PRINCIPAL AND INTEREST. Borrower shall pay when due the interest and the principal of the indebtedness evidenced by the Note from the sources and pursuant to the terms of the Note.

2. COMPLIANCE WITH SENIOR DEED OF TRUST. Borrower covenants and agrees to comply with the terms and conditions of any senior deeds of trust recorded against the Property (collectively, the "Senior Encumbrances").

3. CHARGES AND LIENS. Borrower shall pay all taxes, assessments and other charges, fines and impositions attributable to the Property which may attain a priority over this Subordinated Deed of Trust. Borrower shall promptly furnish to Lender all notices of amounts

due under this paragraph and in the event Borrower shall make payment directly, Borrower shall promptly furnish to Lender receipts evidencing such payments. Borrower shall promptly discharge any lien which has priority over this Subordinated Deed of Trust; provided that Borrower shall not be required to discharge any such lien so long as Borrower shall agree in writing to the payment of the obligation secured by such lien in a manner acceptable to Lender, or shall in good faith contest such lien by, or defend enforcement of such lien in, legal proceedings which operate to prevent the enforcement of the lien or forfeiture of the Property or any part thereof. Borrower shall not allow the attachment of any subordinate lien or other encumbrance on the Property without the prior written consent of Lender.

Borrower shall promptly pay when due all installments or payments required by Lender and comply with all obligations of any deed of trust that is prior to this Subordinated Deed of Trust.

4. INSURANCE PROCEEDS. Any insurance proceeds received as a result of damage to the Property from fire or other calamity shall be first expended as required by any Senior Encumbrances.

5. PROTECTION OF LENDER'S SECURITY. If Borrower fails to perform the covenants and agreements contained in this Subordinated Deed of Trust, or if any action or proceeding is commenced which materially affects Lender's interest in the Property, including, but not limited to, default on or foreclosure of a prior deed of trust, eminent domain, insolvency, code enforcement, or arrangement or proceedings involving a bankrupt or decedent, then Lender at Lender's option, upon notice to Borrower, may make such appearances, disburse such sums and take such action as is necessary to protect Lender's interest including, but not limited to, disbursement of reasonable attorney's fees and entry upon the Property to make repairs, but shall have no right of reimbursement.

6. INSPECTION. Lender may make or cause to be made reasonable entries upon and inspections of the Property, provided that Lender shall give Borrower notice prior to any such inspection specifying reasonable cause therefor related to Lender's interest in the Property.

7. CONDEMNATION. Subject to the rights of the holder of a prior deed of trust, the proceeds of any award or claim for damages, direct or consequential, in connection with any condemnation or other taking of the Property, or part thereof, or for conveyance in lieu of condemnation, are hereby assigned and shall be paid to Lender.

In the event of a total taking of the Property, the proceeds shall be treated as if received from a sale of the Property. The proceeds shall be applied to the sums to be repaid in the amount and manner described in the Senior Encumbrances and the documents evidencing the indebtedness secured thereby. Any part of the proceeds remaining after these amounts have been paid shall be paid to Borrower. In the event of a partial taking of the Property, unless Borrower and Lender agree in writing, the proceeds shall be treated as if received from a sale of that portion of the Property which is taken in the condemnation. The percentage value of the portion taken, as compared to the full value of the entire Property, shall be determined by dividing the condemnation proceeds by the fair market value of the entire Property just prior to the taking. This percentage value, once determined, shall be used in the following manner to allocate the condemnation proceeds:

a. First, to the payment of any amount owing under the Senior Encumbrances and the documents evidencing the indebtedness secured thereby;

- b. Second, to payment of the Note;
- c. Third, to Borrower.

8. BORROWER NOT RELEASED. Extension of the time for payment or modification of amortization of the sums secured by this Subordinated Deed of Trust granted by Lender to any successor in interest of Borrower shall not operate to release, in any manner, the liability of the original Borrower and Borrower's successors in interest. Lender shall not be required to commence proceedings against such successor or refuse to extend time for payment or modification of amortization of the sums secured by this Subordinated Deed of Trust by reason of any demand made by the original Borrower and Borrower and Borrower's successors in interest.

9. FORBEARANCE BY LENDER NOT A WAIVER. Any forbearance by Lender in exercising any right or remedy hereunder, otherwise afforded by applicable law, shall not be a waiver of or preclude the exercise of any such right or remedy. The procurement of insurance or the payment of taxes or other liens or charges by Lender shall not be a waiver of Lender's right to accelerate the maturity of the indebtedness secured by this Subordinated Deed of Trust.

10. REMEDIES CUMULATIVE. All remedies provided in this Subordinated Deed of Trust are distinct and cumulative to any other right or remedy under this Subordinated Deed of Trust or afforded by law or equity, and may be exercised concurrently, independently or successively.

11. SUCCESSORS AND ASSIGNS BOUND; JOINT AND SEVERAL LIABILITY; CAPTIONS. The covenants and agreements contained in this Subordinated Deed of Trust shall bind, and the rights hereunder shall inure to, the respective successors and assigns of Lender and Borrower, subject to the provisions of Section 15 hereof. All covenants and agreements of Borrower shall be joint and several. The captions and headings of the Sections of this Subordinated Deed of Trust are for convenience only and are not to be used to interpret or define the provisions hereof.

12. NOTICE. Except for any notice required under applicable law to be given in another manner, (a) any notice to Borrower provided for in this Subordinated Deed of Trust shall be given by facsimile, a nationally recognized overnight courier service, or by mailing such notice by certified mail addressed to Borrower at 17701 Cowan Ave., Suite 200, Irvine, California 92614, Attention: President, or at such other address as Borrower may designate by notice to Lender as provided in this Section 12, and (b) any notice to Lender shall be given by facsimile, a nationally recognized overnight courier service, or by mailing such notice by certified mail, return receipt requested, to Lender's address stated in this Subordinated Deed of Trust or to such other address as Lender may designate by notice to Borrower as provided in this Section 12. Any notice provided for in this Subordinated Deed of Trust shall be deemed to have been given to Borrower or Lender upon receipt or refusal of delivery by the addressee.

13. GOVERNING LAW; SEVERABILITY. This Subordinated Deed of Trust shall be governed by the laws of the State of California. In the event that any provision or clause of this Subordinated Deed of Trust or the Note as incorporated herein conflicts with applicable law, such conflict shall not affect other provisions of this Subordinated Deed of Trust or the Note which can be given effect without the conflicting provision, and to this end the provisions of the Subordinated Deed of Trust and Note are declared to be severable.

14. BORROWER'S COPY. Borrower shall be furnished a conformed copy of the Note and of this Subordinated Deed of Trust at the time of execution or after recordation hereof.

15. TRANSFER OF THE PROPERTY. If all or any of the Property or an interest in it is sold or transferred by Borrower, excluding permitted transfers pursuant to this Subordinated Deed of Trust, the Note or the Partnership Agreement (as defined in the Note), all the sums secured by this Subordinated Deed of Trust shall be immediately due and payable. In the event of such an acceleration, Lender shall mail Borrower notice of the acceleration in accordance with Section 12 hereof. Such notice shall provide a period of not less than 30 days from the date the notice is mailed within which Borrower may pay the sums declared due. If Borrower fails to pay such sums prior to the expiration of such period, Lender may, without further notice or demand on Borrower, invoke any remedies permitted by Section 16 hereof.

16. ACCELERATION; REMEDIES Except as provided in Section 15 hereof, upon Borrower's breach of any covenant or agreement of Borrower in this Subordinated Deed of Trust or the Note as incorporated by reference in this Subordinated Deed of Trust, including the covenants to pay, when due, any sums secured by this Subordinated Deed of Trust, Lender, prior to acceleration, shall mail notice to Borrower as provided in Section 12 hereof specifying: (a) the breach; (b) the action required to cure such breach; (c) a date by which such breach must be cured, which date shall be extended by Lender so long as Borrower or investor limited partner of Borrower is diligently prosecuting a cure of such breach; and (d) that failure to cure such breach on or before the date specified in the notice may result in acceleration of the sums secured by this Subordinated Deed of Trust and sale of the Property. If the breach is not cured on or before the date specified in the notice, Lender, at its option, may declare all of the sums determined by this Subordinated Deed of Trust to be immediately due and payable without further demand and may invoke the power of sale and any other remedies permitted by applicable law. Lender shall be entitled to collect all reasonable costs and expenses incurred in pursuing the remedies provided in this Section 16, including, but not limited to, reasonable attorneys' fees.

Notwithstanding anything to the contrary contained in this Subordinated Deed of Trust, the Note, or any other document evidencing or securing the Note, during the time prior to the Maturity Date (as defined in the Note) that Lender or any affiliate of Lender acts as a general partner of Borrower, Lender shall have no right to declare a default hereunder or thereunder, foreclose on the Project, accelerate the maturity of any amounts due, obtain a receiver or exercise any other remedy hereunder or under any other document evidencing or securing said note without first obtaining the prior written consent of Borrower, the investor limited partner of Borrower, and the beneficiary of the Senior Encumbrances; provided however, that Lender may bring an action for specific performance to compel any payments due Borrower pursuant to Borrower's Partnership Agreement.

Subject to the terms and conditions provided herein, if Lender invokes the power of sale, Lender shall have appraised, or shall cause Trustee to have appraised, the Property to determine its fair market value. Lender shall also execute or cause Trustee to execute a written notice of the occurrence of an event of default and of Lender's election to cause the Property to be sold, and shall cause such notice to be recorded in each county in which the Property or some part thereof is located. Lender or Trustee shall mail copies of such notice in a manner prescribed by applicable law. After the lapse of such time as may be required by applicable law, Trustee, without demand on Borrower, shall sell the Property at public auction to the highest bidder at the time and place and under the terms designated in the notice of sale in one or more parcels and in such order as Trustee may determine. Trustee may postpone sale of all or any parcel of the Property by public announcement at the time and place of any previously scheduled sale. Lender or Lender's designee may purchase the Property at any sale.

Trustee shall deliver to the purchaser Trustee's deed conveying the Property so sold without any covenant or warranty, expressed or implied. The recitals in the Trustee's deed shall be prime facie evidence of the truth of the statements made therein. Trustee shall apply the proceeds of the sale in the following order: (a) to all reasonable costs and expenses of the sale, including, but not limited to, reasonable Trustee and attorneys' fees and costs of title evidence; (b) to all sums secured by the lien of some other or later security instrument, as evidenced by a note, by and between Borrower and Borrower's lender, that is either prior in lien priority to the priority of this Subordinated Deed of Trust or to which Lender has subordinated its lien rights and, thereby, placed itself in a subordinated position, (c) to all sums secured by this Subordinated Deed of Trust, as evidenced by the Note; and (d) the excess, if any, to the person or persons legally entitled thereto.

BORROWER'S RIGHT TO REINSTATE. Notwithstanding Lender's acceleration 17. of the sums secured by this Subordinated Deed of Trust, Borrower shall have the right to have any proceedings begun by Lender to enforce covenants of this Subordinated Deed of Trust relating to sums due and payable by Borrower discontinued at any time prior to five days before the Property is scheduled for sale pursuant to the power of sale contained in this Subordinated Deed of Trust or at any time prior to entry of a judgment enforcing this Subordinated Deed of Trust if: (a) Borrower pays Lender all sums which would be then due under this Subordinated Deed of Trust and the Note, had no acceleration occurred; (b) Borrower pays all reasonable expenses incurred by Lender and Trustee in enforcing the covenants and agreements of Borrower contained in this Subordinated Deed of Trust and in enforcing Lender's and Trustee's remedies as provided in Section 16 hereof, including, but not limited to reasonable attorneys' fees; (c) Borrower takes such actions as Lender may reasonably require to assure that the lien of this Subordinated Deed of Trust, Lender's interest in the Property, and Borrower's obligation to pay the sums secured by this Subordinated Deed of Trust shall continue unimpaired. Upon such payment and cure by Borrower, this Subordinated Deed of Trust and the obligations secured hereby shall remain in full force and effect as if no acceleration had occurred.

18. APPOINTMENT OF RECEIVER; LENDER IN POSSESSION. Upon acceleration under Section 16 hereof or abandonment of the Property, Lender, in person, by agent or by judicially appointed receiver, shall be entitled to enter upon, take possession of, rent and manage the Property and to collect the rents of the Property. All rents collected by Lender or by a judicially appointed receiver shall be applied first to payment of the cost of management of the Property and collection of rents, including, but not limited to, receiver's fees, premiums on receiver's bonds and reasonable attorneys' fees, and then to the sums secured by this Subordinated Deed of Trust. Lender and the receiver shall be liable to account for those rents actually received and expenditures actually incurred.

19. RECONVEYANCE. Upon payment of all sums evidenced by the Note and secured by this Subordinated Deed of Trust, Trustee shall reconvey the Property without warranty and without charge to the person or persons legally entitled thereto. Such person or persons shall pay all costs of recordation, if any.

20. SUBSTITUTE TRUSTEE. Lender, at Lender's option, may from time to time remove Trustee and appoint a successor trustee to act as Trustee under this Subordinated Deed of Trust. Without conveyance of the Property, the successor trustee shall succeed to all the title,

power and duties conferred upon the Trustee in this Subordinated Deed of Trust and by applicable law.

21. REQUEST FOR NOTICES. Borrower requests that copies of the notice of default and notice of sale be sent to Borrower's address which is the Property address.

22. STATEMENT OF OBLIGATION. Lender may collect a fee not to exceed \$15.00 for furnishing the statement of obligation as provided by Section 2943 of the Civil Code of the State of California.

23. SUBORDINATION ACKNOWLEDGMENT. Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is subordinate in right of payment to the prior payment in full of all amounts then due and payable (including, but not limited to, all amounts due and payable by virtue of any default or acceleration or upon maturity) with respect to (i) that certain , dated as of the date hereof, by Borrower in favor of First American Company, a California corporation, as trustee, for the benefit of Title Insurance _____, a _____, as beneficiary (the "First Deed of Trust"), (ii) that deed of trust relating to the loan from the Department of Housing and Community Development, a public agency of the State of California, or its affiliates, encumbering the Property (collectively, the "HCD Deed of Trust"), (iii) that deed of trust relating to the loan from the Orange County Development Agency, or its affiliates, encumbering the Property (collectively, the "OCDA Deed of Trust"), (iv) that deed of trust relating to the loan from the City of San Clemente, or its affiliates, encumbering the Property (collectively, the "City Deed of Trust"), and (v) any deeds of trust relating to any refinancing of the Property from a financial institution or governmental agency in an amount not to exceed the then-outstanding principal balance of the loans being prepaid plus reasonable closing costs, in the aggregate. Without limiting the generality of the foregoing, Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is also subordinate to the extended use agreement required to be executed by Borrower pursuant to Section 42(h)(6)(B) of the Internal Revenue Code, for purposes of the low-income housing tax credits to be allocated to Borrower. In addition, Beneficiary hereby acknowledges that the loan secured by this Subordinated Deed of Trust is further subordinate to Section 42(h)(6)(e)(ii) of the Internal Revenue Code, which prohibits the eviction or termination of a tenancy, other than for good cause, of an existing tenant of any low-income housing tax credit unit or any increase in the gross rent with respect to such unit, not otherwise permitted under Section 42, for a period of three (3) years after the date the Property is acquired by Lender through foreclosure or instrument in lieu of foreclosure.

24. NON-RECOURSE. This is a non-recourse Subordinated Deed of Trust. Neither Borrower, nor any of its partners shall have any personal liability for the payment of any portion of the indebtedness evidenced by the Note or performance of Borrower's obligations under this Subordinated Deed of Trust. In the event of a default by Borrower under the terms of the Note or this Subordinated Deed of Trust, the Beneficiary's sole remedy shall be limited to exercising its rights under the Note and this Subordinated Deed of Trust, including foreclosure and the exercise of the power of sale or other rights granted under the Note and this Subordinated Deed of Trust, but shall not include a right to proceed directly against Borrower, or any of its partners, or the right to obtain a deficiency judgment after foreclosure against Borrower or any of its partners.

[SIGNATURES ON FOLLOWING PAGE.]

IN WITNESS THEREOF, Borrower has executed this Subordinated Deed of Trust on the day and year first above written.

BORROWER:

AMISTAD HOUSING PARTNERS II LP, a California limited partnership

- By: JHC-AMISTAD II LLC, a California limited liability company, its general partner
 - Bv: Jamboree Housing Corporation, a California nonprofit public benefit corporation, its manager

By: _____ Name: Michael Massie Title: Executive Vice President and Chief

Development Officer

A Notary Public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.

State of California) County of _____)

On _____, before me, _____ (insert name and title of the officer)

Notary Public, personally appeared

who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or the entity upon behalf of which the person(s) acted, executed the instrument.

I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.

WITNESS my hand and official seal.

Signature _____

(Seal)

EXHIBIT "A"

LEGAL DESCRIPTION OF THE PROPERTY

Real property in the City of San Clemente, County of Orange, State of California, described as follows:

LOT 2 OF TRACT NO. 13898, AS SHOWN ON A MAP FILED IN <u>BOOK 817, PAGES 36 TO 40</u> INCLUSIVE OF MISCELLANEOUS MAPS, RECORDS OF ORANGE COUNTY, CALIFORNIA.

EXCEPTING THEREFROM ALL WATER AND WATER RIGHTS, IF ANY, INCLUDED WITHIN AND UNDERLYING THE DISTINCTIVE BORDER OF THIS TRACT MAP AS DEDICATED TO THE SANTA MARGARITA WATER DISTRICT ON THE MAP OF SAID TRACT.

ALSO EXCEPTING THEREFROM ANY AND ALL UNPROCESSED OIL, OIL RIGHTS, MINERALS, MINERAL RIGHTS, NATURAL GAS RIGHTS AND OTHER HYDROCARBONS BY WHATSOEVER NAME KNOWN, GEOTHERMAL STEAM AND ALL PRODUCTS DERIVED FROM ANY OF THE FOREGOING, THAT MAY BE WITHIN OR UNDER THE PROPERTY, TOGETHER WITH THE PERPETUAL RIGHT OF DRILLING, MINING, EXPLORING, AND OPERATING THEREFOR AND STORING IN AND REMOVING THE SAME FROM THE PROPERTY OR ANY OTHER LAND AS DESCRIBED THEREIN, BUT WITHOUT THE RIGHT TO ENTER UPON OR USE THE SURFACE OF THE PROPERTY TO DRILL, MINE, STORE, EXPLORE, OR OPERATE THROUGH THE SURFACE OR THE UPPER 500 FEET OF THE SUBSURFACE OF THE PROPERTY, RECORDED JUNE 28, 2002 AS INSTRUMENT NO. 20020548428.

APN: 701-041-38

PURCHASE AND SALE AGREEMENT AND ESCROW INSTRUCTIONS

by and between

JAMBOREE-TAL HOUSING II, L.P., a California limited partnership

("Seller")

and

AMISTAD HOUSING PARTNERS II LP, a California limited partnership

("Buyer")

TABLE OF CONTENTS

Page

| | | | Iage |
|-----|------------|--|------|
| Ι | AGR | EEMENT TO PURCHASE | |
| | 1.1 | The Property | 1 |
| | 1.2 | Purchase Price | 2 |
| | 1.3 | Carryback Financing | 2 |
| | 1.4 | Assumed Loans | 2 |
| II | ESCF | ROW | 3 |
| | 2.1 | Escrow Holder | |
| | 2.2 | Closing Date | - |
| | 2.3 | Escrow Instructions | |
| | 2.4 | Conveyance of Title | |
| | 2.5 | Additional Closing Obligations of Seller | |
| | 2.6 | Closing Obligations of Buyer | |
| | 2.0 | Delivery of Documents by Escrow Holder | |
| | 2.1 | Derivery of Documents by Escrow Horder | |
| III | | ECTIONS, REVIEW | 9 |
| | 3.1 | Access to Books and Records; Items to be Inspected and Reviewed by | _ |
| | | Buyer | |
| | 3.2 | Physical Inspection | |
| | 3.3 | Preliminary Title Report | 10 |
| IV | CON | DITIONS TO AGREEMENT | 11 |
| | 4.1 | Buyer's Conditions Precedent | 11 |
| | 4.2 | Seller's Conditions Precedent | |
| | 4.3 | Satisfaction of Conditions | 13 |
| | 4.4 | Waiver | |
| | 4.5 | Termination | 13 |
| V | | RATIONS, POSSESSION AND DEPOSITS | 13 |
| v | 5.1 | Proration of Taxes | |
| | 5.2 | Proration of Rents | |
| | 5.2 5.3 | Tenants' Deposits | |
| | 5.3 5.4 | Utilities | |
| | 5.4 5.5 | | |
| | 5.5 5.6 | Maintenance Contracts | |
| | 5.0 5.7 | Method of Adjustment Insurance | |
| | | Possession | |
| | 5.8 | | - |
| | 5.9 | Closing Costs | |
| | 5.10 | Closing Statement | |
| | 5.11 | BREACH BY SELLER | 16 |
| VI | AS-IS | S PURCHASE | 17 |
| | 6.1 | Disclaimer | |
| | 6.2 | Release and Waiver | 18 |
| | 6.3 | Waiver of Civil Code § 1542 | 18 |

Page

| | 6.4 | Representations of Seller | 19 |
|------|------|--|----|
| | 6.5 | Buyer's Representations and Warranties | 21 |
| | 6.6 | Seller's Covenants Pending Closing | 22 |
| VII | DAM | AGE, DESTRUCTION AND CONDEMNATION | 22 |
| | 7.1 | Risk of Physical Loss | 22 |
| | 7.2 | Condemnation | 23 |
| VIII | MISC | ELLANEOUS | 23 |
| | 8.1 | Attorneys' Fees | 23 |
| | 8.2 | Notices | 23 |
| | 8.3 | Entire Agreement | 24 |
| | 8.4 | Successors | 24 |
| | 8.5 | Assignment | 24 |
| | 8.6 | Choice of Laws | 24 |
| | 8.7 | Headings | 24 |
| | 8.8 | Survival | 25 |
| | 8.9 | Time | 25 |
| | 8.10 | Counterparts | 25 |
| | 8.11 | Brokerage Commissions | 25 |
| | 8.12 | Information Report | 25 |
| | 8.13 | Cooperation in 1031 Exchange | 25 |
| | 8.14 | No Third Party Beneficiary | 26 |
| | 8.15 | Limited Liability | 26 |
| | 8.16 | Jury Waiver | 26 |
| | 8.17 | Arbitration | 26 |
| | 8.18 | Acceptance of Grant Deed | 27 |
| | 8.19 | Bulk Sale Requirements | 27 |
| | 8.20 | Incorporation | 27 |



BOARD OF SUPERVISORS MEMORANDUM

SUGA

| TO: | Robin Stieler, Clerk of the Board | |
|---------------------|---|--------------------------|
| FROM: | Vicente Sarmiento, Supervisor Second District | bit thirty |
| DATE: | July 10 th , 2024 | |
| Subject: Meeting | Addition of Supplemental Item to the July 23, 202 | 24, Board of Supervisors |

Supervisor Sarmiento requests Clerk of the Board to add a supplemental agenda item be placed on the July 23, 2024, Board of Supervisors meeting, to adopt and recognize "National Rehabilitation Awareness Week" for the week of September $16^{\text{th}} - 22^{\text{nd}}$, 2024 and adding this recognition to the list of Recurring Resolutions.

0 E CE 11 10: 53 120

NATIONAL REHABILITATION AWARENESS WEEK PROCLAMATION

WHEREAS, the Physical, Occupational, and Speech Therapists at the University of California, Irvine Health Medical Center have consistently demonstrated an unwavering commitment to excellence, compassion, and innovation in the field of rehabilitation; and

WHEREAS, these dedicated professionals play a crucial role in enhancing the quality of life for patients through their exceptional skills in assessment, treatment, education, and support, both in inpatient and outpatient settings; and

WHEREAS, the week of September 16-22, 2024, is designated as National Rehabilitation Awareness Week, an annual event that recognizes and celebrates the vital contributions of rehabilitation therapists to the healthcare community and society at large; and

WHEREAS, the University of California, Irvine Health Medical Center takes immense pride in the remarkable achievements and exceptional professionalism of its Physical, Occupational, and Speech Therapists, who consistently uphold the highest standards of patient care and lead advancements in rehabilitation medicine; and

WHEREAS, National Rehabilitation Awareness Week reflects the tireless efforts of these therapists to provide expert care, support, and education that positively impact the lives of patients and their families;

WHEREAS, the undersigned, recognize and honor the Physical, Occupational, and Speech Therapists at the University of California, Irvine Health Medical Center for their significant contributions to healthcare, their dedication to the well-being of their patients, and their commitment to advancing the field of rehabilitation; and

NOW, THEREFORE, BE IT RESOLVED THAT THE ORANGE COUNTY BOARD OF SUPERVISORS proclaim the week of September 16-22, 2024, National Rehabilitation Awareness Week at the University of California, Irvine Health Medical Center. During this week, we encourage all members of our community to acknowledge and appreciate the vital role that

rehabilitation therapists play in enhancing patients' health and quality of life.



BOARD OF SUPERVISORS MEMORANDUM

SUGP

| TO: | Robin Stieler, Clerk of the Board | 1 1 - 16 |
|---------------------|--|------------------------|
| FROM: | Vicente Sarmiento, Supervisor Second District | hiphint |
| DATE: | July 10 th , 2024 | |
| Subject: Meeting | Addition of Supplemental Item to the July 23, 2024 | , Board of Supervisors |

Supervisor Sarmiento requests Clerk of the Board to add a supplemental agenda item be placed on the July 23, 2024, Board of Supervisors meeting, to adopt and recognize "Respiratory Care Week" for week of October $20^{\text{th}} - 26^{\text{th}}$, 2024 and adding this recognition to the list of Recurring Resolutions.



RESPIRATORY CARE WEEK PROCLAMATION

WHEREAS, the Respiratory Care Practitioners from the University of California, Irvine Health Medical Center have consistently demonstrated an unwavering commitment to excellence, compassion, and innovation in the field of respiratory care; and

WHEREAS, these dedicated individuals play an integral role in promoting optimal lung health and improving the quality of life for patients through their exceptional skills in assessment, treatment, education, and support; and

WHEREAS, the week of October 20-26, 2024, marks the observance of Respiratory Care Week, an annual event that recognizes and celebrates the vital contributions of respiratory care practitioners to the healthcare community and society at large; and

WHEREAS, the University of California, Irvine Health Medical Center takes immense pride in the remarkable accomplishments and exceptional professionalism exhibited by its respiratory care practitioners, who consistently uphold the highest standards of patient care and remain at the forefront of advancements in respiratory medicine; and

WHEREAS, Respiratory Care Week reflects the tireless efforts of respiratory care practitioners to provide expert care, support, and education that positively impact the lives of patients and their families;

WHEREAS the undersigned, recognize and honor the respiratory care practitioners from the University of California, Irvine Health Medical Center for their significant contributions to healthcare, their dedication to the well-being of their patients, and their commitment to advancing the field of respiratory care; and

NOW, THEREFORE, BE IT RESOLVED THAT THE ORANGE COUNTY BOARD OF SUPERVISORS proclaim the week of October 20-26, 2024, as Respiratory Care Week at the University of California, Irvine Health Medical Center. During this week, we encourage all members of our community to acknowledge and appreciate the vital role that respiratory care practitioners play in enhancing the health and quality of life of patients.

.

| STY OF O | County of Orange S49C |
|--------------|--|
| S S | Memorandum 2024 JUL 10 PM 2: 05 |
| CALIFORD | COUNTY OF CRIME BCARD DE SUPE TO TRS |
| July 3, 2024 | 1 (10 2) |
| То: | Chairman Donald P. Wagner, Supervisor, Third District Augul for DW |
| From: | Frank Kim, County Executive Officer |
| Subject: | Supplemental Recommended At Large Nominations To the Commission to End Homelessness |

Pursuant to Board of Supervisors Rule of Procedure 16 amended on July 13, 2021, the Office of Care Coordination recommends the following at large appointments to the Commission to End Homelessness for placement on the July 23, 2024 Board meeting agenda:

- Appoint the following nominee to the Commission to End Homelessness to complete the existing term effective July 23, 2024 and ending December 5, 2024:
 - Cecilia Bustamante Pixa as the Hospital Representative

Information on the recommended at large nomination is contained in the attached Agenda Staff Report and candidate application.

Please contact Doug Becht, Director of Care Coordination, at 714-834-2323 or <u>Doug.Becht@ocgov.com</u> with any questions.

Thank you.

Attachments Agenda Staff Report Attachment A – Cecilia Bustamante Pixa Nomination Form, Application and Resume

Cc: Clerk of the Board

Agenda Item Clerk's Use Only

....

50

 $\langle \rangle$

SUPPLEMENTAL AT-LARGE APPOINTMENT **AGENDA STAFF REPORT**

| AT OF | |) | 221 | |
|------------------------------------|---------------------------------------|-----------|------------------------------|--|
| MEETING DATE: | 07/23/2024 | | | |
| LEGAL ENTITY TAKING ACTION: | Board of Supervisors | | | |
| BOARD OF SUPERVISORS CHAIR: | Chairman Wagner, Supervisor, Third Di | strict | с) | |
| NOMINATION SUBMITTED BY (DEPT): | County Executive Office | | 5 | |
| DEPARTMENT CONTACT PERSON(S): | Douglas Becht (714) 834-5000 | ્રોસેટ્રે | 12 | |
| | Lilly Simmering (714) 834-6234 | ()) | $\overline{\mathbf{\omega}}$ | |

SUBJECT: Supplemental At-Large Nominations for Appointment to the Commission to End Homelessness

| CEO CONCUR | COUNTY COUNSEL REVIEW | CLERK OF THE BOARD |
|------------|-----------------------|------------------------|
| N/A | N/A | Discussion |
| | | 3 Votes Board Majority |

RECOMMENDED ACTION(S)

1. Appoint the following nominee to the Commission to End Homelessness effective July 23, 2024, to complete the existing term ending December 5, 2024:

-Cecilia Bustamante Pixa as the Hospital Representative

SUMMARY:

Appointment of the recommended nominee to the Commission to End Homelessness will fill one vacant seat which helps bring local leadership together to address homelessness in Orange County.

BACKGROUND INFORMATION:

The Commission to End Homelessness (Commission) works in collaboration with the County of Orange, thirty-four city governments, business sector, philanthropic organizations, community organizations, faithbased organizations, health care, public safety, and other interested stakeholders to promote an effective response to homelessness within Orange County. The Commission membership is comprised of twentyone members, including nineteen voting members and two non-voting members.

On March 29, 2024, Christy Cornwall, Hospital Representative, resigned from the Commission as she transitioned out of her role with Providence Health System to the City of Irvine. The Hospital Representative's vacant seat has a term ending on December 5, 2024. Per the Commission Bylaws, the Hospital Representative is an individual with expertise in the local hospital emergency room treatment and discharge system.

On April 17, 2024, the Commission approved a Membership Ad Hoc Committee to support membership recruitment process for the appointment of the Hospital Representative. On April 22, 2024, the Office of Care Coordination launched the recruitment for the vacant Hospital Representative seat on the Commission. The application period was open through May 24, 2024. The Membership Ad Hoc met to review the

applications received and recommended a candidate to be presented to the Commission for approval. Cecilia Bustamante Pixa, Senior Director of Community Health at Providence St. Joseph Hospital, was recommended for appointment to the Hospital Representative by the Membership Ad Hoc Committee.

Cecilia Bustamante Pixa has been a health care professional with Providence St. Joseph Hospital for 25 years. During this time, she has held multiple positions including Director of Community Benefit, Director of Community Health Investment, and most recently, Senior Director of Community Health Investment Orange County and the High Desert. In her current position, Ms. Bustamante Pixa is responsible for the community health programs and services across four hospitals. She leads key initiatives including Access to Care, Mental/Behavioral Health/Substance Use Disorders, homelessness and housing, and economic stability. Ms. Bustamante Pixa holds a bachelor's degree in Community Health and a Health Care Administration certificate from California State University Long Beach, a master's degree in Public Health from University of California Los Angeles, a master's degree in Health Care Mission Leadership from Loyola University Chicago, and a Mission Leadership Academy Certificate from Providence University. Additionally, she has been involved in numerous community partnerships including as an interpreter on the Medical Teams International in Global Partnerships Guatemala, a co-facilitator and interpreter for Leadership Formation Pilgrimage, and a participant in the Movement for a Better World Homeless Ministry and the Sisters of St. Joseph Pilgrimage.

On June 19, 2024, the Commission approved the appointment of Cecilia Bustamante Pixa to serve on the Commission to End Homelessness and recommended to the Board of Supervisors (Board) for final approval.

The County of Orange request the Board approve the recommended appointment to the Commission as referenced in the Recommended Action.

FINANCIAL IMPACT:

N/A

STAFFING IMPACT:

N/A

ATTACHMENT(S):

Attachment A – Cecilia Bustamante Pixa Nomination Form, Application and Resume

| OF | range Coun | ity Board | of Super | Attachment A RVISORS |
|---|--|--------------------------|--------------------|-------------------------|
| CALIFORNIA Nom | ination for Boards, C | ommissions & Con | nmittees (Rev. 8/2 | 9/23) |
| Agenda Date: 07/23 | | | | Item # |
| To: Members of | the Orange County Bo | ard of Supervisors | | |
| | Board of Supervisors | | | |
| | Director, Office of Car | e Coordination | | |
| | | v interaction | | |
| It is my intent to ap Name: <u>Cecil</u> | <i>point:</i> i <u>a Bustamante Pixa</u> | | | |
| Address: | | | | |
| City & Zip: | | | | |
| Day Phone: | Fax Nu | umber: () | E-mail address | : |
| | | | | |
| | mission to End Homele te of Board, Commission | | | |
| Position Slot: <u>Hosp</u> | ital Representative | | | |
| Name of incumbent | t being replaced or last | known member: <u>C</u> h | nristy Cornwell | |
| Term (Choose One): | | | | |
| = | rs Beginning Term Date | | ation Term Date: | 12/05/2024. |
| Term Concurrer | nt with Supervisor's Ter nt with position | m of Office | | |
| acancy occurred due | to: | | | |
| Resignation (atta | ch letter of resignation) | | | |
| | ovide reason for termina | | Death | |
| Expiration of Te | | ovide reason in remar | | |
| Nomination to: | Appoint | Reappointment | Newly Form | ned Committee |
| Qualifications: | Attached (must b | e attached for appoir | itments and reap | pointments) |
| Remarks: | For Clerk | of the Board Use Only | V | |
| Clerk's Initials: | File I.D | | | e Letter |
| Contact Name | Support | ing Agency | Mai | l or 🗌 Pony |
| Appoint/Complete: | Term Years CWS | Term Dates: _ Other | to | and the second |
| Check one: So | cheduled Vacancy | Unscheduled Va | icancy | Page 1 of 7 |

Certification of posting attached.

Posted on _____ to_____

Attachment A



APPLICATION FOR COUNTY OF ORANGE BOARD, COMMISSION OR COMMITTEE (FOR COUNTY USE ONLY)

County Executive Office Office of Care Coordination 601 N. Ross Street, 5th Fl Santa Ana, California 92701

Instructions: Please complete each section below. Be sure to enter the title of the Board, Commission or Committee for which you desire consideration. For information or assistance, please contact the Clerk of the Board of Supervisor's Office at (714) 834-2206. Please print in ink or type.

NAME OF BOARD, COMMISSION, OR COMMITTEE TO WHICH YOU ARE APPLYING FOR MEMBERSHIP (SEE LIST AT HTTP://WWW.OCGOV.COM/GOV/COB/BCC/CONTACT):

Orange County Commission to End Homelessness

Return to:

SUPERVISORIAL DISTRICT IN WHICH YOU RESIDE: First Second Third Fourth Fifth

APPLICANT NAME AND RESIDENCE ADDRESS:

| Cecilia | | Bustamante Pixa | | |
|--------------------------|--|----------------------|-------------|--|
| First Name | Middle Name | Last Name | | |
| Street Address | City | State | Zip Code | |
| N/A Home Phone Number | ər | Cell Phone Number | | |
| Email Address | | | | |
| GURRENT EMPLOYER: | Providence St. Joseph Hos | pital | | |
| OCCUPATION/JOB TITLE: | Senior Director, Community Hea | alth Orange County/i | High Desert | |
| BUSINESS ADDRESS: | | | | |
| BUSINESS PHONE NUMBE | R: | | | |
| | rrent resumé to this application that inclu oful in evaluating your application. Applic | | | |
| ARE YOU A CITIZEN OF TH | E UNITED STATES: NO | | | |
| IF NO, NAME OF COUNTRY | OF CITIZENSHIP: | | | |
| ARE YOU A REGISTERED V | Orange | e County | | |
| | | | | |
| Pavised Date 04/10/23 | | | Page 1 of 2 | |

LIST ALL CURRENT PROFESSIONAL OR COMMUNITY ORGANIZATIONS AND SOCIETIES OF WHICH YOU ARE A MEMBER.

| ORGANIZATION/SOCIETY | FROM (MO./YR.) | <u>TO (MO./YR.)</u> | |
|--|----------------|---------------------|--|
| The Cambodian Family Board of Directors | 05/19 | present | |
| St. Jude Neighborhood Health Centers (FQHC) Board of Directors | 12/19 | present | |
| OCHCA Collaborative Community Health Improvement Plan | 04/23 | present | |

WITHIN THE LAST FIVE YEARS, HAVE YOU BEEN AFFILIATED WITH ANY BUSINESS OR NONPROFIT AGENCY(IES)? SYSS D NO

DO YOU OWN REAL OR PERSONAL PROPERTY OR HAVE FINANCIAL HOLDING WHICH MIGHT PRESENT A POTENTIAL CONFLICT OF INTEREST? DYES IN NO

HAVE YOU BEEN CONVICTED OF A FELONY OR MISDEMEANOR CRIME SINCE YOUR 18TH BIRTHDAY? YOU ARE NOT REQUIRED TO DISCLOSE ANY OF THE FOLLOWING: ARRESTS OR DETENTIONS THAT DID NOT RESULT IN A CONVICTION; CONVICTIONS THAT HAVE BEEN JUDICIALLY DISMISSED, EXPUNGED OR ORDERED SEALED; INFORMATION CONCERNING REFERRAL TO AND PARTICIPATION IN ANY PRETRIAL OR POSTRIAL DIVERSION PROGRAM; AND CERTAIN DRUG RELATED CONVICTIONS THAT ARE OLDER THAN TWO YEARS, AS LISTED IN CALIFORNIA LABOR CODE § 432.8 (INCLUDING VIOLATIONS OF CALIFORNIA HEALTH AND SAFETY CODE SECTIONS 11357(B) AND (C), 11360(C) 11364, 11365 AND 11550 – AS THEY RELATE TO MARIJUANA)?

IF YES, PLEASE EXPLAIN AND ATTACH ADDITIONAL SHEETS, IF NECESSARY.

PLEASE BRIEFLY EXPLAIN WHY YOU WISH TO SERVE ON THIS BOARD, COMMITTEE, OR COMMISSION. ATTACH ADDITIONAL SHEETS, IF NECESSARY.

See attached Word document.

DATE: May 6, 2024

| APPLICANTS | SIGNATURE: | |
|------------|------------|--|
| | · · | |

CLERK OF THE BOARD OF SUPERVISORS USE ONLY - DO NOT WRITE BELOW/THIS/LINE

| Date Received: | | Received by: | Deputy Clerk of the Board of Supervisors | | |
|----------------|----------------|-------------------------|--|----------------|----------------|
| To: | BOS District 1 | BOS District 2 | BOS District 3 | BOS District 4 | BOS District 5 |
| | □ All BOS | BCC Contact Person Name | | | |

Cecilia Bustamante Pixa, MPH, MHCML

v Summary

Dedicated public health/health care professional with a focus on community health investment, health equity, and improving access to care for underserved populations. Proven expertise in designing and implementing initiatives that address social determinants of health and reduce disparities in healthcare. Skilled in building strategic partnerships, advocating for policy changes, and empowering communities to lead healthier lives. Passionate about creating sustainable solutions that prioritize health equity and improve overall well-being for all individuals.

v Experience

Senior Director, Community Health Investment Orange County & High Desert March 1, 2024 to present Responsible for Orange County and High Desert ministries (4 hospitals) community health programs and services.

Lead priority initiatives: Access to Care, Mental/Behavioral Health/Substance Use Disorders, Homelessness & Housing, and Economic Stability.

Maintain responsibilities of previous role.

Director, Community Health Investment St. Joseph Hospital and St. Jude Medical Center January 1, 2019 to February 28, 2024

- Responsible for strategic planning, budgeting, and implementing operational metrics for local community health investment efforts.
- Co-lead change management transition of hospital community clinic dept. to Federally Qualified Health Center.
- Serve as Board Chair on Board of Federally Qualified Health Center affiliated with three PSJH Southern California Ministries.
- Hospital Liaison to the PSJH Community Partnership Fund for identifying community partners and funding opportunities.
- Responsible for maintaining collaborative partnership and supporting local advocacy efforts with PSJH Government & Public Affairs Lead in Southern California region.
- Member of the hospital senior leadership team responsible for ensuring the mission, vision and values as it relates to Community Health Investment, health promotion and medical access for the underserved, vulnerable, and immigrant population.
- Lead local Community Benefit Committee in the Community Health Needs Assessment process and development of the Community Health Improvement Plan.
- Responsible for Regional Community Health Improvement Priorities at local ministry level in areas of Access to Care, Mental Health & Substance Use, and Homelessness and Affordable Housing 2024-2026
- Cultivate relationships with community partners, i.e., Public Health Agency, Community Based Organizations, Health Centers, Advocacy Groups, and local School Districts.
- Serve on Board of Directors for The Cambodian Family Immigrant & Refugee Resource Center.

Director, Community Benefit . St. Joseph Hospital, Orange 2008 to 2019

Responsible for hospital-wide Interpreter Services Department & Language Access.

Procured, implemented, and deployed Video Remote Interpreting devices (VRI) throughout the hospital (inpatient/outpatient settings) to address language access for culturally diverse patient population.

Developed and Implemented Bilingual/Bicultural Hospitality Volunteer Program to address diverse patient population and local community needs.

- Responsible for mission, direction and oversight of daily operations of the SJO Community Clinics and Community Services Depts.
- Responsible for strategic planning process for Community Benefit assessment, planning, implementation and reporting.

Ensured compliance to Advancing the State of the Art in Community Benefit (ASACB) principles and standards Analysis of Community Benefit and Budget Philosophy financial components and overall management of CBISA reporting.

- Cultivated relationships with community partners and collaborative groups to identify and build upon community assets.
- Assisted in the development and implementation of community outreach programs.

Direct supervision for the Joint Clinic Health Education staff.

Demonstrated knowledge and accountability for fiscal aspects of department/unit operations, advocated for resources including new technology to support unit and division goals. Developed and monitored

operational budget and productivity measurements. Submitted and justified capital budget requests.

Manager, Community Benefit • St. Joseph Hospital, Orange 1999-2008

Responsible for coordinating, tracking and managing community benefit monthly and quarterly reports.

Implemented and managed multi-year grant funded programs: Accessing Better Care for Diabetes (ABCD) Program and Medikids Program serving uninsured/undocumented children. Monitored and ensured adherence to budgeted financial and productivity performance targets. Participated in related hospital committees and initiatives (diversity, language access).

v Additional Skills, Responsibilities, and Accomplishments

- Bicultural/Biliterate, fluent native Spanish speaker
- Strong verbal/written communication and organizational skills
- Providence Mission Leadership Academy Discerning Mission Leader graduate cohort #1
- Providence St. Joseph Hospital Health Equity Council Co-Chair
- Board Chair, St. Jude Neighborhood Health Centers (Federally Qualified Health Center)
- Board of Director Member, The Cambodian Family Resource Center
- Catholic Health Association of USA Foundations of Catholic Health Care Leadership 2021
- Providence Diversity Equity Inclusion (DEI) Design Team
- Providence Southern California Activation Network Co-Chair
- CORO Southern California Women in Leadership Program Alumni 2020

Providence University

Mission Leadership Academy Certificate Degree, February 2023

Loyola University Chicago

Master of Arts, Health Care Mission Leadership, August 2019

Integrated courses from theology, spirituality, bioethics, organizational development, social justice, and leadership to offer a well-rounded degree pertinent to the work being done in Catholic hospitals and health systems throughout the world. Strong emphasis in Catholic Social Teaching

University of California Los Angeles (UCLA)

Master of Public Health, Executive MPH Program: Health Services Management, June 2002

California State University, Long Beach

Bachelor of Science, Health Science: Community Health, May 1997 Health Care Administration Certificate, Latino Health Care Professionals Project (LHPP), May 1997

v Community

Global Partnerships Guatemala, 2019 & 2024- Interpreter on the Medical Teams International. Sisters of St. Joseph of Orange Pilgrimage Le Puy, France 2016- Participant. Leadership Formation Pilgrimage, Guatemala 2013, 2014- Co-facilitator and Interpreter. Movement For a Better World- Homeless Ministry, Long Beach since 2007 – collect and deliver food and clothing to underserved/low-income communities and persons experiencing homelessness/unhoused.

v References

[Available upon request.]

Please briefly explain why you wish to serve on this board, committee, or commission.

As a hospital representative, I am deeply committed to social justice and equality, and I believe that addressing homelessness is a critical aspect of creating a more just society. By serving on the Commission to End Homelessness, I would have the opportunity to contribute my skills and knowledge to support initiatives that provide shelter, support services, and resources to individuals experiencing homelessness. As a faith-based major health system in Orange County, Providence is passionate about advocating for policies and programs that address the root causes of homelessness, such as housing affordability, access to healthcare, and economic opportunity. Serving on this committee would allow Providence to be part of a collective effort to create lasting change and improve the lives of those in our community who are most vulnerable.

In addition, as a resident of Orange County, I am deeply invested in the wellbeing of my community and am passionate about addressing the issue of homelessness in our region. I believe that by working collaboratively with key stakeholders, including government agencies, non-profit organizations, and community members, we can develop innovative strategies to prevent and end homelessness in Orange County. Providence is dedicated to advocating for systemic changes, increasing access to affordable housing and supportive services, and fostering a more inclusive and compassionate community for all residents. Serving on this commission representing Providence would allow me to leverage my skills, knowledge, and passion to make a meaningful difference in addressing homelessness in Orange County.